



#### OCCASION

This publication has been made available to the public on the occasion of the 50<sup>th</sup> anniversary of the United Nations Industrial Development Organisation.

TOGETHER

for a sustainable future

#### DISCLAIMER

This document has been produced without formal United Nations editing. The designations employed and the presentation of the material in this document do not imply the expression of any opinion whatsoever on the part of the Secretariat of the United Nations Industrial Development Organization (UNIDO) concerning the legal status of any country, territory, city or area or of its authorities, or concerning the delimitation of its frontiers or boundaries, or its economic system or degree of development. Designations such as "developed", "industrialized" and "developing" are intended for statistical convenience and do not necessarily express a judgment about the stage reached by a particular country or area in the development process. Mention of firm names or commercial products does not constitute an endorsement by UNIDO.

#### FAIR USE POLICY

Any part of this publication may be quoted and referenced for educational and research purposes without additional permission from UNIDO. However, those who make use of quoting and referencing this publication are requested to follow the Fair Use Policy of giving due credit to UNIDO.

#### CONTACT

Please contact <u>publications@unido.org</u> for further information concerning UNIDO publications.

For more information about UNIDO, please visit us at www.unido.org

## OF



1.0 2.2 2.2 2.2 1.1 1.8 1.25 1.4 1.6

$$\label{eq:product} \begin{split} & \mathbf{M} = \mathbf{u}_{1} + \mathbf{v}_{2} + \mathbf{k}_{3} + \mathbf{k}_{4} + \mathbf$$

# 24 **\***

¥



#### $(\mathbf{R})$ AGRO-INDUSTRIAL COMPLEX

#### IN CAN THO, VIETNAM

DP/RW/73/002

(UNIDO CONTRACT 74/21)

MAY. (1975)

V9. . . ;

TOKYO, JAPAN

PACIFIC PROJECTS, LTD. MITSUI CONSULTANTS CO., LTD. TOKYO, JAPAN

DP/RVH /73/002

AGRO-INDUSTRIAL COMPLEX IN CANTHO, VIETNAM (UNIDO Contract 74/21)

Authors: Trank L. Turner, Project Director; Tadayuki Aoike, Soils and Tropical Agriculture: Susumu Tsuji, Food Processing

(PPL Project No. 177)



Prepared for UNIBO P.O. Box 707 A-1011, Vienna, Austria 1

Prepared by Pacific Projects, Ltd. International P.O. Box 5361 Takyo 100-31 Tel. 502-0567/8/9 Cables: PACPROJECTS TOKYO and Mitsui Consultants Co., Ltd. 3-7, 3-chome Nihonbashi-Muromachi, Chuo-ku, Tokyo Japan

Tel. 279-2321/25

Cables: MICONTO Tokyo

May , 1975

#### AGRO-INDUSTRIAL COMPLEX IN CANTHO, VIETNAM

ſ

#### Table of Contents

Subject of Report       10         Tay Do Industrial Park       10         Table I-1,2, Pacilities Built or Planned on Tay Do Industrial       11         Park       11         Figure I-1,2, Layout of Tay Do Industrial Park, Cantho       12         Parpose of This Report       13         Content of the Report       13         Nathedology of the Report       13         Acknowledgements       15         1.       LEPCUTIVE SUMMAT AND CONCLUSIONS         1.       HEPCUTIVE SUMMATION REGARDING THE DELTA REGION AND CANTHO.         2.0       Parpose of Chapter         2.1       Area and Population.         2.1       Area and Population.         2.1       COMPARISON OF THE DELTA REGION POPULATION DENSITY WITH         OTHER AREAS       25         Table 2.1-c       ESTIMATED DREARDOW OF THE DELTA CIVILIAN LABOR FORCE.         23       Table 2.1-a         Table 2.1-b       OUMPARISON OF THE DELTA CIVILIAN LABOR FORCE.         25       Figure 2.1-a         Table 2.1-c       ESTIMATED DREARDOWN OF THE DELTA CIVILIAN LABOR FORCE.         25       Figure 2.1-a         Table 2.1-b       VIETNAN AND THE DELTA         Pigure 2.1-a       THE DELTA.         24       Se	INTRODUCTION	10
Tay Do Industrial Park       10         Table I-1,2, Pacilities Built or Planned on Tay Do Industrial       11         Park       11         Figure I-1,2, Layout of Tay Do Industrial Park, Cantho       12         Parpose of This Report       13         Content of the Report       13         Nathedology of the Report       13         Acknowledgements       15         1. EXECUTIVE SUMMATION REGARDING THE DELTA REGION AND CANTHO.       21         2.0 Parpose of Chapter       21         2.1 Area and Population.       21         Table 2.1-a PEDJECTION OF POPULATION OF VIETNAM AND THE DELTA.       21         Table 2.1-a DELTA STATISTICS.       25         Figure 2.1-a THE DELTA.       25         Figure 2.1-a THE DELTA.       26         Figure 2.1-b COMPARISON OF THE DELTA CIVILIAN LABOR FORCE.       23         Table 2.1-c ESTIMATED BREAMDOWN OF THE DELTA CIVILIAN LABOR FORCE.       25         Figure 2.1-a THE DELTA.       26         Figure 2.1-a THE DELTA.       27         2.2 Soil and Water Characteristics       26         Pigure 2.3, Water Levels at Chaudoc       30         Pigure 2.3, Water Levels at Chaudoc       32         Table 2.1-b VIETNAM AND THE MENDER DELTA.       32         Table 2.3, A	Subject of Report	10
Table I-1.2, Pacilities Built or Planned on Tay Do Industrial Park       11         Figure I-1.2, Layout of Tay De Industrial Park, Cantho       12         Purpose of This Report       13         Content of the Report       13         Mathedology of the Report       13         Acknowledgements       15         1.       HEPCUTIVE SUMMAY AND CONCLUSIONS       16         2.       BACHGARGUND INFORMATION REGARDING THE DELTA REGION AND CANTHO	Tay Do Industrial Park	10
Park       11         Figure I-1,2, Layout of Tay De Industrial Park, Cantho       12         Purpose of This Report       13         Centent of the Report       13         Nothedology of the Report       13         Acknowledgements       15         1.       IMPOUND INFORMATION REGARDING THE DELTA REGION AND CANTHO	Table I-1.2, Facilities Built or Planned on Tay Do Industrial	
Figure I-1,2, Layout of Tay De Industrial Park, Cantho       12         Purpose of This Report       13         Content of the Report       13         Nethodology of the Report       13         Acknowledgements       13         1.       EXPCUTIVE SUPPARY AND CONCLUSIONS       16         2.       Machinary AND CONCLUSIONS       16         2.       Machinary AND CONCLUSIONS       16         2.       Machinary AND CONCLUSIONS       11         1.       EXPCUTIVE SUPPARY AND CONCLUSIONS       16         2.       Machinary AND CONCLUSIONS       16         2.       Machinary AND CONCLUSIONS       11         1.       EXPCUTIVE SUPPARY AND CONCLUSIONS       11         1.       EXPCUTIVE SUPPARY AND CONCLUSIONS       16         2.       Machinary AND Conclusions       11         1.       EXPCUTIVE SUPPARY AND CONCLUSIONS       11         2.0       Purpose of Chapter       21         2.1       Area and Population       21         2.1       Area and Population       21         3.1       Table 2.1-a       PROJECTION OF POPULATION OF VIETNAM AND THE DELTA         3.1       Table 2.1-b       DELTA STATISTICS         2.2 <t< th=""><th></th><th>11</th></t<>		11
Rurpose of This Report       13         Centent of the Report       13         Nathodology of the Report       13         Acknowledgements       13         1.       ENECUTIVE SUMMARY AND CONCLUSIONS       16         2.       MacMandarian       17         2.0       Parpose of Chapter       21         2.1       Area and Population       21         Table 2.1-a       PhouseCrion of Population of Vietnam and The Delta       21         Table 2.1-b       COMPARISON OF THE DELTA REGION POPULATION DENSITY WITH OTHER AREAS       22         Table 2.1-c       ESTIMATED BREAMBOWN OF THE DELTA CIVILIAN LABOR FORCE       23         Table 2.1-c       ESTIMATED DELTA       25         Figure 2.1-a       THE DELTA       26         Figure 2.1-b       VIETNAM AND THE READING DELTA       27         2.2       Soil and Water Characteristics       2	Figure 1-1,2, Layout of Tay De Industrial Park, Cantho	12
Content of the Report	Purpose of This Report	13
Mathedology of the Report       13         Acknowledgement#       15         1. EXECUTIVE SUMMAX AND CONCLUSIONS       16         2. BACHGROUND INFORMATION REGARDING THE DELTA REGION AND CANTHO	Content of the Report	13
Acknowledgements       15         1. <u>EXECUTIVE SUMMARY AND CONCLUSIONS</u> 16         2. MACHOMOUND INFORMATION REGARDING THE DELTA REGION AND CANTHO	Nethodology of the Report	13
1. EXECUTIVE SUPPORT AND CONCLUSIONS       16         2. BACHGROUND INFORMATION REGARDING THE DELTA REGION AND CANTHO	Acknowledgements	15
1. EXECUTIVE SUMMARY AND CONCLUSIONS       16         2. BACHGROUND INFORMATION REGARDING THE DELTA REGION AND CANTHO		
2. BACHEGROUND INFORMATION REGARDING THE DELTA REGION AND CANTHO	1. EXECUTIVE SURVARY AND CONCLUSIONS	16
2. MACHIGARGUND INFORMATION REGARDING THE DELTA REGION AND CANTHO	×	• -
2.0 Parpose of Chapter	2. BACHGINGUND INFORMATION REGARDING THE DELTA REGION AND CANTHO	21
2.1 Area and Population	2.0 Parpose of Chapter	21
Table 2.1-aPROJECTION OF POPULATION OF VIETNAM AND THE DELTA	2.1 Area and Population	21
Table 2.1-bCOMPARISON OF THE DELTA REGION POPULATION DENSITY WITH OTHER AREAS	Table 2.1-a FROJECTION OF POPULATION OF VIETNAM AND THE DELTA	21
OTHER AREAS	Table 2.1-b COMPARISON OF THE DELTA REGION POPULATION DENSITY WITH	
Table 2.1-cESTIMATED BREAMBONN OF THE DELTA CIVILIAN LABOR FORCE23Table 2.1-dDELTA STATISTICS	OTHER AREAS	22
Table 2.1-d DELTA STATISTICS	Table 2.1-c ESTIMATED BREAKBOWN OF THE DELTA CIVILIAN LABOR FORCE	23
Figure 2.1-a THE DELTA	Table 2.1-d DELTA STATISTICS	25
Figure 2.1-b       VIETNAM AND THE MEMONG DELTA	Figure 2.1-a THE DELTA	26
<ul> <li>2.2 Soil and Water Characteristics</li></ul>	Figure 2.1-5 VIBTNAM AND THE MEMONG DELTA	27
2.3 Production of Agricultural, Fisheries Products and Livestock , 30 Figure 2.3, Water Levels at Chaudoc	2.2 Soil and Water Characteristics	28
Figure 2.3, Water Levels at Chaudoc       32         Table 2.3, Annual Production of Principal Agricultural and Fisheries Products, Livestock and Poultry	2.3 Production of Agricultural, Fisheries Products and Livestock .	30
Table 2.3, Annual Production of Principal Agricultural and Fisheries Products, Livestock and Poultry	Figure 2.1. Mater Levels at Chaudoc	32
and Fisheries Products, Livestock and Poultry	Table 2.1. Annual Broduction of Principal Agricultural	
	and Fisheries Products, Livestock and Poultry	
Raising of the Mokong Dolta 1973, Ranked by Market Value in Millions of U.S. Dollars	Raising of the Mokong Dolta 1973, Ranked by Narket Value in Millions of U.S. Dollars	33

ł

2.4	Transportation	, 36
	2.4.1 Highway and Rail	36
	2.4.2 Vaterways	36
	Figure 2.4, Water Depth in the Approaches to the Bassac and Nekong Rivers (Map)	38
2.6	Industry	39
2.6	Buitability of Cantho for Industry	40

#### ). CANDIDATE INDUSTRIAS

	Contact of this Chapter assessment and the second s	43
3.0	Contont of this comparison of Armonindustries as a second strain of the second strain of the second strain of the second strain	43
3.1	Inter-relationship of Agro-induction of the second se	43
3.2	Paddy and By-product processing	44
	Figure 3.1, Theoretical Inter-relationship of Age of Human 100	L.R.
	Table 3.2-A, Exports of Thailand Rice, 1973, 1974	10
	Figure 3.2-A, Delta Rice Production	50
	Table 3.2-B Vietnam Rice Exports and Imports 1954-74	51
	Figure 3.2-B, Rice Production Calendar in Vietnam	52
	Table 3.2-C, Delta Rainfall	53
	3.2.1 Bran Utilization	56
	Table 3.2.1, Estimated Nice Production 1969-70 in Different World Regions	<b>5</b> 6
	3.2.2 Processed Foods Derived from Rice	58
	3.2.3 Parboiled Rice	58
	3.2.4 Utilization of Rice Straw	59
	3.2.5 Utilization of Husk	62
	Table 3.2.5, Possible Industrial Applications for Paddy Hunk	65
	Figure 3.2.5, Industrial Applications for Purfural	67
	Figure 3.2.6, Cooking Stove for Burning Hunk	69
3.3	Pruits and Vegetables	<b>7</b> 2
2.12	1.3.1 Vegetables	72
	Table 3.3.1. Production of Vegetables in Vietnam 1965-73 *****	72
	1.1.2 Fruits	74
	Table 3.3.2, Production of Granges and Tangerines in	• 74

3.4	Sugar Milling	77
	Table 3.4, Production of Sugar Cane in Vietnam	77
3.5	Oils and Fats Processing	80
	3.5.1 Soybeans	80
	Table 3.5.1, Vietnamese Production of Soybeans	81
	3.5.2 Peanuts	82
	Table 3.5.3, Peanut Production in Vietnam 1965-73	82
	3.5.3 Coconuts	83
	Table 3.5.3, Coconut Production in Vietnam, 1965-73	83
	3.5.4 Sesame Seeds	84
	3.5.5 Kapok	84
	3.5.6 Corn	85
	Table 3.5.6-A, Vietnamese Production of Corn	
	1965-73	85
	Table 3.5.6-B, Vietnamese Imports and Exports of	07
	Corn 1900-()	86
• •		86
3.0	Animal Feedstulls	87
3.7	Dairy Products	88
3.0	Meat Processing	89
3.9	Slaughter House	90
3.10		91
3.11	Livestock Ranching and Poultry Framing	92
3.12	Processing of Aquatic Products	93
	3.12.1 Fishing Operations	93
	Table 3.12-A, Data on Fishing Operations in Vietnam         1960-73	94
	Table 3.12-B, Fresh Aquatic Products of Vietnam, 1960-73	95
	Table 3.12-C, Tonnages of Processed Aquatic Products, 1960-73	<b>9</b> 6
	Table 3.12-D, Processed Fishery Products at Fishing Ports (1973)	97
	Table 3.12-E, Fresh Fishery Products at Fishing Ports (1973) •	97

3.12.2 Fish Culture
3.12.3 Cold Storage and Refrigeration ••••••••••••••••••••••••••••••••••••
3.12.4 Processing
3.12.5 Fish Meal ••••••••••••••••••••••••••••••••••••
3.12.6 Fish Sauce 100
3.13 Peanut Processing
3.14 Wood Pulp and Other Materials for Pulp 100
3.14.1 Mangrove Trees
3.14.2 Utilization of the Tram
3.15 Agricultural Implements and Engine Repair Services 102
3.15.1 Engine Repair Services
Table 3.15-A, Estimated Number of Diesel Engines Operating in the Delta by Equipment Application == 1974 (Civil Only)
$\mathbf{T}_{\mathbf{r}} = \mathbf{T}_{\mathbf{r}} + \mathbf{Crvir}  \mathbf{Carry} + \mathbf{Crvir} + C$
of Engines Operating in the Delta 1974 104
Table 3.15-C, Diesel Fuel Imports
Table 3.15-D, Estimated Cost and Waste of DieselFuel in the Delta 1974
3.15.2 Agricultural Implements
3.16 Forest Products
Table 3.16 Types of Timer Used in Cantho Oct. 1974 107
3.17 Gunny Bag Manufacture
Table 3.17Bag Manufacturing Capacity in VietnamOctober, 1974
3.18 Fertilizer and Limestone
3.18.1, Fertilizer Imports and Estimated Use, Calendar
Years 109
3.18.2 Urea Manufacture
). 10.) Phosphatic Fertilizers
3.18 / Location of Drinciple High Acid Soils (Man)
2 10 Les 100 Demon
3.19 LOWIIIT Pumps
Jezu Soap Manufacture 120
J. 21 Starch
of Manioc 1963-73 ···································

- 4 -

3.22 Other Industry Candidates	123
Table 3.22.1, Comparison of Conditions Req.ired for Cultivating Essential Oils	121
3.22.2 Mashrooms	125
3.23 Insecticides and Pesticides	125
Table 2.23, Vietnamese Pesticide and Herbicide Imports CY 1973	126
3.24 Seed Center	127

### ( 4. POTENTIALLY FEASIBLE INDUSTRIES (PRE-FEASIBILITY STUDIES)

4.1	Rice Milling ••••••••••••••••••••••••••••••••••••	129
	4.1.1 Adequacy of Raw Materials and Potential Surplus in the Delta	1 <b>29</b>
	Table 4.1-4, Large Rice Milling Tests Conducted in Vietnam by FAO Rice Study Team October 1974	1 30
	Table 4.1-B, Output Goals of Mekong Rice Mill, Saigon, September, 1974	131
	Table 4.1-C, Assumed Yields of Rice and Paddy By-products           in Delta	133
	Table 4.1-D, Consumption of Milled Rice per Capita,         150 kg/year	1 34
	Table 4.1-E, Per Capita Consumption of Milled Rice in Four         Selected Southeast Asian Countries 1972	135
	4.1.2 Integrated Receiving and Cleaning, Drying, Storage and Milling	137
	Table 4.1.2, Profitability Estimates on Rice Mill Complex         Proposed for Tay Do Industrial Park	141
4.2	Related Agricultural Programs	
4.3	Milling and Drying outside the Tay Do Park	144
	4.3.1 Milling	144
	4.3.2 Drying	144
4.4	Rice Bran Oil •••••••••••••••••••••••••••••••••••	145
	4.4.1 Background	145
	4.4.2 Market for Oil	145
	Table 4.4, Vietnamese Imperts of Refined Edible Oil Calendar and Fiscal Year Data Comparison, 1970-74	147
	4.4.3 Market for Defatted Bran	148

	4.4.4 Problems in Oil Extraction	149
	4.4.5 Outer Limits on Delta Production of Rice Bran Oil	149
	4.4.6 Raw Materials for Bran Oil Plant	150
	4.4.7 Processing Method	151
	Figure 4.4.7, Rice Bran Oil Processing	<b>15</b> 2
	4.4.8 Profitability of Bran Oil Plant	153
	Table 4.4.8, Annual Profitability Estimates on Rice Bran	
	Oil Plant	153
4.5	Nixed Feed Industry	156
	4.5.1 Demand	156
	4.5.2 Locally Available Raw Materials	157
	4.5.3 Ingredient Mix	158
	Table 4.5B, Suggested Feed Mix for Tay Do Plant	158
	4.5.4 Profitability Estimates for Mixed Feed Plant	159
	Table 4.5.4, Annual Profitability Estimates for Mixed Feed         Plant Proposed for Tay Do Industrial Park	160
	Table 4.5-A, Principle Ingredients Used in Mixed Feedsin Asian Countries	162
4.6	Engine Repair, Casting Shop, Agricultural Implement and Pump Manufacture	163
	4.6.1 Retionale for Facilities	163
	4.6.2 Three Facilities Proposed	163
	4.6.3 Doubtful Profitability in Starting Years	163
	4.6.4 Profitability Estimates for Third Year of	
	Operations	104
	Table 4.6.4, Profitability Estimates for Engine Repair, Machine Shop and Casting Shop in Third Year	
	of Operation	16 <b>5</b>
4.7	Fertilizer and Limestone	167
	4.7.1 Fertilizer	167
	Table 4.7.1, Profitability Estimates for Urea Plant	168
	4.7.2 Limestone Crushing	170
4.8	Seed Center	171
4.9	Exported Frozen Food Products Plant	173
	Table 4.9-A, Actual Sales Prices for Marine ProductsFrozen and Exported by Kien Kiang Co.,January-September, 1974	174
	Table 4.9-B, Annual Profitability Estimates on Frozen Food Products (River Fish, Duck, Frogs Legs)	1.80
		1/2

1

- 6 -

5•	<b>HELCEL</b>	TIES IN ESTABLISHING NEW AGRO-INDUSTRY	177
	5.1	Vietnam's Long Range Plans Related to Recommendations	177
	5.2	Summary of Recommendations and Time Scheduling Priorities	177
	5.3	Important Agro-Industries Excluded	179

6.	EXISTI	IG INDUSTRY AROUND CANTHO
	6.0	Existing Manufacturing or Processing Plants
	6.1	Integration with Proposed Tay Do Plants
		Table 6.0Manufacturing or Processing Establishments in Six Provinces Adjacent to Cantho182
7.	POLICI	S AND INCENTIVES TO ENCOURAGE AGRO-INDUSTRY
	7.0	National Goal
	7.1	Foreign Investment Policies
	7.2	Policies Regarding Domestic Investment
8.	EXECUTI	ION OF THE DEVELOPMENT STRATEGY
	8.0	Suggested Sequence of Actions
	8.1	Availability of Local Management for Future Plants at Tay Do 189

- 9. TERMS OF REFERENCE FOR FUTURE STUDIES

9.0 Purpose of this Chapter •••••••••••••••••••••••••••••••••••

9.1	Basic	Nodel for Feasibility Studies	•••191
	9.1.1	Rice Center (Receiving, Cleaning, Drying, Milling)	
		•••••••••••••••••••••••••••••	193
		Major Input Materials, II, 2	193
		Alternative Technologies, III, 3	193
		Locational Factors, II, 4	193
		Market, III, 1	193

- 7 -

#### Product Mix, III, 3..... p.194 Land Site Development, V, 1.1..... 195 Working Capital V, 2 ..... 195 Other Investments, V, 3..... 195 Pricing Policy, VII, 3..... 195 Planned Sales Organization, VII, 4..... 195 9.1.2 Rice Bran Oil Plant..... 196 Major Input Materials, II, 2..... 196 Alternative Technologies, II, 3..... 196 196 Capacity of the Proposed Plant, IV ..... 197 Pricing Policy, VII, 3..... 197 9.1.3 Mixed Feed Plant..... 197 Products, II, I..... 197 Major Input Materials, II, 2..... 197 Locational Factors, II, 4.... 198 Product Mix III, 3..... 198 Capacity of Proposed Plant IV, 1..... 198 Working CapitaI, V, 2..... 198 Planned Sales Organization, VII, 4..... 198 9.1.4 Engine Repair, Casting Shop, Agricultural Implement and 198 Pump Manufacturing Plant.....

#### Table 3.2 AVERAGE ANNUAL QUANTITY AND VALUE OF RICE IMPORTED BY CONTINENTS AND WORLD TOTAL, 1962-1967

	Products, II, 1	199
	Table 3.4 MARKET SHARES IN VIETNAM OF FOUR-WHEEL TRACTOR	
	IMPORTERS == 1974	199
	Input Materials, II, 2	2 <b>00</b>
	Alternative Technologies, II, 3	200
	Selection of Product Mix, III, 3	200
	Investment, V	2 <b>0</b> 0
	Implementation, X	200
	Survey Team	201
9.1.5	Fertilizer Plant (Urea)	201
	Land Site Development, V, 1.1	201
	Building, V, 1.2	201
	Machinery and Equipment, V, 1.3	201
	Working Capital, V, 2	202
	Inventories, V, 2.1	202
	Total Annual Expected Maximum Output, VII, 1	202
	Pricing Policy, VII, 3	<b>20</b> 2
	Financing Proposal, IX	202
	Implementation, X	202
	Research Team for the Feasibility Study	203

9.1.6	Limestone Crushing Plant	203
	General Comments	203
	Products, II, 1	203
	Major Input Materials, II, 2	204
	Locational Factors	204
	Market, III	204
	Capacity of Proposed Plnat, IV	204
	Working Capital, V, 2	204
	Suggested Research Team	204
9 <b>.</b> 1 <b>.7</b>	Export Frozen Food Products Plant	205
	General Comment	205
	Major Input Materials II, 2	20Ó
	Alternative Technologies Available, II, 3	206
	Locational Factors, II, 4	206
	Capacity of Proposed Plant, IV	206
	Conclusion on Frozen Food Plant for Export	207
	Composition of the Research Team	207
o 1 <b>A</b>	6	207
A*1*0	Seed Center	
	Suggested Composition of Research Team	308
9.1.9	Proposals for Industry Located Off the Tay Do Park	309

ſ

1

#### APPENDIXES

A. Statement of Work to be Performed by Contractor	211
B. Export Market Appendix	215
C. Characteristics of Soil and Water in the Vicinity of Cantho	230
D. Layout of Machine and Casting Shops on Tay Do Industrial Park (Preliminary Plan)	23 <b>5</b>
D-1. Machinery Suggested for Engine Repair, Machine and Casting Shops	236
E. Hammer Mill with 350/kg./hour Capacity	239
F. Bibliography	240
G. Feasibility Study Outline	247

#### INTRODUCTION

#### Origin of This Report

The concept of having an agro-industrial complex in the Makang Delta of Vietnam was discussed in 1972 and 1973 by representatives of the Indstrial Development Bank of Vietnam, the United Nations Development Program, and the United Nations Industrial Development Organisation.

A contract to prepare a master plan for the complex was awarded in September, 1974. The contractor selected was a consortium of consulting companies based in Tokyo. Field surveys were conducted in Saigon and the Mekong Delta in September - November, 1974 by a three-member team consisting of a techno-economist, an expert in soils and tropical agriculture, and a food processing engineer.

#### Subject of Report

This report analyzes the agro-industrial potential of the Mekong River Delta region around the city of Cantho. Special emphasis is placed on the potential of the Tay Do Industrial Park, a 151 hectare (373 acres) government-sponsored estate located 9 kilometers northwest of the city of Cantho (See Figure I=2).

#### Tay Do Industrial Park

The Tay Do Industrial Park is one of six industrial estates being sponsored by the Government of Vietnam through an agency called SONADEZI (Societé National du Développement des Zones Industriales). By far the most successful of the zones thus far, is the 511 hectare Bien-Hoa Industrial Park, 30 Kilometers north of Saigon where 90 manufacturing plants are located and 50 in operation as of mid-1974.

The Tay Do Industrial Park is only beginning to attract industry but is essentially complete. Land purchases for the Tay Do Park were started in 1968, surface preparation was completed in 1969, a water tower and well were completed by 1972, a power plant (33 MW) went on stream in December 1974, and only an expanded road and drainage system still remain to be completed in 1975.

The total expenditures, according to SONADEZI, from 1968 through 1975 will be 3.9 billion plasters (at the 1974, exchange rate of 685 plasters to US\$1.00, the investment in the park alone would be the equivalent of US\$5.7 million). Some of the plaster costs of the 33 MW thermal power plant are included in the total but the foreign exchange cost were not included since they were financed by the Japanese government as part of the aid program to Vietnam. (The power plant cost ¥3.2 billion or the equivalent of US\$10.7 million or \$324 per KW of installed capacity.) (See Bibliography Nos. (0, 20-A)

Despite the impressive progress to date, major capital investment, concessional land rentals, and encouragement from the Government, the Tay Do Industrial Park is not yet fully utilized. Of the total 151 hectares, 26.5 hectares is now occupied by completed plants or facilities. Another 19.8 hectares are under construction, or planned. The plants built or planned by the end of 1974 were as follows, covering 46.3 hectares or 31% of land available. (See Table I=2 on the next page.) -10 -

#### Teble I-2

#### PACILITIES BUILT OR PLANNED ON TAY DO INDUSTRIAL PANK

#### (as of Oct., 1974)

Staint.	Product or Service	<u>Hectores</u>	Capital- instion USEmillions	
Deilt	Puol eil and asphalt tanks (Base)	0,2	.04	••
м	Thormal power plant	10.0	10.7	
Ħ	Pier and va <b>rehouse</b>	15.0	R	••
Plannod	Carbonated beverages beer, and ice (B.G.I.)	5,0	6,2	
milt	Acetylene for welding (E)	0,6	0.7	••
11	Raw mgar	0.7	0.7	
Planned	Ganciene rotail stand	0.2	M . A .	••
	Commt Plant	10.0		••
**	Acatylons for volding II)	0.6	0.7 (est.)	
#	Ray signs	0.5	0,4	
Ħ	Nice mill (VIMCO)	3.5	0.6	•-
	Tutal	46,3	\$20.4	Capitalization total is not complete

Source: SCHABEZI, Cantho and HAZAMA Gumi Co., Tokyo ĺ

····



Source: SONADEZI Cantho

#### Purpose of This Report

This report is intended to identify inter-related and potentially viable industries to utilize local agricultural raw materials, and, in a few cases, to use imported materials for an interim period. Most of the industries described herein would be located on the Tay Do Industrial Park (see Appendix A for "Statement of Work").

In the case of certain industries, however, siting of plants outside the Tay Do Industrial Park is preferable. In a few cases, i.e. the marine products processing plants, siting near the sea coast is more economical and sometimes essential when it becomes necessary to have an assured source of fresh, high quality raw material.

#### Content of the Report

This report describes the setting of the Tay Do Industrial Park and the city of Cantho, the 16-province Mekong Delta Region, the population, labor force, principle products, and relationship of the Delta region to Vietnam as a whole (Chapter 2).

In accordance with the contract (See Appendix A), a series of 19 potential industries are evaluated as candidates for the Tay Do Industrial Park or other localities in the Cantho region (Chapter 3).

From the 19 candidate industries, plus others identified during the survey, the most feasible industries are described in terms of their potential output, capital requirements, and profitability (Chapter 4), and the time phasing and priorities are suggested (Chapter 5).

The existing agro-industries of the Cantho region are identified (Chapter 6) and policies and incentives recommended to encourage the development of industry and specifically further utilization of the Tay Do Industrial Park (Chapter 7).

The suggested method for executing the development strategy is outlined (Chapter 8) and further required feasibility studies are described (Chapter 9).

Appendixes include a bibliography, supporting statistical data, and exhibits to clarify the text of the report.

#### Methodology of the Report

The concept of preparing a master **plan** for the agro-industrial complex at Cantho was conceived during discussions in 1972 and 1973 between representatives of UNIDO, UNDP, and the Industrial Development Bank of Vietnam.

- 13 -

A consortium, consisting of U.S. and Japanese firms, was selected by UNIDO on May 16, 1974 to carry out the survey and to prepare the master plan. The two member firms of the consortium are: (1)Pacific **Projects, Ltd.**, an American-owned firm based in Tokyo and (2)Mitsui Consultants Co., a Japanese firm also of Tokyo. A contract (UNIDO No. 74/21) was signed on September 23, 1974 at the UNIDO headquarters in Vienna.

To conduct the survey in Vietnam, a project team was organized consisting of the following three persons:

<u>Frank L. Turner</u>, Senior Industrial Development Economist, with prior experience in industrial planning in the Ryukyu Islands, Korea, Taiwan, Japan, Ecuador, Peru, Chile, Bangladesh, Cameroon, and Nigeria. Most of Turner's studies were prepared for the Stanford Research Institute of Menlo Park, California.

<u>Tadayuki Aoike</u>, Soils and Tropical Agriculture Expert, with prior experience in Sumatra, Nepal, Indonesia, New Guinea, Iran, and the Philippines. Aoike has also made surveys of agricultural machinery.

<u>Susumu Tsuji</u>, Doctor of Agriculture, Food Processing Engineer, is Chairman of the Food Processing Engineers Assoication of Japan (affiliated with FIDIC). Tsuji's prior experience includes feasibility and planning studies for industry in Nepal, Sri Lanka, Taiwan, Korea, Philippines, Indonesia, Singapore, Thailand, Malaysia. Tsuji is author of a reference work entitled "Food Processing Guidebook" in Japanese.

Turner left Tokyo on Sept. 15, 1974 for Saigon, Cantho and Bangkok, Vienna, Washington D.C. and Menlo Park, California. The objective was to meet representatives of the UNDP, the Vietnam Industrial Development Bank and SONADEZI, in Saigon and Cantho; officials of the Mekong Secretariat in Bangkok; UNIDO in Vienna; and USAID and IBRD (World Bank) in Washington, D.C.

In Menlo Park, Calif., Turner met with persons familiar with the Stanford Research Institute study of land reform in Vietnam, and then returned to Tokyo on Oct. 1, 1974.

On Oct. 20, the three-man survey team left for Saigon and members of the team remained in Vietnam until November 29.

From Oct. 22 to Nov. 7, the team was based in Cantho in order to visit the Tay Do Industrial Park, agro-industrial plants, farms and to interview knowledgeable people in the local government, Cantho University (5 Japanese professors were in residence), and USAID.

The team travelled to the coastal town of Rach Gia to see fish meal processing and the freezing of marine products for export.

Turner spent Nov. 29 and 30 in Bangkok to inspect rice bran oil manufacturing plants in cooperation with FAO representatives.

From Dec. 1 to date, the survey team in cooperation with marketing specialists and engineers in Tokyo, analyzed data and prepared this report.

#### <u>Acknowledgements</u>

We would like to express our appreciation for the help extended to the project team by the following persons, listed in the sequence in which they were met: (1)Richard H. White, UNDP, Saigon; (2)Eric R. Adams, UNIDO, Saigon; (3)Pierre Sales, UNDP Saigon; (4)Khuong-Huu Dieu, Industrial Development Bank of Vietnam; (5)V. R. Pantalou, Mekong Secretariat, Bangkok; (6)Eric D. Manning, UNIDO, Vienna; (7)Andras Miklovicz, UNIDO Vienna; (8)David Roberts, UNIDO, Vienna; (9)D. F. Mant, UNIDO, Vienna, (10)Nguyen Huu Thu, Industrial Development Bank of Vietnam, Saigon; (11)Tran Thanh Tung, SONADEZI, Cantho (12)Luu Phong Loi, Industrial Development Bank of Vietnam, Cantho; (13)Toshio Mizunaga, Project Manager Hazama Gumi Field Office, Cantho; (14)Toshio Iga, General Manager, Hazama Gumi, Ltd., Tokyo; (15)N. V. Truong, University of Cantho; (16)Clayton C. Ingerson, USAID, Cantho.

H. B. Cushing and Wendell P. Knowles, Director and Deputy Director respectively of the USAID group in Cantho, furnished substantial data and introduced the team to staff members knowledgeable of conditions in the Delta region.

#### CHAPTER 1

#### EXECUTIVE SUMMARY

#### 1.1 Conclusions

• This report concludes that it will be feasible to establish an agro-industrial complex on the Tay Do Industrial Park near Can Tho, Phong Dimh Province, about 170 kilometers southwest of Saigon.

• The irrigation water of the Mekong River makes the Delta one of the most potentially productive paddy-growing regions of the world. The current 5.1 million tons of paddy can be substantially increased and premium rice exports and paddy by-product can offset Vietnam's trade deficit.

• The branches of the Makong have a high potential for the culture of fresh water fish that can be frezen for export.

• Most of the recommended facilities would serve the purpose of achieving better utilization of the Delta's agricultural, fisheries, and livestock raw materials or would help to augment the supply of such raw materials.

• The fixed capital requirement for the agro-industrial complex is estimated at \$260.3 million and the annual savings and earnings in foreign exchange are estimated at \$172.4 million. The duplication of recommended projects would achieve higher earnings later on.

#### 1.2 Agro-Industrial Projects Proposed

A key facility recommended would be a rice center having annual capacity of 37,000 tons of exportable premium rice. A related near-by facility would produce 960 tons of edible bran oil, 408 tons of industrial oil and 10,200 tons of defatted bran. The bran would help support a mixed feed plant for the hog, poultry and cultured fresh water fish industries.

To aupport the above facilities and to raise the Delta's productivity, other related plants or services are suggested: (a) a seed distribution center (b) engine repair plant, (c) casting shop, (d) implement and pump manufacturing plant, (e) urea plant (f) intermediate ammonia plant, (g) limestone crushing plant, (h) the culture and freezing of fresh water fish for export.

More regarding the proposed projects is shown in Table 1.2 on the next page.

## Table 1.2

•

# PROPOSED ACMO-INDUSTRIES FOR THE TAY DO INDUSTRIAL PARK AND VICINITY, TINING PRIORITIES, INTEDRELATIONSHIP, ODETS, SAVINGS AND ENFLOTMENT

Anomalia Tours	will declime by half with better maintenance Annuming multiple shifts fereign exchange savings could be \$179 million If exnerted and brown rich	imported, foreign exchange earnings would be 1/3 to 1/2	Assuss 22,000 tons imported from Thailand @1160/ton yearly	Atter Plant Fish operation, chicken and pork could also processed Assumes 104 mills and 10 drying	plants(10 dryers x 8 hours/day x 300 days yr x 9 tons paddy/hour -16,000 tons of paddy 350 kg/hr , x 30 mills x 8 hours/ day x 300 days.	
direct direct times) times) 15	31 335	136 34	5 8	6	97 97 96	1, 323
Ammuna I Production Capacity Salas of 1,000 tens/year of TRU variety seeda	forgines, services werth \$200,000/year 480,000 to 612,000 toms per year de- pendigg on Aift	)/, united rice multed rice plus 5, 100 tens bran 960 tens edible oil dustrial oil	51,000 tons of mixed feeds 200,000 tons crushed limestone	1,200 tons of frozen duck, fish or frogs legs	<ul> <li>I.I.I.d rice plus</li> <li>19,000 tons brar</li> <li>25,200 tons of</li> <li>crushed lime-</li> <li>stone</li> </ul>	
Ammal Savings or Larnings in Foreign Exchange (U.4 0001s) Effect on foreign exchange indirect through Plant C \$5,000 (estimated	savinge in fuel losses) \$140,145 (value of urea if jamertod )	\$18,627 (from export of total output) \$1,089 (import substitution)	\$5,414 (import substitution) Effect on foreign exchange indirect through C or other mills	\$2,100 (exports) Ear local	ror tocat consumption Effect on foreign exchange indirect through C and I	\$172,375
Capital Investment in Fixed Assets (US\$ 000's) \$30	1221 1221,717	<b>5</b> 4, 601 <b>56</b> 06	\$568 \$2,000	<b>W</b> 15		\$261,333
Marre Plant Bescription can be fotion in This Maport 4.8	4.6.2 4.7.1	41.2 447 448	4=5=3 4=5=4 4=7=2	6**	4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	-
Interrelationship between Plants and Pacilities Helps assure uniform paddy for Plant D Trector repair	and Props belp supply peody for Plant D Helpe assure paddy supply for plant D and other mills	Supplies bran to Plant E Supplies defat- ted bran to Plant F	Supplies feed to Plant H Improves paddy supply for Plant D	Cultures fish with feed from Plant F and defatted bran from Plant E	For local food su ply and bran for Plant E Helps assure padd for D, bran for E, and feed for H.	
Mar of Plant of Plant of Plant of Plantsion in Five-year Maral Discussion in Five-year Maral Second Canter No. 67) Second Canter (p. 127, 130) (p.127, 130)	Carting Sop, Agricultural Inglemental Urea Flatt and Intermediate Amonia Plant. (g.1)!)	Paddy Cleaning, Drying, Storage and Willing Center (p. 32) Rice Bran Oil (See p.255 of Source 68)	Mixed Feed (p.67, 131) Limestone Crush- ing on Tay Do Park (p.131)	Exported Frozen Food Products (p.43)	Milling and Drying of Paddy Outside Tay Do Park(p.126) Limestone Crushing outside Tay Do Park, 3(k) kg/hour milla (p.131)	
Timing Frity	(161•d)		111	VI	IV IV	ls
a > a	ο υ	<b>۵</b> ۵	<b>n</b> 0	Ŧ	<b>1</b>	Tota

Note: The last two proposed facilities, I and 1, would be located outside the Tay No Industrial Park.

,

#### 1.3 Location of the Proposed Agro-industrial Complex

The proposed location of the complex is 9 km NW of Cantho City on the Tay Do Industrial Park, consisting of 151 hectares of reclaimed land extending along the Bassac River, one of the major branches of the Mekong River.

Preparation of the estate from 1968 to 1975 cost the piaster equivalent of \$5.7 million. Infrastructure built includes an industrial water tower, roads, pier, warehouse and other facilities. Located on the estate is a 33 megawatt thermal power plant, built with Japanese aid at a cost of \$10.7 million. There are already 11 manufacturing plants, built or planned, occupying 46.3 hectares or 31% of the estate area.

#### 1.4 Setting of the Complex

Located near the provincial capital city of Cantho, the Tay Do Industrial Park is close to the commercial and transport hub of the Mekong Delta. The urban population of Cantho of 174,422 (1973) can provide a labor force of unskilled and vocationally trained manpower, and the University of Cantho can provide management personnel for some of the proposed plants.

The population of the Delta's 16 provinces is 7.4 million persons or about one third of South Vietnam's population. Growing at an annual average rate of 2.1%, population of the Delta should reach 9.1 million persons by 1985. The population density is moderate, 199 persons per square kilometer (as compared with 296 in Japan and 546 in Bangladesh).

#### 1.5 Agricultural and Fishery Resources of the Delta Region

The top ten agricultural resources in the Delta, as measured by production value in U.S. dollars are as follows (values are for 1973 production):

1.	Paddy\$652.9	million
2.	Hogs\$198.6	H1
3.	Oranges and tangerines\$ 91.5	H
ĺ.	Fresh water fishery products\$ 77.4	11
5.	Marine shrimp\$ 72.3	11
6.	Marine fishery products\$ 51.7	11
7.	Chickens\$ 40.8	11
8.	Duck\$ 32.3	11
9.	Sugar cane\$ 22.1	**
10.	Clams, in shell	**
	Subtotal\$1,259.8	11
	Others \$ 60.5	**
	Grand total	**

With the exception of marine products (shrimp, red snapper, clams, cuttlefish and fish meal), almost all the Delta's production is consumed within Viotnam. Of the 5.1 million tons of paddy produced in the Delta, about 1.3 million is surplus to the food needs of the Delta, and has traditionally been shipped to Saigon. Vietnam has slipped from being a major rice exporter to a rice importer. Peak exports were 346,000 tons in 1960 whereas in 1965 Vietnam began to import heavily, reaching peak imports of 559,000 tons in 1970.

The potential for Vietnam to resume exports is high. There are several actions that could cause major increases in the Delta's farm productivity and rice milling efficiency such as: higher applications of urea, phosphatic fertilizer, limestone; more deep plowing by tractors; further plantings of the "TN" varieties ("miracle rice" adapted for Vietnam from IRRI varieties); low lift pumping from canals and rivers during the dry season; and improvements in milling practices by using rubber rollers and disc hullers.

Given these improvements, productivity will increase substantially and lead to much higher yields. Improved milling can provide a valuable bran oil by-product not currently possible with inferior mills.

Some of the possible foreign exchange earnings and savings attainable in Vietnam are (see Chapter 7):

Potential value of Delta's exportable rice surplus (1.3 million tons @ \$480/ton)\$ 624 million
Correction of current milling losses from inferior milling practices (5% loss of head rice)
Potential edible oil production from rice bran \$ 38 million
Potential industrial oil production from rice bran
Total\$776 million

The above total is greater than South Vietnam's total imports of \$678 million in 1972 and \$717 million in 1973.

The principal goals of the proposed program are: (a) to improve paddy processing and cultivation, (b) recover bran and bran oil, (c) provide mixed feeds for animal husbandry and fish culture and (d) to export frozen river fish, duck and possibly frogs legs.

To execute the proposed projects, careful efforts will be required to develop lines of communication with the world's principal sources of financing, both private and public.

Demostic programs will be needed to assure future availability of uniform paddy and adequate supplies of ingredients for a blanced mixed feed for hogs, chickens and cultured river fish.

Managerial skills for some of the proposed plants can be found locally, but in the case of the urea plant, rice bran oil plant, and the rice center, foreign management may be temporarily required during the period of plant start-up.

(Executive summary ends here)

#### 1.6 Comments

The use of the word, "Vietnam" in this report refers to the territory from the 17th parallel to the southern tip of An Xuyen Province.

The plant cost calculations in Chapter 4 are based on the exchange rate of VN\$685 to U.S. \$1.00, the rate prevailing in December, 1974. There has been devaluation of the Vietnamese plaster. The authors believe that with the return of more normal economic conditions, local currency prices in relation to foreign exchange will tend to seek an equilibrium. In short, it is believed that local currency prices will decline as normalcy returns.

In the text of this report, references are made from time to time to source materials. These source materials are listed in the Bibliography, attached to this report as Appendix F, starting on page 241.

#### Chapter 2

BACKGROUND INFORMATION REGARDING THE DELTA REGION AND CANTHO

#### 2.0 Purpose of Chapter

This chapter provides basic demographic, geographical, soils, and other data relevant to the future planning of agro-industry in Cantho. (Some of this material is taken from Source 10 in the Bibliography and was written in Feb., 1974)

#### 2.1 Area and Population

The Mekong Delta consists of 16 provinces comprising almost all of Military Region IV (MR IV). The land area is 37,184 square kilometers (14,357 square miles) or about one tenth of the land surface in Japan (372,050 square kilometers). The area of the Delta is about twice that of the American state of New Jersey, is slightly larger than the area of Taiwan and one third the area of Bangladesh (See Tables 2.1 - a and b).

Population in the Delta is rising at an estimated annual average rate of 2.1%, slightly slower than the 2.6% growth rate assumed for all of Vietnam. (Bibliography Nos. 4 and 10.)

We estimate the Delta's 1975 population at 7.4 million persons, or 36% of the total Vietnam population of 20.8 million. We assume that population growth will continue at current rates. On this assumption, the Delta's population by 1985 will be 9.1 million and 34% of the total population of Vietnam of 26.9 million. These projections are tabulated below:

#### Table 2.1-a PROJECTION OF POPULATION OF VIETNAM AND THE DELTA (Unit: millions)

	Vietnam (@2.6% ann. avegrowth)	Delta (@2.1% ann. _ave. growth)	Percent of Vietnam Population in Delta
<b>197</b> 3	19.9 million	7.1 million	36%
1974	20.3 "	7.2 "	36%
1975	20.8 "	7•4 "	36%
1 <b>980</b>	23.7 "	8.2 "	36%
1985	26.9 "	9•1 "	34%

Population density of the Delta is moderate judging by the following comparisons. Phong Dimh Province is where Cantho is located and is one of the most populated provinces in the Delta.

#### Table 2.1-b

COMPARISON OF THE DELTA REGION POPULATION DENSITY WITH OTHER AREAS

Country or Region	Square <u>Kilometers</u>	Estimated 1975 Population in millions	Population per Square Kilometer
Delta Region	37,184	7.4 million	199
Phong Dinh Province	1,631	0.575 "	353
Rest of Vietnam (i.e. MR's I, II, III.)	133,756	13.4 "	100
All Vietnam	170,940	20.8 "	122
Ja <b>p</b> an	372,050	110.0 "	2 <b>96</b>
Bangladesh	142,776	78.0 "	546

Sources: Japan Almanac, 1974; Table 2.1; and calculations by Pacific Projects. .

Official Vietnamese Estimates of the 1975 the labor force for Vietnam as a whole show that 69% of the economically active population are engaged in agriculture, 6% in industry and 25% in services.

Since these figures include Saigon, Bien Hoa, Dalat, Hue and other urban areas, the percentage of persons engaged in industry and services is higher than in the predominantly agricultural Delta. On the basis of interviews, we estimate the labor force composition as shown in Table 2.1-c below. Farmers constitute some 80% of labor force.

#### Table 2.1-c

#### ESTIMATED BREAKDOWN OF THE DELTA CIVILIAN LABOR FORCE -- 1975 (Source: Estimated by Pacific Projects Ltd. from interviews and source No. 21)

(A)	Total Population of Delta	7,400,000
(B)	Labor Force (40% of A)	2,960,000
	(Active Population)	
(C)	Farmers (80% of B)	2,365,000
(D)	Fisherman (2% of B)	65,000
(E)	Commerce, Industry, Government	530,000
	(18% of B)	

The farm population is largely Vietnamese and thus homogeneous from a language and cultural point of view. About 7% of the Delta population are Cambodians most of whom are engaged in farming. The population of Chinese ethnic origin is small, estimated by the U.S. Consulate in Cantho as being 2% of the toal population or 180,000 persons. Almost none of the ethnic Chinese are farmers, although some are owners of chicken farms. (Persons of Chinese origin, although few in number, are very influential in commerce and industry; we estimate that 80% of the Delta's manufacturing facilities are owned by people of Chinese origin).

The Delta's farm population, when compared with the populations of other developing countries, measures up very well. The farmers are diligent, energetic, skillful and adaptable. Some examples of their adaptability are:

- Starting in 1968, the IRRI varieties of rice ("miracle") were introduced to the Delta. Production of IRRI rice rose at an annual average growth rate of 26%, from 1968 to 1973 (120,000 tons to 3,850,000 tons); hectarage rose even faster from 30,000 hectares in 1968 to 750,000 in 1973. (10)
- The mechanization of agriculture has proceeded rapidly (thanks in part to USAID policies of importing engines in the 2 to 20 HP ranges). The number of 4-wheel tractors imported to Vietnam rose from 484 in 1968 to an estimated 3,600 in 1973 (See Chapter 3); of these about 76% were destined for the Delta. Tractors have almost entirely replaced the traditional water buffalo in farming regions close to the roads.

- Farmors were quick to see the financial gains from growing sugar which gives them a yield of about 1.0 million plasters per hectare as compared with about 0.6 million plasters per pectare for two crops of paddy (4,000 kg/ha x June, 1974 price of 5.015 plasters plus Dec. 1973 price of 8,713 plasters 581,120 plasters for 2 rice crops. (Bibliography, No. 19.)
- The use of mechanical threshers (rotating drum type with about 4 HP gasoline engine) were introduced by USAID in 1969 and are now used almost universally as opposed to the traditional hand-pounding method of threshing.
- Hog and chicken breeds have been upgraded wherever farmers have supplies of feed.
- A system of fresh water fish culture -- probably unique to the Mekong -- has accelerated very rapidly. We found local estimates of the percentage of cultured fish in town markets running as high as 50%. (See Chapter 3.11)
- Farmers appreciate the importance of irrigation pumps in the dry season, January through March.

Despite the adaptability of the farmers to better methods, there is a fundamental problem of organization that will require careful handling later on when large scale industrial facilities become dependent on local raw materials. The farmers in the Delta are small-holders. About 79% of the agricultural land consists of farms of 1 to 10 hectares in size(\*) While this scale of Delta farming may seem better than Japanese or Korean samll-holder farming, it is far smaller than the plantation type of agriculture that characterizes the cultivation of rubber elsewhere in Vietnam, oil palm in Malaysia, corn in Thailand, cocoa in Cameroon, etc.

Therefore the cultivation and collection of raw material for future agro-industries will pose organizational and financial problems. The existing agro-industries in the Delta, such as rice and sugar milling have largely overcome raw material problems. However, future larger scale plants, discussed later in this report, will have to cope with the problem of organizing and financing numerous independent farmers to assure a steady supply of standardized raw material.

On balance, however, the homogeneity and human qualities of the Delta's farming community, the moderate density of population, and even the size of farms(measured by standards elsewhere in Asia) are all plus factors which we believe will contribute to continued and rising productivity of the region around Cantho.

(\*) From Bibliography, No. 45

• 24 •

#### Yette 2.1-4

#### MARA STATISTICS

BELTA BELTA POUTA V DESIMA		197)		1928 (Section 1928)		88C 81078 1973	
		6,700 18,718	6,700,763 18,718,710		7,001,340 19,336,397		
	tucils Newsys C 37 385 L 3CTS L GBS Ets	•	14 3 95 798 ,297	۹.	16 3 95 799 386	10 99 06 4,34	
					VELLAGE		
۱.		636,416	2,004	•	115	620	
2.		649,383	1,900	٩	36	254	
3.	Dam Turns	688,430	1,997	7	10	478	
4.	CILLY DOC	636,678	2,075	5	57	270	
5.	V BHH LONG	594,842	1,705	7	65	200	
●.	FUERS D'STA	<b>363 , 63</b> 3	1,691	7	50	367	
7.	K 38H 6 3AH6	510,000	5,266	•	47	233	
₿.		408,869	2,530	۲	54	326	
э.	K SAN FURIES	432,000	2,303	٠	51	210	
10.		498,198	2,363	7	56	410	
11.	EAC LINE	366,072	2,599	٩	19	194	
12.	<b>10. 10.</b>	314,027		٠	36	129	
13.		279,143	5,009	۲	24	179	
₩.	Cupous This	306,999	2,298	5	34	160	
<b>15.</b>	<b>eo cene</b>	211,907	570	٩	32	260	
۱6.	K28H 798H6		2,400.	٤		<b></b>	
	to take	7,242,243	<b>57,264</b> (24,397 49.	95 -100)	802	4,340	

Sources V.J. Consult to Ameral, Conthe, Stong Bink Province, Vietnas

-----





Note: The digits appearing in the provinces relate to the digits at the left of Table 2.1-d; see the preceeding page.







VERIONARI AND TERE REPORT DELETA

Burges Industrial Development Bank of Vietnam

#### 2.2 Soil and Water Characteristics

Our survey group analyzed Delta soils and water in order (a) to determine the potential for growing certain crops; (b) to assess the need for agricultural support industries such as urea manufacture and limestone crushing; (c) to help reach conclusions on the best utilization of rice straw and husk; and (d) to determine the need for mechanization in agriculture.

To accomplish the testing, one of the survey members (Aoike) took instruments to the Delta and tested the soil and water in six localities in the outskirts of Cantho and 1 near the high acidity zone near Kien Tan on the road (LTL 80) to Rach Gia.

Beginning in May and continuing through December, most of the Delta is flooded to a depth of 40 to 80 cm and in some places as deep as 3 meters. From January to May, the soil dries out and becomes "brick" hard. The flood means that the minerals and other properties of both the soil and water have an influence on the suitability of the region for certain crops.

(\*) The U.S. Department of the Army (in Source 35) describes the soils as follows:

"Added to the water variations are problems resulting from heavy soil texture, high acidity, inadequate internal and surface drainage, poor surface tilth, slow permeability, and rather low fertility. The problems of poor tilth, crusting, soil hardening and cracking when dry and slow permeability can be ameliorated somewhat with the incorporation of organic residue. To improve the soil, straw and stbble should, as a rule, not be burned unless it immedes tillage. Organic matter can also be added by plowing under cover crops, applying barnyard manure, and growing perennial grasses in rotation with other crops." (\*)

Our water and soil analysis resulted in the following findings:

• There is no salinity problem in the vicinity of Cantho, although brackish water in the lower Delta is a problem.

• The pH value in the water was 7.3 which helps offset the soil acidity of 4.3 to 5.0 in our texts of moist samples. (Department of the Army tests on moist soils ranged from 4.2 to a high of 7.0. The neutral pH value of 7.0, however, was found in 5 out of 57 tests only. In the 57 tests, the most frequently found pH values were in the 4 and 5 range. The tests of dry soil specimen s were invariably more acid by as much as 24%, suggesting the beneficial effect of the irrigation water.)

(\*) See Appendix F, Bibliography Source No. 35

• The soils are extremely dense, lacking in organic material, poorly ventilated, and lacking in bacterial action.

From our tests, we concluded that the application of limestone is important for most of the soils, to offset acidity. Although further analysis is needed, we estimated that from 300 to 500 kilograms of limestone per ha. should be applied in most areas to improve productivity. (According to the "Wildman" report (source 45) (\*), acidity of Delta soils in the immediate vicinity of the Makong streams tends to be lower than the acidity of soils located further away from the streams).

Our tests also showed the non-suitability of soils for soybeans or papaya and banana or corn or vegetables having long roots. This was further confirmed by observing the stunted appearance of coconut and papaya which contrast with the very large trees that grow in other tropical areas.

Deep plowing by mechanized methods is important in raising the productivity of soils for all types of agriculture. As a soil conditioner, mulch consisting of husk and straw should be plowed into the soil. This will improve ventilation of the roots, stimulate bacterial action, and improve the absorption of phosphorous.

Phosphatic and ammonia-based fertilizers (e.g. urea) are important to offset the deficiency of elements in the soil.

Aside from the diligence of farmers, the traditional productivity of paddy in the Delta is partly accounted for by the properties of the Mekong irrigation water which compensate for deficiencies in the soil.

These findings are described more fully in Appendix C.

(\*) See Bibliography, Appendix F

#### 2.3 Production of Agricultural, Figheries Products and Livestock

The value of crops, figheries products and livestock produced in the Delta's 16 provinces is on the order of U.S.\$1.3 billion,Vietnam's GNP in 1973 was US\$3.1 billion (1973 dollars from Source No. 24,Appendix F) before prices rose sharply in 1974. We estimate that the Delta's production of the above products would account for about one third of Vietnam's GNP in 1974.

Consistent documentation of values from product to product is difficult because some of the prices we have obtained are based on our own field investigation in the fall of 1974 and a few are December 1973 prices taken from Ministry of Agriculture records (Source 19, Appendix F). The result of the analysis of production is shown in Table 2.3 and summarized below.

The top ranking product in terms of market value is, of course, paddy, valued at US\$652.9 million (in terms of October 1974 prices at mills in Phong Dinh Province); paddy thus accounts for about half the Delta's output of products listed in Table 2.3.

The second ranking product is hogs, valued at \$198.6 million; the third ranking product is oranges and tangerines (193,000 tons), valued at \$91.5 million; the fourth ranking product is fresh water fish, valued at \$77.4 million, followed by marine shrimp valued at \$72.3 million. The market value of the ten leading products is summarized below from Table 2.3:

Rank in Terms of Value Produced in 16 Delta Provinces	Product	Market Value of Production (U.S. Dollars)	Date of Pricing
l st	Paddy	\$652.9 million	(10/74)
2 nd	Hogs	\$198.6 million	(10/74)
3 rd	Oranges and Tangerines	\$ 91.5 million	(12/73)
4 th	Fresh Water Fish	\$ 77.4 million	(10/74)
5 th	Marine Shrimp	\$ 72.3 million	(10/74)
5 <u>th</u>	Salt Water Fish	\$ 51.7 million	(10/74)
7 +h	Chickens	\$ 40.8 million	(10/74)
8 <u>th</u>	Duck	\$ 32.3 million	(10/74)
9 th	Sugar Cane	\$ 22.1 million	(10/74)
)0 th	Cleme. in shell	\$ 20.2 million	(1974 est.)
Su	b-Total:	\$1,259.8 million 60.5 million	(95.4°) (-4.6°)
01 Te	ners : otal :	\$1,320.3 #illion	(100.0%)

Not listed in Table 2.3 is an important crop, sorghum, which has been vigorously encouraged by USAID. Sorghum can become a substitute for imported corn and used as an ingredient in mixed feeds. Sorghum is grown in the region where floating rice is found. (Floating rice has a tall stalk which keeps the grain above the water level during the flood season.) The sorghum potential is high in five provinces near Cantho: (1)An Giang, (2) Chau Doc, (3) Kien Giang (4) Kien Phong and (5) Sa Dec. There is some potential in Phong Dinh as well. (see Figure 2.1-a). As flood waters recede in in October (see Figure 2.3), the moisture content of the soil is ideal, according to USAID, for the planting of sorghum. Harvesting is completed in December, well before planting the next crop of floating rice.

USAID representatives in Cantho believe that the sorghum program will result in substantial growth of hectares planted as follows:

Crop Year	Hectares Planted	Production (MT)	Yield/Hectare (MT)	
1970-71 (est.)	200	300	1.5	
1971-72 (est.)	6,000	9,000	1.5	
1972-73	17,900	21,000	1.2	
1974-75 (est.)	20,000	26,000	1.3	

The total area now devoted to floating rice is 500,000 hectares, suggesting a very large potential for sorghum. The current yields of 1.3 to 1.5 tons of sorghum per hectare are small, about one third the potential, given more seed and fertilizer. (Assuming the hectarage were to be 100,000, as opposed to the current 20,000, and if proper yields were obtained, the total production could be 500,000 tons, valued at Feb. 1975 prices F.O.B. U.S. Gulf at \$111/MT or \$55.5 million.)

Despite the generally high productivity of agriculture and fishing in the Delta, there are problems such as: (a) salinity in the lower reaches of the Mekong branches, (b) serious underutilization of land in the dry season (December into April), (c) silting of canals used to ship produce, (d) egregiously poor drying, storage of and milling of paddy, (e) shortages of animal feed, fertilizer and limestone, (f) underutilization of farm tractors owing to inadequate maintenance, (g) security problems along the coast that inhibit fishing, (h) shortages of good seeds (i) fallow land caused by the undermaintenance of irrigation canals, (j) excessive flooding in Long Xuen Province and elsewhere.

- 31 -



# WATER LEVELS AT CHAUDOC

(This figure aloue the resulting of when in the region where floating rise to green. Buill conditions are idenlity emited for growing sorghum, shorting in finishing and emiting in incontere.)


## Teble 2.)

# anda prosesso of namedia anticheral do francia frontas. Livienta do public de su da su da su da su da su da su Provens of 15 milita antichera 1973, bullito de factor parto de sultante of 1.1. publica

# (heren: Olean A and ) are the free forme 39. Mintle & Articliani Multicliani Maria Jawa Jawa 1994. The state in column 4 and the free for how that the Contractional Multicliani Maria Jawa 1994.

		act. 1976 (State auroy), price		1 11vv pic unide 100 No.1 Reute is Nuc Thank breaking Fame, Can The, Oct. 1976.	imerup muit prios of Jrugas. Rugariana, source 19	interes 19	source 19	angt. 1976, Cantho (fiaid auroug) 10/76	00 Planters per Kg., 1 chlohm = 1.5 Kg: source is Plaid survey is Cantho. 10/76.	00 Piasters per Kg., I dmok = L-5 Kg: source is field aurony in Canthe, 10/74.	acific Projects estimate
<b>.</b>		(USE =(1) (	::	9.9 9.9	\$ 94.5	•	• 2.3	2.12	e. G	€. 2.	<b>•</b> 52.1
Take at		<b>980''4</b>	::	73.000	355.000 (mt.)	000,007	1,270,000	000'06'1	1.350	1.200	120.000 (t.)
and in the	\$	. #42	(11) (11) (11) (11) (11) (11) (11) (11)	ł	3	i k	3	<b>N</b>	19	<b>B</b>	¥
	3	5,141,200	(2,096,000) (3,105,000)	1.864,000	192.000	7.78	<b>39.6</b> 2	78.657	80, 60, 8	18,459,100	126.400
	344	7,005,100	(3,190,200) (3,834,900)	3,776.600	2 <b>0</b> 5, 000	94.36	62,0 <b>0</b> 5	906,537	006.455.66	0008, 1770, 65	986 ° 623
I	i			1	¥					Head	¥
	ł	1	<b>]</b> []	}						1	j }
	ijii	i		Ň		•	*	s.		<b>e</b>	*

(Cont'd next page)

----

ť

	stimuted by Pacific Projects	1 2 1	alue is sverage for carrots and abbage in Passe Much Province a Dec. 1973, (Source No. 19)	scific Projects estimate	1 2 1	dve animal weight, 100 Kg. outs 32,800 Flasters in New Lah, Dec. 1973, 1 cow = 300 Kg est.), Source No. 19	ive animal weight, 100 Mg. cets 30,500 Plasters in Phong inh, Dec. 1973, 1 buffaln = 00 Mg (est.), Source No. 19	cource 19, umit price PL estimate	973, Dec. Source No. 19 . Phong Disk)	1 8 1	1
(D) Values of Belta Production (Convertion U.S.4 0 VM 605 to UBEL.00) (B ± C)	US <b>\$ millions)</b> 820.2 E	\$ 4.44	<b>8</b> 15-6	<b>\$12.</b> 6	<b>\$</b> 10.6	۲۵۵۴ ۲۰ ۴	* *	6. 2. 9. 2.	<b>\$</b> 2.2	<b>\$</b> 1.9	8 1.1
(C) Value/Unit	50.00 ()	000,03	108,000 (est.)	40,000 (est.)	55,000	<b>9</b> 6	6005.16	70.000 (ast.)	120,000	335,000	365,000
×.	<b>1</b> 22	1.3%	Ŕ	and the second se	ş	392 25	X	8	1	<b>X</b> 45	19
Dalts Pro-	27.600	50,600	021.66	216,100	132,340	040° TK	<b>23</b> ,690	28,355	12,405	3.845	5,000
(a) Martin Prostition	<b>36</b> ,710	002.625	<del>3</del> 57, <b>600</b>	258,000	279,800	119,000	<b>986</b> , 43	966. <del>1</del> 6	50,500	10, 300	10,600
Beit		t	t	¥	t	Ilead	Head	<b>£</b>	4		Ŀ. <b>¥</b>
Ĩ			Vegetables		Summer Potato	Cattle	Buffalo	Piereples	Corn, Mi te		Soybeans
	10.	. [[	1 2	13.	4	15.	16.	17.	18.	19.	8

(Cont'd next page)

Table 2.3 (Cent\*d)

- 34 -

( <sup>2</sup> )	Parts all years	1	1	
(a) Malue of Jule Protection (Convertion) (Convertion (Convertion (Convertion) (Convertion (Convertion) (Co	(US\$ millions) BluO	1	eldigilges	<b>\$1</b> , 320, 3
(C) Valent La Plater		1	000*056	ta Productions
₹ <b>A</b>	ž	<b>K</b> 1	1	Total Wel
	RL', 93	3,700	R	
(T) Martin Protection	88°81	ant, ot	000° **	
land t		<b>N</b>	8	
į		Teleoco	jł	
	<b></b>	શ્રં	<b>3</b> .	

Table 2.3

\$1,320.3 million = WHS you billion(@ WHSOB5 = \$1.000.)

Total for Available Itams produced in the lelter

- 35 -

### 2.4 Transportation

### 2.4.1 Highway and Rail

There are good and bad features about the transportation serving the Delta and the Cantho District. The roads are in general good to excellent thanks to the priority placed on their construction and repair by the MR-4 Corps of Engineers.

The road to Saigon (about 170 kilometers) is designated as QL-4. (The numeral, 4, denotes the 4th Military Region and "QL" is an abbreviation for the Vietnamese word for Military Region). This road runs in a northeasterly direction from Cantho and goes through Vinh Long to Saigon. There are no bridges across the Bassac and the Mekong and this entails delays for buses and trucks so that the trip can take 4 to 6 hours. The same road proceeds southwest as far as Bac Lieu.

Another good road, Interprovincial Road "LTL 27" is also a wide twolane highway and passes by the Tay Do Industrial Estate. Other first class roads go northwest to Sadec and southwest to Vinh Binh.

In short, the road aystem is quite adequate to permit the hauling of produce from various parts of the Delta to the Tay Do Industrial Park and for hauling goods to and from Saigon.

There is, however, no rail transportation connecting the Delta with Saigon or any other destinations.

### 2.4.2 Waterways

Starting in the early 1900's, the French administration built a network of canals throughout the Delta. As seen from the air, these canals look like an angular, criss-cross network. From the maps, it appears that the canals are spaced parallel to each other at intervals of about 10 kilometers and in some places as close as 5 kilometers. (In Phong Dinh Province where Cantho is located, there are three major canal systems, the O Mon-Co Do, the Vi Thanh-Can Tho, and the Phung Hiep -- dug in the 1920's.)

These canals, in conjunction with the branches of the Mekong, constitute a low-cost waterway for transportation by tug and barge. (Truck transport, Cantho to Saigon, costs 4,000 piasters per ton but barge transport costs 2,400 piasters, 40% less than by truck).

However, large, ocean-going bulk carriers cannot enter the Delta ports. We found different opinions on prospects for a future port in the Delta; some thought that Cantho could be built for 5,000 GT ships, others preferred a location 30 KM northeast of the city of Khanh Hung, capital of Baxuen Province, where perhaps 5,000 GT vessels could discharge, but only after major dredging. The recent report (<u>Rehabilitation</u> of the <u>Inland</u> <u>Waterway</u> <u>Fleet</u> in the <u>Republic</u> of <u>Vietnam</u> and the <u>Khmer</u> <u>Republic</u>, UN/ECAFE Commission for <u>Coordination</u> of Investigations of the Lower Mekong Basin, May 1974) recommends the future development of Vinh Long on the Mekong River.

Figure 2.4 shows the many shallow points in the approaches to the Bassac and Mekong. Other depths (including Saigon for comparison) are tabulated below:

à

	Depth at High Tide (Meters)	Depth at Low Tide (Meters)
Nouth of Bassac River	6.5	2.7
Mouth of Mekong River	3.3	1.8
Cantho	4.3	3.0
Rach Gia (fishing port)	2.2	1.8
Tay Do Industrial Park Pier	4.3	3 (estimated by Tai Do engineer)
Saigon	9.1	8.2

The water depths in the mouth of the Bassac, i.e. the approach to Cantho is enough to accommodate a 500 gross ton coastal vessel having 3.72 draught at high tide only.

Dredging costs worldwide are often calculated at US\$1.50 per  $M^3$ . Costs in Vietnam are about the same, 1,000 piasters, according to Ministry of Public Works in Saigon. There is an unfulfilled dredging backlog of 50  $M^3$  annually for all of Vietnam including the Delta. Saigon is now being given priority. Only by spending U.S. \$75 million annually (50 million  $M^3 \times $1.50/M^3$ ) for several years, it might be possible to assure the Delta of navigable waterways for vessels larger than 500 gross tons.

This means that the traditional towed barge transport system will probably have to continue for some time. This is not necessarily inefficient because a single tug can tow a chain of three or four barges through the waterways. (We saw barges loaded with roasted lime stone, i.e. clinker, being carried from Nha Com, the specialized loading port near Hatien, to the Bien Hoa Industrial Park.

Cantho is served by two airports located just off LTL 27, the interprovincial road to Long Xuyen, which runs past the Tay Do Industrial Park. Piston aircraft are using the fields, and expansion would be required to accommodate jet transports.



- 18 -

### 2.5 Industry

Industry in Canthe will be discussed in greater depth in Chapter 6 later in this report. In general, industry is of the artisanal type, manned by family members. However there are some substantial industrial facilities and public utilities. The major facilities are:

(a) The 33 Mogawatt thermal power plant on the Tay Do Industrial Park.

(b) B.G.I. (Brasseries et Glacières de l'Indochine), bottlers of carbonated beverages and boor and ice manufacturers.

(c) Tay Bo Mung Hoang, a bakery making bread, cookies (i.e.biscuit ) and moodles.

(d) A 4-ten per hour medium size limestone crushing plant, located 40 km northwest of Cantho, close to the junction of LTL 27 and LTL 8, named Than Nong Phanbon (meaning, Miracle Rice Fertilizer) is being built.

(e) Four fishmul plants are located in Nach Gia (HATICO, SUMACO, FITECO, and INCOMAP).

(f) Preezing plant (Kien Giang Co.) for exporting marine products.

(g) Sugar mills making unrefined sugar.

Unfortunately, some of the plants are operating below capacity because most of them sould to serve local markets where demand is down owing to inflation or because of a shortage of raw materials. The city of Cantho is well served with potable water, thanks to an Australiafinanced storage tower and pumping system capable of supplying 7.2 million yallons, a large enough volume to meet the Cantho City needs until 1985.

Aside from agricultural raw materials for industy, the area has limestone reserves at Ha Tien, now being worked to supply clinker to a comment plant in Him Hoa Industrial Park, as mentioned.

A second raw material is a post layer reported to be 1 to 2 meters in thickness, located in An Xuon and Kien Giang provinces. (See Figure 2.1-a above.) Security in the peat area has been poor for several years and we wore not able to learn much except that briquettes are made and u-ed for fuel by the local people.

Wood and pulp resources are discussed in Chapter 3, Sections 3,14,1 and 2 later in this report. In general there is little prospect for industrial uses of the wood resources and the possibilities of using rice straw or hush are, we believe, likewise doubtful.

### 2.6 Suitability of Cantho for Industry

As in any developing area, the Cantho region lacks the industrial support base usually associated with modern industry. There are no foundries for example, and castings are usually made in Saigon. However such impediments have been overcome elsewhered and could be more readily overcome in Cantho than in many developing countries.

Although there were labor shortages in 1974 due to the military conscription, Cantho's urban population can provide a labor pool. As of September 30, 1973, the population of Cantho City was 174,422, making it the largest town in the Delta. Of this total, about 70,000 persons are of working age.

Because of the war and the recent-year imports of farm machinery and vehicles, there are mechanical aptitudes, innovative skills, and adaptability in the labor force. Unlike some other predominantly farming areas in developing countries, it will be possible to recruit persons with manufacturing skills in Cantho.

The University of Cantho, the only one in the Delta, offers a curriculum in business subjects. Most young men with managerial qualifications residing in places like Saigon or Dalat are reluctant to move to Cantho, but the local university in Cantho can help in meeting this manpower requirement.

Because of the roads, canals, and airfields, Cantho has become the Delta hub for commerce. There is noticeable growth in the population and business activity and several new office buildings have been erected despite the economically depressing withdrawl of U.S. forces following the ceasefire (Jan. 28, 1973). Wages in Cantho compare very favorably with those in other localities as tabulated below for the autumn of 1974:

	COMPARISON OF MONTHLY WAGE RATES Sept., 1974 (Unit: U.S. Dollars)						
	Skilled Workman (electrician, mochanic_etc.)	Unskilled Labor	Remarks				
Cantho	<b>\$</b> 29 <b>- \$</b> 44	\$14 - \$18	Converted at VN\$685=\$1.00				
Thailand	\$105	\$ 33	Plus bonus of 1 mo, pay annually				
Malaysia	\$109	\$ 38	Plus b <b>onu</b> s of 1 mo. pay annually				
Philippines	\$143	\$ 77	Plus bonus of 1 Mo. pay annually				
Korea	\$160	\$100					
Singapore	<b>\$</b> 218 <b>- \$261</b>	\$ 55	Plus bonus of $4$ mos, pay annually				
Ta i wan	\$300	\$120	Bonus included in wage figures				
Australia	\$384	\$346					

Cantho does not compare favorably with other locations with regard to electric power costs. This is because power has been supplied by diesel generators. In Cantho, the installed capacity was 10 MW of diesel capacity generators in Oct. 1974. Now, however, since the 33 MW plant at Tay Do has gone on stream, the total is 43 MW. By contrast, the peak load (night-time) is only 7 MW. This means that surplus power will be available for many years until the industrial load builds up.

COMPARISON OF POWER RATES IN CANTHO WITH RATES ELSEWHERE -- OCT. 1974 (Unit: U.S. cents per KWH)

Korea	1.0 ¢
Singapore	1.96¢
Japan	3.07e
Australia	3.46e
Philippines9	3.54¢
Can <b>tho</b>	5.4 ¢ (According to Vietnam Power Co., the charges would decline after Tay Do 33 MW went on stream in Dec. 1974.)

Source: Pacific Projects, Ltd. survey for commercial client

- 41 -

The costs of building a steel frame factory building in Cantho are lower than most other localities as tabulated below:

COMPARATIVE COSTS PER SQUARE METER OF BUILDING STEEL FRAME CONCRETE SLAB FACTORY BUILDINGS -- Oct. 1974

(Unit: U.S.\$ per M<sup>2</sup>)

Cantho	\$102 (70,000 plaster)
Ja pan	\$216 - \$300
Singapere	\$150 - \$250
Philippines	\$123 - \$150
Australia	\$127
Korea	\$ 55

Source: Pacific Projects, Ltd. survey for commercial client

The cost of leasing industrial land at Tay Do appears quite reasonable when compared with other industrial parks as tabulated below:

COMPARISON OF LEASING COSTS PER SQUARE METER OF LAND ON INDUSTRIAL ESTATES, 1974 (FOR CANTHO AND SINGAPORE), 1975 FOR PHILLIPPINES

> (Unit: US\$ per M<sup>2</sup>) Singapore (Jurong) \$0.72 Philippines (Export \$0.115 (after June 1975) processing Zone)

Cantho (Tay Do) 50.30 - 0.90

### Chapter 3

### CANDIDATE INDUSTRIES

### 3.0 Content of This Chapter

The contract on which this report is based contains a statement of work attached as Appendix A. The statement requires that we review the possible suitability of 19 candidate industries. We are commenting on these candidates below plus others that have occurred to us since the contract statement was prepared.

### 3.1 Inter-relationship of Agro-industries

In the developed countries, there is a close inter-relationship between industries using agricultural raw materials. This means that almost total utilization of by-products takes place. In the Delta, full utilization will take many years, but a near future start can be made by concentrating on a few key industries.

Figure 3.1 shows the theoretical inter-relationship of agroindustries. Some of the industries shown are not suitable for the Delta and some will take years to develop. We believe the nearterm priorities should be placed on three basic industries plus the raising of poultry, hogs and fish culture:

- Rice milling including storage and drying of paddy
- Oil extraction from rice bran
- Feed Mixing
- Hog, and poultry raising and fish culture

These activities will take advantage of the Delta's paddy, marine products, and the potential for animal husbandry and fresh water fish culture.

### 3.2 Paddy and By-product Processing

The future of paddy processing (i.e. processing of unhusked paddy into rice and bran) in the Cantho area is promising. The foreign demand is high and rising since rice is the staple food from Japan westward to Iran and is a supplementary staple in other countries.

In 1971, 189 million metric tons of rice were consumed in the rice-eating countries from Japan to Iran and was valued at \$29 billion or \$153 per ton. Bibliography Source No. 39.

Since 1971, we estimate the tonnage consumed in the same area has risen by nearly 10% to about 208 million tons having a value ranging from about \$42 billion to \$83 billion or \$200 to \$500 per ton, depending on grade.

Some illustrations of the per ton value of rice moving in international trade are discussed after Figure 3.1:



### Drying and Storage

As the production of paddy in the Delta rimes, more will be shipped to other parts of Vietnam as well as overseas. As a precaution against bad-crop years in the future, larger volumes of rice will be placed in storage. In preparation for expanding production and a greater need for storage, improvements in drying and storage will be essential. The traditional methods of drying, storage and milling are inefficient and costly in terms of the high percentage of brokens and the loss of by-products (Bibliography No. 28).

In the Delta, farmers are now raising 2 or 3 paddy crops per year. When only one crop was grown annually, harvesting was traditionally undertaken in the dry season. With the introduction of multiple rice crops, however, some crops are harvested during the wet season. This means that the farmers need drying facilities for their wet-season harvests. Figure 3.2-B shows that harvesting of the TN variety first crop occurs in September, a month of peak rainfall in the Delta (see Table 3.2-C). For the increase in TN production, see Figure 3.2-A(Bibliography No. 28).

### Paddy Separation

Many paddy separation methods are being practiced in the Delta but most of them are inefficient and uneconomical. These methods should be improved(Bibliography No. 40).

### Rice Milling

Rice milling has to be viewed from two aspects that relate to the distribution of the milled rice. Most of the milling of paddy occurs in the villages where the farmer pays the miller a fee and then takes the milled rice home for his own consumption. The traditional mills are mostly involved in this type of distribution. Large scale mills, if built to supplant the small village mills, would result in many problems of distribution. (Many of these traditional small scale mills are of the "kiskisan" or Engelberg type hullers which abrade the grain and cause a 15% - 20% loss and increase the number of brokens (\*) This type of mill is being replaced slowly by more efficient small scale mills.

A second type of milling is on a larger in scale, but mills are small by world standards having daily capacity of over 20 paddy tons per day. These mills perform the function of milling rice for shipment to urban areas in the north, chiefly Saigon, Although more efficient than the village mills, the larger mills nevertheless are inefficient, in terms of the loss of by-product, the failure to separate husk from bran, the high percentage of brokens and excessive power consumption (Bibliography No. 40).

(\*) Bibliography, Source No. 40

There are about 1,600 rice mills in the Delta. The Phong Dinh Province (where Cantho is located) Inspector of Industry (i.e. local representative of the Ministry of Commerce and Industry) told us that these 1,600 mills are broadly classified as "big" and "small" mills. There are 145 big mills which typically have a daily capacity of 20 tons of paddy or more. They work more than one shift during the harvest season, and for this analysis, we are assuming that these big mills processed about 870,000 tons of paddy and that the small mills processed about 4,271,000 tons of paddy in 1973. The total paddy processed was 5,141,000 tons in 1973 (see Table 2.3 earlier in this report ).

The big mills are located in five Delta provinces as **shown** below:

Province	Number of Mills	Estimated Tonnage of Paddy Processed
Ba Xuen	40	244 <b>,000 MT</b>
Phong Dinh	30	183,000 "
Bac Lieu	30	183,000 "
Kien Giang	25	148,000 "
An Giang	20	112,000 "
Total	145	870,000 MT

Location and Estimated Paddy Capacities of Major Rice Mills in the Delta -- 1973

> Source: Inspector for Ministry of Commerce and Industry, Phong Dinh Province

These types of mills will be the most important suppliers for Vietnam's urban areas and eventually for export. Therefore the milling efficiency should be raised not only to produce a higher quality of rice but to obtain the maximum yield of bran  $(\circ)$ .

The milling capacity for the future will be determined in large part by the availability of storage capacity, bran defatting capacity (i.e. oil extraction) and the volume of paddy that can be readily collected from the surrounding farms. (\*).

(\*) Bibliography, No. 40

### Japanese Imports of Polished Rice, 1973 and 1974 (SITC No. 042-200)

	Tonnage I <b>mpo</b> rted	<u>Total Value</u>	Value per Ton
1973	14,773	\$ 3,604,300	\$244
1974	44,074	\$23,957,803	\$544

Source: Ministry of Finance, Tokyo

Appendix B to this report describes the imports of 15 countries that are the leading trading partners of Vietnam. In 1972, these countries imported 1,206,999 tons of rice valued at \$191.2 million or \$158 per ton.

In 1974, the price of rice moving in international trade rose sharply, more than doubling from 1973 levels ranging from \$220 to \$250 in 1973 up to more than \$500 in 1974.

Thailand's exports of rice and the per ton value are shown in Table 3.2-A on the next page. Thailand's exports rose from 834,512 tons in 1973 to 1,016,818 tons in 1974 an increase in tonnage of 22% in one year's times. In value terms, the exports rose from \$184.6 million to \$487.7 million, an increase of 164 percent. On a value per ton basis, the increase was from \$221.21 per ton in 1973 to \$479.63 in 1974, an increase of 117 percent.

(Narrative continues after Table 3.2-A)

### Table 3.2-A

EXPORTS OF THAILAND RICE 1973, 1974

### 1973

Exports in Metric Tons

834,512 MT

1,016,818MT

1974

Value of Exports in Babt

in Baht 3,692,000,000 9,754,000,000 Baht 9,754,000,000 Baht Baht Baht Converted to \$U.S. (20 Baht=\$1.00) \$ 184,600,000 \$ 487,696,417 Value per Metric Ton \$221.21/Ton \$479.63/Ton

Note: Exports in 1974 went to about 60 countries. The largest consumers were: Hong Kong, Singapore, Malaysia, Indonesia, Taiwan, Arab countries.

> Source: Royal Thai Embassy, Jan. 31, 1974

- 48 -

------

Some rice eating countries are striving to restore their erstwhile self sufficiency such as Indonesia, Sri Lanka, India and Bangladesh and success in such countries could reduce world demand. However, the Middle East countries have limited rice-growing capability. In general the spectre of food shortages seems likely to continue for many years.

In the light of probable shortages, Vietnam could play a very important role as a supplier. For example, if rice of export grade could be processed in Vietnam, it is possible that Delta exports could reach Thailand's levels.

Also exports would help to alleviate the chronic foreign exchange shortages. For example if Vietnam's peak exports in the 1960's could be duplicated, the foreign exchange revenue would be on the order of \$166.1 million (346,000 tons x \$480/ton = \$166.1). Table 3.1-B shows the historical trend of rice Vietnam's rice imports and exports. From 1957 to 1963 exports ranged from an annual low of 86,000 tons to a high of 346,000 tons. Beginning in 1965, Vietnam switched from a rice surplus to a rice deficit position so that imports are now about 300,000 tons annually.

Rice production has been steadily rising in the Delta as mentioned, thanks in large measure to the new varieties. This growth is shown in Figure 3.2-A. In short, the Delta's unique ricegrowing capability is a valuable asset not only for Vietnam but for rice deficit countries worldwide. This report recommends that every effort be made to improve processing so that some of the Delta's surplus rice can be upgraded to export standards.



THIS PROJECTION OF RICE YIELDS IS ASSUMING 1) 500.000 HA OF DELIA RICE LAND SUITABLE FOR THE FRODUCTION.2) AVE THE YIELDS 5.0 MT/HA.)DOUBLE CROPPING TH 8 ON 250.000 MA.L) FREEDOT TRADITIONAL HA VIELDS RLMAIN STABLE.5) THE 13 UPDD TO DEMOTE ALL DEPONTE VALUE AS TH-5 TH-20 TH-22.

(Sources USAID Can tho)

### Teble 3.2-B

Vietnam Rice Exports and Imports - 1954 - 174

(Unit: 000's of Tons of Milled Rice)

<u>Y•ar</u>	Exports	Imports
1954	173	-
1955	80	-
1956	3	-
1957	202	-
1958	122	-
1959	250	-
1960	346	14
1961	156	2
1962	<b>8</b> 6	42
1963	333	-
1964	49	-
1965	•	130
1966	-	434
1967	-	765
196 <b>8</b>	-	478
1969	•	326
1970	-	559
1971	-	437
1972	-	276
1973	-	304
1974	-	<b>301 (est.</b> )

Source: Directorate of Customs, via FAO Rice Team, Sept. 1974





-----

Rears : -1972 Agricultural Resistion yearbook

		*	*
ť	RKEBNCKRARSSRAR	R	
		R	ŝ
ł	yesterstatest	R	I
ł		*	1
۱	*****	ş	R
5	*********	X	
ľ	Researcher and the second	3	R
ť	ALSERTANDER STR	X	R
<b>N</b>	<b>L</b> MRDERGERARESM®R	8	8
	anena sa katata		<b>1</b>

7 

Ì

Wax - Your

In the Delta where paddy cultivation dominates and where the value of the paddy crop substantially exceeds all other forms of agriculture, priority should be given to paddy as the most important base for agro-industry. (See Chapter 2.3 above).

The processing of paddy involves, (a) drying, (b) storing, (c) husking and(d)polishing. Processing also involves the effective utilization of the by-products, especially bran.

UNIDO has long recognized the importance of by-product utilization and sponsored a conference called the "Interregional Seminar on the Industrial Processing of Rice" in Madrar in October, 1971. As the basis for discussion several important papers were submitted by experts describing world experience in the utilization of by-products (sources 31, 32, 33, 37, 41, 42, 43, 46)(\*). Other important publications have come out in recent years describing the research and recommendations of experts in FAO, UNIDO, and USAID (sources 28, 34, 36, 39, 40, 44, 45).(\*) We would like to recapitulate as follows some of the keypoints in the literature in so far as they relate to the Delta region.

\* See Appendix F Bibliography

(narrative continues on next page)

There are various papers and reports on drying, storage, separation, and milling of paddy in the Delta, one of these, called the "Wildman Report" written in 1970 contains specific recommendations on the large scale, silo-type storage and distribution of paddy in Vietnam (Bibliography 45) including a major storage depot at Cantho.

Another report in 1974 on paddy processing was sponsored by the FAO. Separate sections were written on marketing, storage, drying and mixling (sources 27, 28, 34, 40 \*) These reports, prepared by experts in their respective fields, point out the importance of better utilization of paddy by-products, improving the drying and storage of paddy, and upgrading milled rice for export.

\* The sources cited above are listed in the Bibliography, Appendix F starting on page 241.

(Narrative continues, next page.)

### 3.2.1 Bran Utilization

Rice bran is the beige-colored, oil-bearing layer called the "pericarp" located under the husk. Beneath the bran, is a further oil-bearing layer called "polish" or "aleurone". Both these are recovered by rice mills in the Delta and sold for feed, but owing to the unsophisticated technologies, much of the bran is classified as "course" bran and includes a percentage of husk. Husk is indigestible and therefore only a form of roughage. (Bibliography No. 37.)

The bran, if properly recovered, contains 12% oil. This oil is useable as edible oil and industrial oil, the latter being an ingredient used in soap manufacture. Technologies for the production of edible rice bran oil have steadily improved since World War II. In the early 1950's, rice bran oil had a distinctive odor and **flower**, but since then, the technologies have improved so that even the most fastidious housewife cannot distinguish between rice bran oil, corn, safflower, soybean or palm oils or blends of these products.

By weight, rice bran constitutes about 8% of the paddy and the polish about 2% as shown in the following table (from D. Halliday's paper prepared for UNIDO in February 1971(Dibliography No. 37)

### Table 3.2.1

### Listimated Rice Production 1969-70 in Different World Regions

	Production (Thousands of Tons)					
Region	Paddy	Milled Rice	Husk	Bran and Polish		
S.E. Asia/Far East	260,193	182,135	52,039	26,019		
North and South America	12,914	9,040	2,583	1,291		
Middle and Near East	3,730	2,611	746	373		
Africa, South of Sahara	2,793	1,955	559	2 <b>79</b>		
Vestern and Southern Europe	1,610	1,127	322	161		
Soviet Union	1,500	1,050	300	150		
Niscella <b>neous</b>	3,760	2,632	752	376		
Total	2 <b>86,500</b> (100%)	200,550 (70%)	57,301 (20%)	28,649 (_10%)		

Rice bran is highly appreciated in the Delta by persons engaged in the raising of hogs, chickens and fresh water fish. Evidence of this is the selling price of bran at rice mills in Cantho, namely 100 piasters (about US\$0.15) per kilogram as compared with course rice selling for 111 piasters (about US\$0.16). In short, the bran for animal feed is nearly as valuable as course rice for human consumption.

Rice bran is recognized world wide as a useful ingredient in animal feed. Although bran is being used in the Delta, the material is underutilized because of poor milling.

In some developed countries, rice bran is "defatted", that is the valuable oil products are extracted and the residual bran is sold for feed. In the Delta, however, the bran is not defatted and therefore the oil is wasted. The bran is also unstable and becomes rancid in about 10 days. The rancid bran is not as nutritious, contains less protein by weight, and is even toxic for poultry. Therefore the Delta is losing valuable oils and the nutrient value of the bran is less than it could be.

In Japan about 100,000 tons of rice bran oil are produced annually. The Delta's annual output of paddy, about 5.1 million tons, could in theory yield 331,500 tons of bran which contains about 39,800 tons of oil. Of this, about two thirds or 26,300 tons would be edible oil and one third or 13,500 tons industrial.

The edible oil has a retail value in Vietnam of 1,000 plasters per liter and therefore would have a retail value of about US\$38 million (26,300 tons x 1,000 = 26 million liters x VN\$1,000 = VN\$26 billiondivided by VN\$685 = US\$38 million).

The industrial oil is worth about 46% of the edible oil; therefore the 13,500 tons of industrial oil would be worth about US\$11 million (13,500 tons x 1,000 = 13 million liters x VN\$460 per liter = VN\$6.0 billion divided by VN\$685 = US\$ 8.8 million). (\*)

Therefore oils worth about US\$47 million are being lost each year in the Delta. At the same time, inferior feed in the form of fatty rice bran is being consumed by hogs, chickens and cultured fresh water fish. If the bran were defatted (i.e. oil extracted), the protein content of the bran would rise from 15% to 17.6% on a weight basis. In other words, animals would receive more protein per kilogram of feed.

The improved utilization of bran has two important implications for the future use of bran in agro-industries in Vietnam:

a. The farmers now are feeding fresh bran (i.e. bran containing oil) directly to hogs, chicken, and cultured fish. The Delta has a major potential for raising hogs, chicken and cultured fish on a large scale. If these animal husbandry industries are to develop, it will

(\*) See page 133 for full tabulation of paddy by-products.

be necessary to establish a feed mixing industry in order to make full use of feed raw materials available in the Delta. Bran by itself is not a perfect feed-stuff, but when other elements are added to make a well balanced feed, bran is a key element in feed. (See Table 4.4)

b. In Vietnam, substantial amounts of foreign exchange have been spent both through Public Law 480 and the Government's own funds to import edible oils. These expenditures of scarce foreign exchange could be reduced or perhaps eliminated if rice bran oil were produced locally. It will be most important therefore to develop industries for extracting oil from bran. This will be described in greater detail in Chapter 4 later in this report.

## 3.2.2 Processed Foods Derived from Rice

Rice is of course the staple food in Vietnam and is consumed in various ways such as rice wine, crackers, noodles, and many other foods derived from rice. However, the traditional Vietnamese rice cooking processes are deeply rooted and of very long standing. The modernization of cooking and processing methods for rice is important to make full use of the broken rice as a raw material. Since the traditional methods of cooking are closely related to individual taste preferences, the expansion of the market in Vietnam for processed foods derived from rice can only take place gradually.

However, there may be quicker ways of utilizing brokens for making rice flour for export. In other countries, rice cake fried in deep fat is a popular canape.

Various processed products are described by J. Sakurai in his report of July 31, 1974 entitled, "Development Prospects of a Modern Rice Products Processing Industry in the Lower Mekong Basin." (\*)

### 3.2.3 Parboiled Rice

The parboiling of rice is closely associated with the rice milling processes in certain Southeast Asian countries. Such as Ceylon, India and Bangladesh. Thus far parboiling has not been practiced in Vietnam. Apart from the peculiar flavor of parboiled rice, parboiling serves the purpose of hardening the grain and thus reducing brokens in the milling process. A second advantage of parboiling is the longer shelf life of parboiled rice. A third advantage is the better retention in the rice of nutrients valuable for human consumption.

In the Delta, where rising rice production may soon result in a major surplus, and where storage could become a problem, parboiling could be one of the solutions for storing rice for export. Parboiling is relative inexpensive because the husk, usually a waste byproduct, can be used as fuel.

(\*) Bibliography No. 63

### 3.2.4 Utilization of Rice Straw

There is a long history of straw utilization in Asia. Mats, rope and bags have played an important role as packaging materials for agricultura crops. In recent years, straw products have been replaced by more efficient and durable jute, multi-ply paper, and bags made of woven polypropylene ribbon.

About 100 years ago, sandals made from rice straw were widely used as Japanese shoes. Japanese floor mats (tatami) are still used in place of carpeting in Japanese homes. The tatami mat contains a filler made of rice straw about 3 centimeters thick. With the introduction of western style housing, the demand for tatami has declined. Straw thatching for roofs in rapidly declining. It is unlikely that such uses of straw will offer Vietnam a promising market for rice straw.

Straw can be used, however, as a raw material for pulp in the manufacture of paper. Rice straw fiber is short and is not an ideal material for pulp. In the past, low quality paper was produced from rice straw pulp in Japan and China, but there is little demand for such low quality paper at the present time. As a raw material, rice straw itself is inexpensive in Japan, but to use it for pulp, caustic soda and chlorine to remove impurities are required in larger quantities than in the case of wood pulps. When these chemicals are expensive, straw pulp made from rice straw becomes uneconomical (\*).

Nevertheless, the Japanese paper industry is trying to utilize straw pulp because of the growing scarcity of wood pulp, world wide. In such cases, the straw pulp is blended with the long-fibered wood pulp to attain an acceptable fiber length.

It is likely that the rice growing countries of Asia will endeavor to utilize straw pulp in a similar way. (W. Pierce of USAID/Saigon recommends the use of straw for pulp production on the Tay Do Industrial Park in Cantho. His paper is entitled, "A Rice Straw Pulp Plant in the Cantho Area", Source No. 46-A.)

Several factories have been established in Southeast Asian countries to make pulp and paper from rice straw. These projects have not been entirely successful in terms of their effective utilization of rice straw. Pulp manufacture from straw has been tried in Indonesia, the Philippines, and in Vietnam (Bien Hea Industrial Estate), but all plants have been forced to supplement the rice straw pulp with expensive, long-fibered, imported pulp.

COGIDO, a paper mill in the Bien Hoa Industrial Estate near Saigon, is producing paper from local straw, bamboo, and imported long fiber pulps. Straw pulp made by COGIDO is blended with other pulps and accounts for 20% to 24% of the pulp used for the manufacture of paper. The COGIDO plant operates at only 20% to 40% of capacity throughout the year largely because of the deficit in pulp raw materials (Bibliography, No. 52-A). The rice straw used by COGIDO is baled

(\*) Bibliography No. 52.

in the Delta and shipped to Saigon. Despite COGIDO's vigorous efforts, including the distribution of an ingenious mechanical baler, insufficient quantities of straw are arriving at the plant. Because of the straw shortage, bamboo and imported pulps must be used to keep the plant in operation. This situation suggests several important conclusions in planning new straw pulp factories.

Siting a pulp factory in Cantho would facilitate the gathering and transport of straw from nearby growing areas, but the following problems will probably occur.

a. To make paper from rice straw pulp, a larger volume of long fiber, better quality pulp are required for blending. To be assured of a steady future supply of imported pulp will be a serious problem as it now is at COGIDO.

The use of certain local trees such as mangrove and tram could become a pulping raw material but the harvesting of these trees poses problems, especially the harvesting of mangrove. (See Section 3,14 later in this report.

b. Economical production of pulp requires a major capital investment and high production volumes. Even if a sufficient volume of straw could be gathered, it will be difficult to consume all the products domestically in Vietnam. If the surplus is to be exported, it is doubtful whether the price and quality would be competitive on world markets in view of the high costs of chemicals (caustic soda and chlorine) needed to process the rice straw pulp.

c. The principle problem is the collection of raw material in sufficient volume and quality. So far, the COGIDO case suggests that the collection of the raw material has not succeeded. The very nature of straw means that small volumes must be collected over a wide area. The larger the volumes collected, the higher the cost of collection. In Japan, farmers dry their straw and store it, but in Vietnam, the straw becomes wet during the rainy season. Straw must be kept dry for transportation purposes as well. Storage capacity will be a further problem especially in view of the bulky nature of rice straw. Other pulp material such as wood is less bulky and is usually stored outdoors.

d. Farmers prefer to leave straw on their paddy fields and burn it to clean the field of infestation. They also use small amounts for animal fodder.

In summary, we doubt that further analysis of the feasibility of utilizing straw for pulp would be warranted.

It is possible to produce particle board from straw, as is being done with jute stalks in Bangladesh (The product is called "Partex" and is based on German technology). However the cost of the binder and the cests entailed in assuring a steady supply of raw material rice straw would probably result in a product cost that would make the particle board more expensive than natural wood. In Mangladesh, there is a shortage of natural wood, but Vietnam is an exporter of wood.

Agronomists and specialists in horticulture agree regarding the effectiveness of straw as a soil conditioner. In future years, as cultivation intensity increases and soil nutrients are depleted, the inherent productivity of the soil will decline. The use of chemical fertilizers to supply plant nutrients is expensive and will not fully restore the balance of organic and chemical elements in the Delta's densely packed, clayey soils.

1

A good balance of organic materials in the soil is important especially for the cultivation of vegetables and fruits. In paddy fields, supplementary elements such as silicon dioxide  $(SiO_{c_1})$  are important in addition to the essential plant foods, N, P and K. Experts suggest that the best utilization of straw is to return it to the soil by procedures such as the following:

a. In harvesting paddy, only the upper extremity of the stalk should be removed leaving the straw in the ground to be plowed under when preparing the soil for the next planting.

b. Burn the straw, leaving the ash on the surface of the soil.

c. Mix straw with dung as a fertilizer after decomposition.

These three methods have good and bad points, but detail is cutside of the scope of this report.

### 3.2.5 Utilization of Husk

In many rice milling areas of the world, the disposal of husk has posed serious problems. The burning of husk creates air pollution, the storage takes up space, and dumping into rivers pollutes the water.

One of the worst problems has occurred in Malaysia which Eldon Beagle described as follows in a paper prepared for UNIDO:

"The rice mills in the Kuala Lumpur area can be considered an example of the ultimate stage in the husk enigma. The ecological implications of their disposal methods have boomeranged on them. Originally located in Kuala Lumpur proper, these mills moved to the suburbs some years ago to escape from the rapid expansion of the city. However, during the past ten years, this suburb has grown into a highly industrialized area and the mills are surrounded by large businesses that exert significant political influence. As a result, the rice mills have been subjected to numbrous fines for polluting the air and surrounding area with the ash from their practice of burning husks in incinerators or open piles." (Bibliography Source No. 31).

Despite the problems posed by husk disposal and the challenge to obtain commercial benefit from the husk, there are few entirely effective means of utilization for developing countries. In developed countries, where there is a sizeable infrastructure of supporting industry, commercial and large-scale utilization of husks is in fact taking place.

Some of the commercially proven uses of husk are: soft grit blasting (i.e.an abrasive), oil obsorbtion (greasweep), material for roofing paper, a carrier for vitamins or medicines, various feed uses (extender, poultry feed, ruminant feed, retarder feeds), an anti-caking agent for fertilizer, fuel for steam generation, insulating material, fruit juice processing, a source of ferrosilicon, a source of furfural silica (for refractory brick), soil conditioning mulch, litter and nesting for poultry.

The authors observed the following utilization in the Delta:

1. Farmers use the ash from burned husks as a soil conditioner when cultivating paddy seedlings. This is traditional practice that Delta farmers have learned from experience of many years.

2. Some of the older rice mills are burning husk, primed with kerosene, to generate steam for steam engines. This is a very low cost fuel. (The Mekong Rice Mill near Saigon has capacity for milling 100 tons of paddy per day; there are two engines, one is a 77 HP diesel and one a 100 HP steam engine. The owner hopes to replace the diesel engine with another steam engine, but none were available for sale.)

3. Sale as a milling by-product for fuel (husk sold for 300 plasters per 50 kg, bag in 1974.)

4. Litter and nesting material for poultry.

In other Asian countries where rice is traditionally parboiled (Bangladesh, India, Sri Lanka), husk is used as a fuel for the boiling process as well as for a fuel for steam engines.

In times of serious fuel shortages in Japan, farmers have used a portable home range for cooking. A prototype of this kind of stove is reportedly available through the Ministry of Agriculture in Tokyo. (See also Figure 3.2.6. p. 69)

We have Calculated that rice hulls, as a fuel, cost about one fourth as much as fuel oil as follows (the calculation assumes that husk will yield 6,000 BTU/lb. or 13,200 BTU/kg.):

		Rice Hulls	Fuel Oil
A.	BTU yield	13.2 million BTU/MT	37 million BTU/Kl
в.	Cost in Vietnam	<b>\$ 8,7</b> 6 (6,000 pias- ters per MT)	\$100.73 (69,000 piasters per kiloliter)
c.	Cost per 1000 BTU	\$ .0007 or 0.45 piasters	<b>\$ .003 or 1.8</b> 6 piasters

The conclusion from this data is that 1,000 BTU from husk is one fourth the cost of 1,000 BTU from fuel oil (0.45 plasters/1.86 plasters=24%).

We have conducted a survey in Japan to determine if small steam engines in the 20 to 100 HP range are being marketed. We found that no manufacturer in Japan is making steam engines and that if such engines were made, the cost per horsepower would be prohibitively higher than diesel, electric or other types of engines.

It is not practical to think in terms of a widespread use of husk as a fuel for steam engines until such time as high production volumes bring down the cost. But the use of husk as a fuel for other heating purposes is feasible. However this is a project that CARIC, the machinery manufacturer in Saigon, should investigate. Also the manufacture of a portable husk stove for home cooking would merit study. (See Figure 3.2.6.) After burning, the ash weighs 15 to 22% of the original husk. The ash contains the following elements:

Silica	(SiO <sub>2</sub> )	4.50 per cent
Calcium Oxide	(Ca0)	0,25 per cent
Manganesium Oxide	(MgO)	0,23 per cent
Potassium Oxide	(к <sub>2</sub> о)	1.10 per cent
Sodium Oxide	(Na <b>0</b> )	0.78 per cent
Phosphoric Oxide	(P <sub>2</sub> 0)	0.53 per cent
Sulphates		1.13 per cent
Natrium	(Na)	(not available)

Aluminum, Managanese and Ferric Oxide traces

The noticeable fact about this analysis is the high silica content of the ash. It is this component which accounts for the number of processes used and proposed for the utilization of rice hulls based on their ash content (Bibliography, No. 32).

Of the commercially proven uses of husk reported in the literature, applications in Table 3.2.5 (next page) would appear to have some future promise in the Delta. In the long run, an important utilization could be for the manufacture of furfural. In the United States, furfural is made from oat husk, corn cobs, and paddy husk. Consumption of furfural in the United States was about 60,000 tons annually in 1969 and 1970 and was valued at \$18.8 million and \$20.2 million respectively. The main uses for furfural are for plasticizers and synthetic rubber (see Figure 3.2.5). The theoretical maximum yield of furfural from the Delta's 5.0 million tons of paddy produced annually would be 40,000 tons (5,000,000 tons of paddy x .16%=800,000 tons of husk x 5% = 40,000 tons of furfural).

Some prerequisites for the manufacture of furfural are: (a) assured volumes of husk on the order of 100,000 tons per year and (b) assured export markets until such time as Vietnam has a domestic requirement for furfural. The collection of husk from scattered mills would jeopardize raw material supply, but the establishment of a large mill or mills capable of generating 100,000 tons of husk is conceiveable but would require capacity larger than presently exists. (A mill to supply 100,000 tons of husk would have an annual milling capacity of 600,000 tons or 2,000 tons per day; one of Vietnam's largest mills, e.g. S.E.D.I.C. in Cholon has a daily capacity of only 648 tons per 24-hour day or 194,400 tons per 300-day working year.)

The manufacture of furfural would require large capital investment, probably on the order of \$10 million (estimated costs were \$5.5 million in 1971, according to source No. 32). There are large scale furfural plants in the United States owned by the Quaker Oats Company and there are smaller scale plants in Europe. (See list of plants following Table 3.2.5). (See Appendix F for sources.)

### Table 3.2.5

POSSIBLE INDUSTRIAL APPLICATIONS FOR PADDY HUSK (Bibliography: 29, 30 and 31)

(This is only a partial list; please see sources.)

1

Product to be Manufactured from Husk (Biblio- graphy Source)	Estimated Mini- mum Husk, Raw Material Required (MT)	Estimated Initial Capital	Suitability for Delta
Feed for ruminant animals (31)	3 Tons/day	n.a.	Product not competitive with rice straw for buffalo in Delta. Also requires use of some bran which has more effective applications.
Silica for fire- brick or grease absorption compounds (31)	40, <b>000</b> Tons/year	<pre>\$ 1.0 million (est.)</pre>	Possible if export markets assured.
Furfural and Silica 5,000 Ton/year of fur- fural used for chemical applications (29, 31, 32)	100 <b>,000</b> Tons/year	\$10.0 million	Possible if export markets assured and if 2,000 Ton/day rice mill were built.
Pressing aid for fruit juice manufacture (31)	1,000 Tons/year		Parboiled husks are pre- ferred and parboiling is not practiced in Vietnam. Transport costs to developed country markets would probably make this product non-competitive
Activated carbon (32)	n. a.	n.a.	High volume of zinc chloride required and transport costs to Delta plus export market pose problems.
Panel boards using coment by tropical Products Institute or Italian "Secondite" process (32)	j n. e.   	n.e.	Probably not competitive with wood

Product to be Manufactured from Husk (Biblio- graphy)	detimpted Hini- mam Heak, Nav Haterial Required (NE)	Botimical Initial Capital	Boltability for Bolta
Panel boards using French "Silicior" pro- cess (32)	n	N.A.	Probably not compositive with wood
Bricks made of rice hunk ash and soil or comput		if done in arteeanal donner, capital would be donimal	Could be feasible cancidering dense clayey Dolta quality of soils. Sum-drying would avoid hilm costs.

(table 3.2.5 ands here)



### INDUSTRIAL APPLICATIONS FOR PURPURAL



Seurce: Stanford Research Institute, <u>Chemical Griaine and Markets</u>, 1987, Nonie Park, California 1

<u>Outre r</u>	Location	Annual Capacity in Metric Ton=	
The Quaker Oats Co., Chemicals Division	Belle Glade, Florida	22,675	
11	Cedar Rapide, Iowa	9,070	
**	Memphis, Tenn.	18,140	
19	Omaha, Nobra <i>s</i> ka	18,140 (same)	
International Petro-Chem, Inc. (Planning Stage)	Bayou Teche, (Lafayette) Louisiana	10,140 (same)	

### PURFURAL PLANTS IN THE UNITED STATES -- 1971

Source: Stanford Research Institute, 1971 Directory of Chemical Producers

The furfural possibility and others noted in the literature (Bibliography **DUPCON** 29, 31, 32) are shown in Table 3.2.5. With the exception of the final item listed, namely bricks for building construction, there appears to be some impediment to effective utilization of husk, for example high raw material costs in the case of the production of activated carbon or raw material limitations in the case of furfural. Husk is used as roughage in feed for ruminant (i.e. cud-chewing) animals, but in the Delta, the readily available low cost rice straw for water buffalo would probably render the husk non-competitive. (There is no significant dairy cattle industry in the Delta.)

We conclude that until rice milling capacity is raised substantially and concentrated in one locality, that the best uses for husk are: (a) as a fuel if suitable stoves and engines can be manufactured, (b) as a mulch to loosen the dense, clayey soils, and (c) as a soil conditioner in ash form in keeping with the traditional practice in the case of paddy seedlings, (d) litter and mesting material for chickens.
### Fimre 1.2.6

#### COOKING STOVE FOR BURNING HUSK

The following article from the March 17, 1975 <u>Aschi Nevemener</u>, Tokyo, describes a stove used to burn husk. Such a stove could be made in the Delta for household cosking. The retail price is the yen equivalent of \$93 (about 70,000 plasters). Farmers having 1.5 hectares of paddy can produce enough husk to cook for 5 persons for one year. The required husk is stored in a small warehouse having 3 square meters of floor space. The manufacturer planned to sell 50 units in 1975, but has already sold 250. (The manufacturer is Hitachi Koki Haramachi Factory in Pukushima Prefecture.)

1

1

,

工業がとって	きのチェ 16月17日	A 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
			を見て、二世氏の副を通っている。「世長の副を通っている。」「世界の男人」という男もの男子、 男をくた」という男もの男人 たっている。「世界の男人」で
(See translation	acat page )		

#### FACTORY START'S TO PRODUCE HUSK-FIRED STOVES UNLIMITED SUPPLY OF FREE FUEL

#### SALES EXCEEDED EXPECTATIONS

#### (This is a translation of Japanese News Article on Page 69)

The Haramachi Factory of Hitachi Koki Company located in the outskirts of Haramachi City, is one of the main foundries of the Hitachi group of companies. A joint conference attended by representatives of management and labor took place in the conference room of the factory in November, 1974. The subject for discussion was, "How can the company cope with the reduced level of sales and profits chused by the business recession?". The recession had caused a reduced scale of operations ranging from 60% to 80% of full vapacity.

"Max you any suggestions to make? We are very hard-pressed" the factory manager asked the group.

A vice-chairman of the labor union rose to his feet. "Suppose we make stoves that burn paddy husks. Farmers seem to be harrassed by mountains of husks." Another member of the nuion remarked," Electric po er and petroleum are too high priced to be used these days."

The factory operations manager, the manager of the general affairs department and the foundry manager went out to visit many rice-growing areas in Pukushima prefecture the following day. Information started to come in soon afterwards. "There is a stove manufacturer, located in Kawaguchi City, Saitama prefecture: because of the inadequate supply of parts, however, only a few stoves are actually in use." "Most farmers will produce enough husks to burn for a full year." "All farmers had huskburning stoves in pre-war days. "

In December, stoves were manufactured for trial purposes based on the prototype made by the Kawaguchi manufacturer. Since the validity of the patent registration and utility model had expired, the model was copied without going the trouble of makeing prototypes. The factory decided to sell stoves for V29,500 (U.S.193), which was cheaper by more than V10,000 (1334) than the stoves manufactured in Kawaguchi.

A trial of the stoves was held at the factory: 3 <u>sho</u> (or 5.4 litres) of nice was concluded in 20 minutes and 5 sho (or 9 litres) in 30 minutes. The taste of the concludentice was superb, the grains were well rounded and the heat remained after the cooking was completed.

From the beginning, the factory did not expect a large volume of orders. While castings for electric equipment that ditach makes are 6 milimeters thick or more, the castings for husk-burning stoves are only 4 milimeters thick. We thought we would have to obtain technology for producing this castings," confided Nr. Hiroshi Canigawahara, the casting manager. They anticipated a maximum order of 50 stoves. It turned out that orders for as many as 210 units have been received as far. Orders for wood-burning stoves came in also, which the factory has also begun to manufacture. At a price of ¥25,000 (\$US 85) 200 stoves have been sold. At present, a special production team for manufacturing stoves has been formed with 30 employees out of the total of 350.

Some of the repercussions are as follows: (1) A farmer close to the factory has paddy field of 1.5 hectares. From the paddy grown there, enough husk will be produced to fill to the ceiling a warehouse with 3 square meters of floor space. Rice lasting for three meals or a full day for a 5-member family is cooked once a day in the morning hours and kept in a jar. Husk stored in the storeroom will last a year. The head of the family declares, "The rice tastes perfect." (2) "Stoves are sold in the adjoining Miyagi prefecture. The popular reception of stoves is a surprise to us. As the stoves require an uninterrupted supply of parts, we intend to continue even after the present depression is over," says Mr. Hide Seya, the factory manager.

In the city of Haramachi in Fukushima prefecture, a manufacturer of electric equipment suffering from a reduced volume of business, noticed that paddy-growing farmers were finding it difficult to dispose of husks. He now manufactures husk-burning stoves. "The farmers who use the stoves have no fuel costs and they can cock efficiently." Orders are coming in from farmers. The factory says that they will start full production since there is an inexhaustible supply of husks. In view of energy shortages, the revival of husk-burning stoves is likely to become a success.

> (Tranulation of article ends here; text of report continues on next page.)

# 3.3 <u>Venetables and Fruits</u>

# 3.3.1 Yemetables

The production of vegetables in Vietnam has increased steadily in recent years as shown in Table 3.3.1 below. Out of the total production of 357,600 tons in 1973, the Delta accounted for 28% (Chapter 2, Table 2.3.)

# Table 3.3.1

	Area (ha)	Production (m. tons)	Yield ( <u>MC/ha</u> )
1965	11,720	132,955	11.3
1966	12,100	143,470	11.9
1967	13,540	192,180	14.2
1968	13,620	193,055	14.2
1969	16,885	234,955	13.9
1970	17,850	217,550	12.2
1971	18,065	230,155	12.4
1972	22,900	294,000	12.8
1973	30,160	357,600	11.9

# PRODUCTION OF VEGETABLES IN VIETNAM -- 1965-73

Source: Ministry of Agriculture Source No. 30 The principle vegetables produced are: tomatoes, cucumbers, egg plant, water melon, chives, garlic, red pepper and cabbage, scallion, ginger root, white radish, Chinese cabbage, carrots.

Figure 3.1 in Chapter 3 shows that vegetables can be processed to produce tomato juice, puree, ketuchup, etc. However the domestic market for these products is as yet small and they are not part of the traditional Vietnamese diet. In cities like Saigon, there is a limited demand in hotels; also urban residents will probably begin to consume these products in a few years. This growth in demand should permit the establishment of a small-scale processing industry later on, but it would probably be premature to establish a large-scale, modern, efficient plant in the Delta at this time.

An industry for the manufacture of tomato juice for export faces possible problems of acceptance by food and health authorities overseas. These must be overcome in order for the juice to compete effectively in international markets.

A substantial volume of pickled (i.e. brined) vegetables are being consumed in Japan and some portion comes from Taiwan. Because of freight costs, it is likely that the future supplies of pickled vegetables will originate in areas closer to Japan than Vietnam.

We believe that if a steady sales outlet were available, farmers in the Delta would be prepared to grow vegetables because the return per hectare is high. One hectare can yield more than 11 tons of vegetables, according to Table 3.1.1. We do not have the retail price of tomatoes in Cantho, but the price of cabbage in October, 1975 in Cantho was 200 piasters (US\$0.29) per kg.

A broker buying vegetables for a manufacturing plant would pay 60 to 70% of retail. Therefore the farmer would probably receive the equivalent of US\$174 per ton ( $$0.29 \times 1000 \text{ kgs} = $290 \times 60\% =$ \$174). The return to the farmer on a hectare of vegetables could be about \$1,740 (10 tons of vegetables per hectare x \$174). The return on a hectare of paddy would be only \$848. (581,120 piasters)

In planning procurement of tomatoes, long term procurement contracts would have to be made. Tomatoes require greater labor inputs to preserve the crop during the growing season and more preparation to aerate the soil than is the case of paddy.

# 3.3.2 Pruits

The principal fruits grown in the Delta are tangerines and oranges. Of the total 282,000 tans produced in Vietnam, 192,880 tons or 68% were produced in the 16 Delta provinces in 1973 (Table 2.3 in Chapter 2.).

There are a few medium-size orchards; one is at Tho Long Village, Phong Phu District (20 kilometers WNV of Cantho on Noad LTL 27), According to USAID in Cantho, the owner has 10 hectares planted in oranges. In Thot Not, Anglang Province (40 kilometers NV of Cantho on LTL 27), there are more orchards, some reportedly having 10 to 20 hectares planted in oranges. Orange and tangerine production in Vietnam is shown in Table 3.3.2 below. As mentioned, about two thirds are grown in the Delta.

# Table 3.1.2

PRODUCTION OF ORANGES AND TANGERINES IN VIETNAM -- 1982-73 (Note: these figures include about 5% papaya)

Year	Area Planted (ha)	Production (m. tons)	Yiold <u>(MT/ho</u> )
1962	35, 090	219,625	6,259
1963	35 <b>, 58</b> 0	26), 540	7.345
1964	34,570	231,730	6.703
1964	37,225	233, 260	6,266
1966	351990	201, 900	5.610
1967	33,730	191,165	5.568
1968	32, 340	221 <b>, 88</b> 0	6,681
1969	32,030	222,865	6.999
1970	32,920	235, 705	7.160
1971	33,600	254,000	7 <b>.945</b>
1972	38,300	230, 500	6.018
1973	421850	282+000	6.581

Source: Ninistry of Agriculture (Source 20)

The production of oranges and tangerines is profitable and would become more so for the farmers if a steady industrial outlet for the fruit were available. For example, 1 hectare of orange plantings could yield the farm the equivalent of almost \$4,000 per hectare per year (60 plasters for one,quality orange at retail x 60% = wholesale price x 12 oranges per kg. x 1,000 = one metric ton x yield per hectare of 6 tons = 2,592,000 plasters divided by 685 plasters per one USS=\$3,784 per hectare).

Bananas are also grown in the Delta. Of the total Vietnam production of 258,000 tons in 1973, the Delta's 16 provinces produced 216,100 tons or 84% (Table 2.3 in Chapter 2). However, bananas are grown on a small-holder scale rather than on plantations as in Latin America, West Africa or Jamaica.

Although the quality and taste is satisfactory, the size of the fruit is small and the trees are stunted. This is because the species need upgrading, the roots have difficulty in penetrating the soil, and lack agration. During our field survey in October, we inspected banana trees standing in water which could be a further factor in the stunted growth.

We believe that banana growing in the Delta on a commercial scale for export does not warrant further investigation, although the supply of banana for local markets will remain profitable for the Delta farmers.

Pineapples are grown in the Delta. Of the total Vietnam production of pineapple of 34,350 tons in 1973, the Delta produced 28.155 tons or 87% of the total.

As in the case of banana, pineapple are grown by small holders and it would probably not be feasible to re-arrange farming on a large-scale plantation basis. Even if certain soils in the lower Dolta can be desalinated in the future, the large-scale planting of pineapples would probably not be an optimum use of farmland owing to the densely packed nature of the soil (Appendix C).

In tropical countries such as Vietnam, it is usually difficult to develop a local market for processed or preserved fruits because froch fruits are in abundant supply throughout most of the year. Howover, Vietnam's processed fruits, especially oranges, can eventually find a market in the United States, European countries and Japan. The international market for canned or bottled fruits is keenly competitive, and under these circumstances, it is not easy to penetrate foreign markets. For Vietnam to compete, the following points would have to be considered:

a. It is important to secure a steady and sufficient volume of good quality ray material. The quality has to meet the taste criteria of commons in the market countries and should equal or surpass the quality of competing countries. It is also important to recognize the sonitation and quality regulations of the importing countries. b. It is important to increase productivity in Vietnam. Nore study is required regarding the productivity of other competing countries and it will be necessary to improve Vietnam's productivity to the level of the competing countries.

c. Containers and packaging are also important to success in exporting. In the case of canned food exports, local supplies of tin plate and cans would be preferable. The shipment of cans from abroad to Vietnam would not be feasible, and probably the best container would be blown polyvini bottles, maintioned in Chapter 4 in connection with rice bran oil.

Fruits, specially oranges, in the Delta have an export potential as juice or purce but it will be necessary to study the market demand in prospective importing countries and to promote appropriate export policies and strategies aimed toward specific export markets. We believe that despite the long-run potential for future exports, conditions in the Delta do not now offer sufficient assurance for the collection of raw material from scattered orchards. More important is the fact that export channels are not sufficiently developed to consider fruit processing as a successful agro-industry.

We believe however that the processing of jams, marmalade for the domestic market should start on a small scale and gradually expand. In view of the Delta's leadership in orange and pineapple production, Cantho would be a logical site. However, we are not recommending this for the Tay Do Park since production would probably be on an artesanal scale for some years.

# 3.4 Buger Milling

Imports of sugarare large and have risen in recent years to 362,000 tens in 1973, cesting \$55 million as tabulated below:

Vietnamese Imports of Sugar, 1971-73

	Tene	U.S. Dollar Value
1971	228,493	\$37,580,000
1972	163,971	\$16,9 <b>88,000</b>
1973	362,000	\$55,044,000

Source: Sydney M. Cantor Associates (See Bibliography Source 9 and Source 20)

Buring the peak of hostilities, sugar cane cultivation was banned in some areas resulting in the a decline from the 1965 peak of 1.1 million tens of cane down to a low in 1969 of 321,445 tons as shown in Table 3.4.

#### Table 3.4

liner.	Area (ha)	OF SUGAR CANE IN Production (D. tons)	VIETNAM, 1965-73 Yield <u>(HE/ba)</u>
1965	33,900	1,092,850	32
1966	30,050	935,670	31
1967	25,770	769,960	30
1968	15,265	426,070	28
1969	11,670	321,445	28
1970	11,620	335,720	29
1971	12,600	340,500	27
1972	12,400	)) <b>1,00</b> D	27
1973	17,400	529,900	30

- 77 -

Of the total sugar cane cultivated, 529,900 tons in 1973, the Delta accounted for 126,400 tons or only 24% of the total. (Table 2.3)

The Government is providing incentive programs to encourage the farmers to cultivate sugar cane, chiefly loans on preferential terms from the Agricultural Development Bank (ADB). The purpose of the incentives is to save some portion of the heavy costs in foreign exchange occasioned by sugar imports. The current world-wide high price of sugar has primed domestic prices; this has encouraged the farmers to start sugar cane cultivation. (In July 1972, sugar cost \$136/ton; in March 1974, \$480 per ton f.o.b.) At present, sugar cane cultivation and processing in Vietnam is expanding rapidly. However, sugar imports remain substantial.

In spite of the Governmental incentives, however, and the farmers eagerness to grow cane, the productivity in Vietnam is low by international comparisons.

The average sugar yield in Vietnam has consistently been low since 1960, fluctuating between 27 and 30 tons of came per hectare (Table J.4 ). This compares with the average yield in Taiwan of 76 tons per hectare. The sugar content of the Vietnamese came is typically 8% compared with Taiwan's 12% (A new variety is being introduced into Vietnam which reportedly has a sugar content as high as 14%, but we were not able to confirm this during our survey.)

The sugar cane observed by the writers in the Delta was stunted and the sugar content probably below 8%. This seems to be caused by the undersirable (for sugar) characteristics of the Delta soil and insufficient drainage where the water table is high. In addition, the methods of cultivation are crude i.e. shallow plowing, and some farmers are not as familiar with cane as they are with paddy.

The sugar refining process consists of: (a) crushing the cane, (b) boiling the syrup in open pans,(c) pouring viscous syrup into molds to make a product called "lump sugar". This actually is a sugar cake about 6 cm. long and 2 cm. thick. The resulting sugar has a dull-sweet flavor and would be classified as raw sugar. The "lump sugar" is probably adequate for local consumption, but could not be exported.

There is some modern refining capacity in Vietnam, four mills with a combined capacity of 650,000 tons of cane, capable of producing some 60,000 tons of plantation whites annually (See Source 9 and Pearson, J.L. and Summers, L.V., "Processing of Sugar Cane and Naw Sugar in Vientam -- An Economic Appraisal, "July 1972, USAID.) However the four mills, as of August 1974 had been shut down since 1967 because of a lack of raw material. (\*)

Even if the four large mills were in operation, they could account for no more than 15 to 20% of estimated Vietnamese consumption of 300,000 ton

(\*) Source 9 in Hibliography

400,000tons (Source 9). Therefore the bulk of the sugar consumed is produced by the small-scale, crude methods described above.

In the six provinces immediately adjacent to Cantho, there are 66 sugar mills producing sugar similar to the lump sugar metioned. (Source 10 in Bibliography).

The largest of these has a production capacity of only 3 tons of lump sugar per day. These small-scale, scattered, low-productivity plants are primitive, and do not produce molasses.

The construction of a modernized, highly efficient sugar mill at Tay Do will not be promising for the following three reasons:

(a) Efficient modern sugar mills require economies of scale and large volumes of cane grown in the immediate vicinity. The current small-scale, widely scattered farms could not support a modern sugar industry.

(b) The quality and sugar content of the cane species is below international standards.

(c) Delta soils are less suited to cane than soils elsewhere in Vietnam.

Given the handicaps facing the sugar industry, future expansion should be geared to the growth in local demand.

Two lump sugar factories are being established on the Tay Do Industrial Park in Cantho, and one was scheduled to start operations during the harvest season which began in October-November of 1974.

In a modern sugar mill, the molasses produced as a by-product is a valuable material for cattle feed and for industrial use. No molasses is produced from the present lump sugar factories.

Bagasse (i.e. cane residual after crushing) is being used as a fuel in the lump sugar factories. Bagasse can also be used as a raw material for particle board or paper. However since bagasse is a lowcost fuel for the lump sugar mills, it is doubtful that enough bagasse could be diverted to the manufacture of particle board or paper. See 3.21 below for commonts on starch as a partial sugar substitute.

#### 3.5 Oils and Fats Processing

Raw materials available for oils and fats in the Delta are: rice bran, soybeans, peanuts, coconuts, sesame meed, kapuk meed and corn. Other oil-bearing raw materials produced in minor volumes are: rubber, rubber meed, jute and kenaf meeds, tish oil, tallow (from water buffalo), lard (from hogs). Among the raw materials mentioned above, rice bran will be dealt with in detail in chapter 5 of this report. (See Bibliography No.60% ing, Frank P. "Feasibility Study: Edible Oil Seed Production, Marketing, and Processing in Vietnam, 1971.")

#### 3.5.1 Soybeans

Soybeans are an important naw material not only for oils and fats but also for mixed feed and for food processing industries. This is because the protein content is high. Table 3.5.1 shows production of soybeans in Vietnam for 1965 through 1971. Out of a total production of 10,600 tons in 1973, the 16 Delta provinces accounted for 198 of the total, namely 2,000 tons. Most of production in the Delta comes from Chau Doc, An Giang and Prong Dinh. However, the Delta is not ideally suited for soybean cuttivation, and other areas of vietnam have climatic, soil or humidity advantages.

The soil in the Delta is generally acht (See Appendix (), the pl value can be as high as 3.5. Soybeans tend to thrive in mentral soil (i.e. pH 7 soil) and thus it is necessary to neutralize the achtiv by applying calcium, usually in the torm of limestone. Segmetables and fruits can be grown in the same acht suits and thus compete with soybean cultivation. Under the present situation, vegetable cultivation is usually more profitable to the tarmer than the growing of soybeans.

The variety of soylean new being grown in the Bella is intended for direct human consumption rather than for all extraction. The cost of cultivation is comparatively high and it oil-bearing varieties are introduced, the oil will probably not be competitive on the international market. It is estimated that 50 percent of soylean consumption in 1970 was used for bean curd, 50% was used directly is edible beans, and 10% for bean sprouts and other uses. flictlingraphy No. 48).

Under these circumstances, it is difficult to count on large volumes of soybeans as now external for all extraction. Vietnamese authorities are planning to increase the production of soybeans in the future for all extraction as well as for the production of soybean cake for use in mixed feeds. If the increased use of soybeans in fact materializes, the following developments will have to take place: (a) improvement of the soybean valieties, (b) increase in the yield per hectare, (c) decrease in the production costs, (d)creation of a stable market (e) and incentives for the farmers wifficiently attractive to induce them to cultivate soybeans rather than other creas. The increase of seyboan production should be encouraged where oblis are suitable. Seyboans are important not only as ray unterial for oil entraction but also for human consumption.

Total Vietnam production of soybeans is shown in Table 3.5.1 below.

# Jable J.S.L

## VIETNAME PRODUCTION OF SOVIEANS, 1965-71

Year	Area (the )	Production (D. topo)	Yield (MR (be)
1965	3, 180	4,110	0,805
1966	6,610	7,983	1,140
1967	7,999	5,660	0.749
1968	7,800	7.445	0.955
1969	6,963	5,965	0.911
1970	6,875	7.495	1,091
1971	7,700	A, 400	1, <b>191</b>
1978	8, 100	6, 60D	0.771
1973	11,170	10,400	0.491

Source: Ministry of Agriculture (No. 20)

#### 1.5.2 Promis

An a raw Unterial for fats and oils, poanuts have greater promise than soybeans. Table 3.4.2 below shows the production of peanuts in Vietnam from 1965 to 1973. Out of the total production of 44,000 tens, in 1973, the Delta's 16 provinces accounted for only a fractional 520 tens in 1973. Other parts of the Southern Region accounted for 34%, the Contral Lowland Region accounted for 39%, and the Central Highlands, 6%.

# Teble 3.5.2

Year	Area (he)	Production (Os tons)	()E(he)
1965	31,720	32,595	1.038
1966	10,640	34,420	1.123
1967	10,190	13,710	1.119
1968	<b>19,68</b> 0	32,055	L.allo
1969	31,295	14,410	1.100
1970	30, 240	12,185	1.064
1971	14,40D	17,000	1.076
1972	15,300	18,900	1,105
1973	<b>19, 200</b>	44 <b>, 8</b> 00	1.141

## PRANUT PRODUCTION IN VIETNAN -- 1965 - 1973

Saure: Hinistry of Agriculture, Sauree No. 20

Peaket cultivation does not compote with paddy cultivation because the peaket grave in namely, well-drained, and generally peaker solls where paddy cannot be grave. Peakets have a long history of production in Vietnam and peaket utilisation is well Mann, According to the report of the 1971 Chinese Agricultural Technical Group (via Source No. 48), of the total production of 25,590 tons in 1970, two thirds was command directly as food, and about one third, namely 8,307 tons, was used as raw untertail for all extraction, yielding 2,900 tons of peaket all. This calculation is based on the assumption that about 70% of the total peaket erop is shelled and the oil context is about 50%. The residual peaket code, after extracting the oil, is used as fortilizer for the cultivation of tobacce or as feed for animis or as raw untertail for the manufacture of may names. The variety of peakets being grown in Vietnam is not ideally suited for eil extraction and thus is similar to severan in the sense that the main utilisation is for human communition. However, the productivity per hectore is greater than that of severans. Under these circumstances, peakets should be given greater emphasis as a raw material supply source for agre-industry. However, we cannot foresee peakets becaming a probising raw material for agre-industry on the Tay Do Industrial Park because of the Dolto's low estput.

# 1.5.1 Coconsta

Table 3.5.3 shows the production of cocanuts for 1965 through 1973. Of the total production of 133,600,000 cocanuts in 1973, the Dolta accounted for 816. The Contral Lowlands, accounted for 1966.

The main growing areas in the Bolts are in Kien Hes and Dish Tueng provinces,

# Table 3.5.3

COCOMPT PRODUCTION IN VIRTHAN 1965-1971

Yoar	Area (ha)	() () () () () () () () () () () () () (	Yield ( <u>loop mit</u> e)
1965	48, 190	147, 110	1.648
1966	19,015	129,460	1.118
1967	38,220	1 10, 100	1.5.25
1968	29,905	110,705	1.702
1969	12,981	48, 141	
1970	32,290	110,5%	1.471
1971	33 <b>, 18</b> 0	125,000	1.711
1972	<b>39, 18</b> 0	116,100	1.404
1471	11,440	£ 11,6000	1,992

Baseco: Hinistry of Agriculture, Baseco 20

Viotnamo cocannet production is, however, very anall as compared with the Philippines(7.8 billion cocannets), Philiand (900 million), Mulaysia (830 million) in 1970. It is clear from the Vietnam statistics (Table 3.5.1) that the use had a seriously advorse effect on cocannet production. The compactively poor draimage of the Deita solis is not ideal for cocannet trees and the trees are standed as in the case of bandon and mager case. Number the constant is usually considered as one of the most important raw materials for agro-industries in Gropical climates. The oil can be consumed both as a food for human consumption and as a raw material for industry. The residual cake can be used for mixed feeds. The fiber (called coir) can be used as the raw material for rope and matting: the coconut milk can be used as pure drinking water, and the shell can be used to manifacture activated carbon and shell powder. The copra, prior to extraction of oil, can be used as desiccated coconut, a raw material for the food and confectionary industries. The sap from the coconut tree can be formed to make an alcoholic beverage. Thus coconut derived interials have a wide variety of applications.

Assuming that at least 100 grams of tats and oils are extracted from a single count, the total count production of Vietnam could in theory vield 11,80 tons of oil. However, in practice, mory commuts are commund as food or beverages. In the Philippines, less than built the counsit crop is being used for oil extraction. Although data are not available, we believe that only 20% of the communits grown in Vietnam are used for oil extraction.

In the Delta, coconst all is not used as conding all whereas other southeast Asian constring fallow the practice of using coconst off for cooling. However coconst all is computed for the magn industry in Vistnam and some is being imported trom Indonesta via Singapore to supplement local supplies (Truck-Macstan Scap Co., Saigner).

forders pre-events constant constants on the pre-even to get constants pre-events to even and the pre-events to the term of te

The transment concerned production is bighty desirable tor listened but will be difficult to efficie due to such factors as the liberear period required for trees to metric and the replanting of trees don't reval doring the war. There are only a ten suitable creas for remonent trees to green in the bella is the statistic or angest when the sectors that the comparatively shall velocies of raw material and the scattered leveline of tenes would pose supply problems for a mederic plant in the term

#### 1.1.4 Break Break

The president into a construction in the reason becaused in the reason of the reason of the reason of the production of the production of the production of the production of the product of the term of the the term of term of term of term of term of terms of term of terms of term of terms of te

#### 1.5.5 Mannh

Unto on Kopek production to limited, but most of this product to cultivated in the Sauthorn Negion of Victuan, but not in the alluvial plains of the Nolta observice transmission stifus from eccentive Moisture. Total Vietnam production in 1973 was only 500 tans (Bibliography, 193. Rapak can be a valuable expert product in tropical countries. The application of the fluffy, cotton-like material is for the manufacture of such products as pillows, cushians, matress fillers, life jackets, sleeping bags and insulation material for odd weather garments.

The hard wood parties of the kapak ball can be crushed to make a high quality edible oil or industrial oil. Kapak trees do not grow in paddy lands and therefore would not compete with the Delta's principal crop.

After oil extraction from the hard seeds, the residual cake Online an excellent raw unterial for animal feed.

Since Kapak is not available in large volume, we believe this Enterial cannot constitute the basis for a medern agro-industry in Cantho.

# 1.5.6 Carp

Tables 3.5.6-A and 8 abov Vietnam's production, experts and imports of carm. Of 50,500 tans of carm produced in 1973, the Delta accounted for 25%, that is 12,405 tans (Table 2.3).

#### Table Ja5.6-4

Tear	Aren (ha.)	Product ( an (Se ( ann )	Vield (ML/ha)
1963	<b>76,180</b>	43,830	1.211
1966	29,180	19, 190	1 1
1967	<b>38,95</b> 5	11,820	1.104
1968	<b>38,78</b> 0	11,760	1.104
1969	38, 165	10,515	1.000
1970	38,640	14.415	1.048
1971	11, GRD	11,790	1.000
1972	11, 180	41.700	1.175
1971	19,680	10, 100	1.375

VIETNAME PREMICTION OF COM -- 1965-1971

Source: Nonistry of Agriculture, Source No. 20 Ann. P

VIETNAMER	EMPORTS AND EXPORTS OF CORN - 1960-73			
Jaar.	Quantity ( <u>Quantity</u> )	<u>Yalue</u> ( <u>1,000 YW</u> S)	Quantity ( <u>0. tons</u> )	Value ( <u>1_000_VN</u> #)
1960	1,445	2,714	1,549	5,878
1961	1,117	1,917	36	68
1962	227	395	5,600	23,674
1963	**	ŧ4	95,500	216,212
1964	11	**	26,800	54,752
1965	Ħ	**	43,000	149,855
1966		<b>+1</b>	33 <b>,00</b> 0	140,580
1967	*	++	20,000	132,100
L 968	*	+1	41,000	236,447
1 969	*	++	82,540	<b>597,</b> 755
1 470		Ħ	116, 964	988,191
1971		*	70,885	631,357
1972		Ħ	99,705	2,762,066
1971	•	*	77,177	4,636,535

# Table J.s.6-1

Source: Directorate of Customs, Source No. 30 in Hibliography.

The noil condition of the bits does not noon expecially miled to the cultivation of cars and other areas of Vietnam are better mited.

Although the production of corn in the Dolta is limited, it is one of the important raw unterials for the mixed feed industry.

An industry dusigned to extract fats and oils, protein, and starch cannot be planned on the basis of the Delta's Geoger ogen graduetion alone and would require supplementary ray exterials.

## 3.9.7 Innel and date

The nords of these (there crops can also be used for edible oil, but the volumes in the bolts are small and could not, as in the case of corn, be counted on as the ray unterial base for industry. These products are discussed in Section 3.37 (regarding gammy bage) later in this chapter.

#### • 👪 •

#### 3.6 Animal Feedstuffs

Animal feedstuffs (sometimes called, "animal feed supplements") have been imported largely from the United States under the AID program and the PL 480 program. The principal materials imported in recent years have been: meat meal, fish meal, soybean cake, alfalfa meal, bone meal, and corn. We were told by USAID that imports of corn were being discontinued because low cost corn imports act as a disincentive to the growing of sorghum (See Section 2.3 above). Imports of animal feeds have been as follows in recent years. (Source No. 9 in Bibliography),

Fiscal Yoar	<u>PL480/I 9/</u>	USAID b/	Vietnam's S	TOTAL	
1967	0.	0.	0.055	0.055	
196 <b>8</b>	1.248	0.	0.661	1.909	
1969	2.862	0.	6.822	9.684	
1970	7.688	0.	5.060	12.748	
1971	8.086	5.527	3.417	17.030	
1972	9.136	7.300	0.706	17.222	
1973	5.530	1.748	0.664	7.942	

Imports of Animal Feeds in Millions of U.S.S.

🖌 Corn

b/ Meat and bone meal, soybean meal, alfalfa meal

c/ SBN O61 - feeds and fodder through 1972

The currently high cost of imports, about \$500 per ton, is forcing the commercial hog and poultry raisons to curtail their operations and the West German-financed slaughter house in Saigon (VISSAN) is finding that their meat prices are becoming non-competitive on world markets, according to our interviews with the management.

The price situation means that hog and poultry raisers are turning more and more to domestically produced feeds, i.e. especially rice bran, sorghum, broken rice and low grade brown rice, and fish meal.

The currently tight supply of feed on world markets and the high price is making it difficult for the Delta to even maintain the existing level of livestock raising.

The Delta however has a high potential for becoming self supporting in feed. The Delta should be able to raise livestock and poultry at low cest assuming that food self-sufficiency can be attained. The Delta can theoretically produce important feed raw materials such as: (a) rice bran, (b) corn, (c) serghum, (d) maniec, (e) oil cake, (f) molasses, (g) fish meal, (h) bene meal, (i) feather meal, (j) the residual materials from fruit processing, (k) vegetable processing wastes. Sorghum could become one of the most important crops in the Dolta largely due to the promotional efforts of USAID. (See Section 2.3 earlier in this report).

The Dolta production of sorghum accounts for more than 9D percent of the Vietnam's total production, and about 21,000 tens were produced in 1972-73. Although the current output is still small, production can be increased rapidly and will probably become one of the most important raw materials for the mixed-feed industry, playing a role similar to corn.

In order to make full utilization of these ray material resources, a mixed feed industry should be established in the near future. Lifestock, poultry and freshwater fish culture on a large scale in the Delta vill become possible only by the establishment of a mixed-feed industry. This will be discussed later in greater detail in Chapter 4.

#### 3.7 Dairy Products

There are almost no dairy cattle in the Delta, except that the water buffalo can produce small yields of milk. The population of these animals was as follows in 1973 in the 16 provinces of the Delta (Hibliography, 19).

Buffalces	194,4 <b>R</b> 0	(14% reparted slaughtered each	ve ar i
Draft and Deef Cattle	246,140	(196 reported slaughtered each	year)
Hegs	2, 145, 140	(81% reported simplered each	year)

The absence of grazing land and the high cost of food, would make it difficult for a dairy products industry to be viable in the Bolta.

Nilk is, however, appreciated and condensed and powdered wilk- are being increasingly communed and imported; 23,190 tons of condensed and powdered milk, valued at \$11.6 million were imported in 1972, and the imported volumes have probably rises substantially since them.

Imports of all dairy products including powdered and concentrated milk, butter or anhydrous milk fat, cream and superimed condensed milk, were as follows in recent years;

- 🖬 -

	America of Joiry Products, PY 71-73			
Bear co of Places lin	<u>PT 71</u>	<u>PL 72</u>	<u>FT 73</u>	
United States PL 180	\$13.3	\$12.0	91.3	
Vietnan's Chin Pareign Exchange	3.4	11.7	14.4	
uba ID	••	••	1.2	
Total	18.9	23.7	19.1	

Despite the most for a datey product. Industry, areas of Listnam other than the Delta vould be far better suited as a site.

#### 1.8 Mat Pressellet

There is strong varie domain for park, chicken and in certain countries a domain for dock, all of which are being produced in the belts by formers and by a few unders positry and hog forms rateing upgraded species. The latter, however, have almost dust down their operationor are facing financial uncertainty using to the high price of feed. (Par production, we Table 2.1, items rathed 2, 7 and R1.

Appendix B, whering expert markets, indicates that Vietnamis 15 most important trading partners imported processoric parties valued at \$995.3 million in 1972. West Gorman imports accounted for \$226.5 million or 706 of the total imports.

As in the case of processed fruits and vegetables, there is a negligible domant within Viotnam for processed meats. This is because from chidnen, duch, and park are available the vear-around in public furthets and because communes would not be propared to pay higher prices for processed and packaged meats. (Preferences will doubtless change over time, assuming a continued improvement in the and per-mal incomes, 1

However, the intermediate-term goal would be expert maket- for processed meats. As most sened, there is a modern staughter houses and most processing plant mear Batgon called VIBBAN (the full non- is Vietnam By-Highe MacsBan). The plant monogeness people and German advisors - had different opinions, but the conservate was that dask and chicken experts (the latter, only if species are upgraded) and park experts had promises. Sepanses and German importors profer frozen meats, but there is a varial method for canned meats as well. Nowever, even the export market for chicken and pork cannot be penetrated at present because of one conclusion that emerged from our visits to VESSAN and to a commercial hog raiser. The conclusion is that exports from Vietnam are not competitive now and will not become so until the price of feed decilines.

Therefore chicken and pork processing should be a future goal, but not pursued until the creation of a successful mixed-feed industry.

Expert of frozen duck is a more readily attainable goal, Given the very large flocks of ducks in the Delta, the absence of a feed requirement (ducks feed on insects and plankton), it may be possible to organize develop and industry; see Chapter 5.

#### 1.9 Slaventer Hause

A staughter house at the Tay Do Park in Cantho is a very desirable tong run objective. Aside from meat, such a facility could supply house and meat meal with a high protein content similar to that at fish meal, i.e. over MPA protein. The lips, ears and other parts could be used for glue monifacture and the lipinge of stomachs and other internal organs can be used to monifacture enzymes. However the visbility of the stanghter boxes usual depend on the shility of bayers to bring animals to the slaughter boxes at a price low enough to well competitively on the market. With the current obsertage of animal ford adverwely affecting the hog raisers, we think a slaughter boxes usual face difficulties in obtains ing enough raw material.

There are two principal slaughter practices in the Vietnam, are is controlled slaughter under Government regulations and the other is unscantrolled slaughter usually performed in rural districts. Of the 1.8 million bogs slaughtered annually (Table 2.3), we estimate that about 1.0 million or only one fourth are slaughtered under Government controls (0.0100, phys.55).

A claughter hause in Casthe would have to compete with the uncentrolled slaughter by paying higher prices for hoge which could in turn raise its selling price of park or heaf cattle to non-competitive levels. The VISSAN company (slaughter house near saigns; see Section 3.8) is operating at only 30% of capacity and the main reason is that it must buy animals at high prices and then try to sell most competitively on the market.

We Learned from the Ngme-Thank PiL Preoding Farm in Cantho, that promium hoge (Veriabire - 7%, 5=1) were solling in Ortabor for 7%,(ND plasters per 100 kg. The name purchase price of 7%,(ND) plasters was quoted to us by VIBBAN, the slaughter bases near Salgon. The open method price for live hoge in Cantho in Ortaber 1974 was 7%,(ND), or about 105 less than the price paid by VIBBAN. The pig farm in Cantho told us that to raise a hog to a weight of 100 Kg. in Cantho, cost 65,000 plasters in terms of feed alone in October 1974. Interest on debt, overhead, wages and profit could not be covered by the 10,000 plaster difference (75,000 = 65,000 plasters). The main reason for this was that the wholesale price of meat has risen more slowly than the cost of feed. Fine rice bran is one of the elements composing hog feed, usually about 20% of the total. The increase in fued prices is similar to that of other elements. Necent trends in the relationship of meat prices to fred prices show that feed and most of the feed components) rose 67% from January to October, 1974 while the wholesale price of live hogs rose only 4% as shown below:

		Wholesale Price of Live Hosp	Cent of Band		
Jan .	1974	VN\$7:20/kg.	VN\$ 90/ka.		
May	1974	VN8740/kg.	VN\$110/Kg.		
Ort .	1974	VN\$730/Kg.	VN\$1 %0/Kg.		
Incre	an from				
Jan.	to Ort.	· •	+67%		

Until the prices for park and feed are brought into a better alignmont, and until the uncontrolled slaughter practices are sharply reduced, a modern slaughter houses at Tai is usual probabily not be viable.

If there were more forme like the Ngac-Thank Fig Wreeding Form in Cantho and if much forme could operate viably, then a slaughter house would be assured of example raw material. However, woder the present circumstances, a slaughter house would have to draw on widely scattered, individual formers to obtain raw material. A large scale modern slaughter facility would probably be underutilized. We believe that a solution to the feed problem (see Chapter 5) should come tirat.

The major center of most commutation in Vietnam is the city of Satgen and its convirging. We believe that since the transport of live cattle 170 hms to Satgen in comparatively inexpensive, it would be dure economical to mapply most from slaughter bounse in Satgen uncredit fact there already is a major, molern slaughter bounse (VISSAN). Transport of cattle from Cantho to Satgen costs  $L_1(OD)$  planters per LOD by as LO planters per by so live than 2 Md per by.

#### 1.10 fammers

An montioned in Nection 3.7 above, the cattle population of the Neltain 1: provinces was Job, 24:1 ii 1971 (\*). Of these animals, 31,070 were claughtered. The claughter figure is about 196 of the total which miggets that nome of the slaughter may not be reflected in the official statistics. Whe buffale population was 196,680 in 1973 and about 196 are hilled yearly.

(\*) How Nesser (# 14 )rs Hitelings ageby

In any event, the slaughter of cattle occurs ever a wide area, rather than being concentrated in a few single facilities. Obtaining raw material for a tannery would require collecting hides from a wide area and rapid transportation to the tannery to avoid bacterial action and spoilage.

In the case of pig mkin, the same problems would arise and the uncertainty of viable, large-scale hog farming, mentioned in Section 3.9 would make the supply of pig mkin raw material questionable for the tannery.

#### 3.11 Livesteck Ranching and Poultry Farming

In Cathe, there is a modern, accentifically operated hog form an montioned. There is also a chicken form raising birds bred from imported stock. (Leghorns, Plymouths, Node Island Neds). Noth arr impressive facilities. The former is facing financial problems because of the high price of feed and competition trom the lower cost hogs, bred and slaughtered in the traditional manner.

The chicken ranch in Cantho (Huynh De at Hinh Dhuy, Cantho) was barely in operation in October, 1974. The chicken farm raised a peak of 20,000 birds in 1973, but the number fell to about 2,000 in October 1974.

Form-riy, the outer used to export frozen chicken to Thailand and Singapole, but because of the high price of feed, these operations have been discontinued.

There is no mointantial cattle ranching in the Delta. This is because of the shortage of grazing land and the predominant use of land for paddy. (Rhor places in Vietnam north of Saigon are better suited for grazing because of the available pastures.

Given the a solution to the problem of feed, poultry and hog relating should become viable.

Buck farming is being undertaken by the small holder farmers. As montioned in Section 3.7 above, duck can thrive an the inserts and planktan of the rice publies and therefore in not require feed. We now many herde of ducks numbering themands of birds each, changing highests and in publy fields. These ducks already constitute the basis for one of Vietnamin major experts namely duck feathers valued at U.S.Sb21,675 in 1972, UBSE, 3 million in 1973, and \$525,541 in January-June 1976, (data from the Expert Development Contor, Seigen), Neck forming is a viable industry.

#### 3.12 Processing of Aquatic Products

The Dolta has a major potential for fresh water and marine products. Table 3.12-A shows the number of persons engaged in fishing, and the number of boats powered and manually propelled. Table 3.12-B shows fresh aquatic products and Table 3.12-C processed aquatic products. Tables 3.12-D and 3.12-E show further data on the fishing industry and processing for 1973.

The tables show that the Delta produces 87% of the fresh water fish of Vietnam. The volume of fresh water fish caught is about the same as the catch of marine products in the Delta. The Delta also accounts for a high percentage of the output of shrimp and mollumks as well.

These aquatic products of the Delta contribute greatly to the animal protein consumed in Vietnam. We believe that the inhabitants of the Delta are producing large volumes of cultured, fresh water fish which are not adequately reflected in the statistics. The acquatic products of the Delta constitute resources ranking along with paddy as a major national asset, and they hold promise for a variety of processing industries.

#### 1.12.1 Fishing Operations

Fishing operations for marine shrimp and fish are centered in tishing parts in Kien Giang and Hac Lieu provinces. The fishing part of Bach Gia in Kien Giang province is the Largest. With the current marine fishing methods, it will be difficult to increase production further. It will be necessary to expand the fishing area by increasing the size of vessels, equipping them with somer, radio communication gear and freezing equipment both an above on board the vessels. Shrimp could probably cultivated in coastal waters to increase the catch whatantially.

The fresh water fishing is especially successful in An Atanu and Chas Dar, but such fishing is generally practiced elsewhere in the Belta as well. There are special local methods for the culture or farms ing of fresh water tish, and thus overfishing of tresh waters in the Belta is not a problem.

> (Norrative continues after Tables 1.12-4 through 0.)



# 82.09.22





OF



¥

24 ×

## Table 3.12-A

Year	Fisherman	Fishing Boats	Motorized Boats	Non- Motorized Boats
1961	198,000	39,000	-	-
1962	198,610	35,600	3,600	32,000
1963	210,000	41,500	6,500	35,000
1964	245,520	56,470	9,710	46,760
1965	159,331	42,177	4,324	37,853
1966	171,480	41,737	32,360	9,377
1967	270,408	76,190	23,195	52,995
1968	272, 304	79,959	29,968	47,991
1969	277,118	81,956	39,001	42,955
19 <b>7</b> 0	317,442	88,209	42,603	45,606
1971	335,690	91,424	48,842	42,582
<b>197</b> 2	342,797	95 <b>,0</b> 62	55,247	39,815
1973	354,894	92,265	63,709	28,556

# DATA ON FISHING OPERATIONS IN VIETNAM, 1961-73

Directorate of Fisheries via Source No. 20, Bibliography.

------

## Table 3.12-B

# PRESH AQUATIC PRODUCTS OF VINTNAM - 1960-73

Year	Sea Fish	Presh Water Fish	Total	
	ton -			
1960	-	-	165,000	
1961	-	-	250,000	
1962	200,000	55,000	255,000	
1963	298,000	34,000	<b>332,00</b> 0	
1964	345,000	52,000	397,000	
1965	318,000	57,000	375,000	
1966	315,790	64,710	380, 500	
1967	351,200	59,500	410,700	
1968	358,995	51,045	410,000	
1969	400,171	63,673	463,844	
1970	530, 310	74,140	577,450	
1971	516,424	71,066	587,490	
1972	501,278	81,772	583,050	
1973	508, 537	91, 342	599,879	

Directorate of Fisheries via Source No. 20, Biblingraphy

Ì

- 95 -

-----

## Table 3.12-C

Year	Fish Sauce (000's of liter	Dried Fish ) (MT)	Salted Fish (MT)	Canned Fish (MT)
1960	-	-	-	194
1961	-	-	-	235
1962	-	-	-	140
1963	-	-	-	234
1964	41,116	16,011	17,365	214
1965	42,650	11,264	16, 322	232
1966	53,783	12,329	17,557	187
1967	61,000	15,000	35,024	182
1968	59,000	20,205	24,830	80
1969	60,850	20,769	30,242	20
1970	64,192	27,981	34,425	-
1971	69,830	41,608	36,728	-
1972	68,737	46,050	43,327	-
1973	64,480	45,993	45,035	-

# TONNAGES OF PROCESSED AQUATIC PRODUCTS -- 1960-73

Source: Directorate of Fisheries via Source 20; Mibliography

.

- 96 -

. . . . . . . . . . . . .

-----

## Table 3.12-D

# FROCESSED FISHERY FRODUCTS AT FISHING PORTS (1973)

1

	Dried fish, sharks fins & bladders	Dried Sh- rimp	Dried cuttle fish	Steam- ed salted fish	Salted fish	Ground salted shrimp	Total	Fish sauce (1.000 Lit.)
Vietnam Total	39,307	5,665	1,021	20, 576	24,459	1,965	92,993	64,480
Southern Region	28,603	5,471	252	8,026	17,179	1,648	61,179	24,511
(Delta)	(12,912)	15,337	) (150	(6,268)	(14,096)	(539)	(39,322)	(23,475)
(Delta %) Vietnam	(33%)	(94%)	(19%)	(30%)	(58%)	(27%)	(42%)	(36%)
Central lowlands	10,704	194	769	12,550	7,280	317	31,814	39,969
Central highlands	-	-	-	-	-	-	-	-

(Unit: Tons)

Source: 19 Mbliography

## Table 3.12-E

# PRESH FISHERY PRODUCTS AT THE FISHING PORTS (1973)

(Unit: Tons)

	Marine fish	Fresh Water fish	Shrimp	Cuttle- fish	Mollusks	Total
Vietnam Total	<b>508,</b> 537	91,342	62,085	12,922	38,710	713,596
Southern Region	206,255	83,155	55, 569	7,819	30,967	383,765
(Delta)	(78,657)	(79,756)	(39,603)	(723)	(27,680)	(226,419)
( <u>Delta</u> %) Vietnam %)	(15%)	(87%)	(64%)	(6%)	(72%)	(31%)
Contral lowlands	302,282	8,187	6,516	5,103	7,743	329,831
Central highlands	-	-	-	-	-	-

Source: 19 Mbliography

## 3.12.2 Fish Culture

Relatively shallow water and good sunshine produces large volumes of plankton. These factors together with fresh bran from the rice mills creates good conditions for fish culture in the Delta's waterways.

If similar efforts were made to practice eel culture, the exports could become very successful since the Japanese consider eels a delicacy and consume eels in large volumes.

We believe some comment is warranted in this report regarding fish culture. This is because Dr. V. R. Pantalou, Fishery Advisor to the Mekong Secretariat in Bangkok urged our survey group to inspect methods of fish culture in Delta and because the outlook is promising for industry.

The method of culture is referred to in the literature as "cage culture". The species of fish vary, but a common fish is the <u>cyprinus</u> <u>carpio</u>, which sold last fall in the Cantho fish markets for 800 plasters per kg. or \$1.17 per kg.

The fish cage is an extension of house boat, moored in shallow water along the river bank. The cage we inspected was  $4 \ge 6 \ge 2$  meters, about the same size as the house boat occupied by the fisherman and his family. (Dr. Pantalou thought the cages could be substantially larger,  $16 \ge 6$  meters; this size cage could not be handled by a single fisherman and his family but could be operated in conjunction with a freezing plant.)

The  $\frac{4}{x}$  x 6 x 2 meter cage which we inspected contained 25,000 fingerlings, 10 cm. long when they were first cought. When sold on the market 10 months later, the fish will weigh 0.2 kg each and be 20 cm. long. The fisherman pays about 6,500 (U.S.\$9.50) piasters per day for feed (course rice bran, pumpkins, vegetables) or a total of 1,950,000 piasters (US\$2,847); he sells his 5,000 kg. of fish on the market for 3,500,000 piasters (US\$5,109)

The oxygen supply imposes an upper limit on the number of fingerlings as they grow in size and are packed densely in the cage. However, the oxygen supply is better in flowing river water than it would be in sea or lake water which is not moving. As mentioned earlier (Section 2.1), the Delta fish culture practices may be unique, are very promising, and could become the basis for frozen fish industry for export to Hong Kong and Singapore. Dr. Pantalou believes that the Delta fresh water fish is more appreciated in Asian markets than in Europe or the U.S.

We have made tentative calculations about the feasibility of culturing fresh water fish in rafts in large enough volumes to supply a fish meal plant at Tay Do. We have concluded that (a) the suitability of fresh water species for fish meal would have to be confirmed, (b) the capital required to build enough rafts would be substantial (e.g. To supply a 100 TPD paint would require 962 M of fish per day or 289,000 M per year weighing 30,000 tons. To raise such volume would require about 1,000, 20M x 6M x 2M rafts working on a 10-month cycle. 1 M of fish weighs 104 kg. Plant would require 100 tons fresh fish to produce 20 tons fish meal per day or 6,000 tons per year. For edible fish, culture should be efficient (see  $h_{1,9}$ )

## 3.12.3 Cold Storage and Refrigeration

There are ice plants in almost all villages and, of course, in urban communities. The latter are fairly well equipped and efficient. (The Brasserie et Glacière de l'Indochine in Cantho has very modern facilities). These plants are supplying ice largely for the transport and preservation of fresh fish.

Freezing facilities for shrimp, fish, and clams to be exported have been established Rach Gia, Kien Giang Province. Plants for domestic distribution (through a cold chain, for example) have not yet been created. The domestic market for frozen fish or shrimp is not promising. The reasons are: (1) The distribution of frozen fish requires chilling equipment and also a complete system including freezing, refrigerated storage and transportation, sales channels, and defrosting both in commercial establishments and in private residences. Under present conditions, such elaborate facilities would be too costly. (2) Fresh fish are generally available in local markets at reasonable cost throughout the year and are generally more palatable than defrosted fish. Frozen fish differs from frozen meat in regard to the deterioration of flavor after defrosting. (Frozen fish is regarded as inferior in guality to fresh fish in Japan.)

The export market for frozen aquatic products is far more promising than the domestic market. (See Appendix B).

#### 3.12.4 Processing

Drying, salting and smoking are inexpensive methods for preserving marine products, but in Vietnam where there are numerous rivers and canals and a long coastline, people in most areas can generally obtain fresh marine products and thus the demand for processed products is very limited.

There is a small factory in Rach Gia equipped with facilities for drying, but sales are not promising at present.

Canning potential for the domestic market is not promising in view of the abundance of fresh fish in most areas, plus the cost of cans (See Section 3.3.2, para c). However, an export potential does exist if black and tin plate were imported. Freezing is the best process.

## 3.12.5 Fish Meal

In view of the recent world wide shortage of fish meal caused by the decline of Peruvian anchovy, the fish meal industry developed rapidly in the Delta. It is estimated that the annual fishmeal production in Vietnam is about 20,000 tons. But with the 1973-74 improvement of Peruvian production, several fish meal factories in Rach Gia and nearby Phu Quoc Island have been shut down due to the non-competitive cost of their product. The industry has also been troubled by the high cost of

- 99 -

fuel and insufficient raw material. Fish meal is an important protein ingredient for the mixed feed industry in the Delta. It will be necessary to procure the needed volume of fish meal if necessary by providing special incentives for the industry, since fish meal is of prime importance as an import substitute for mixed feeds (See Chapter 4).

# 3.12.6 Fish Sauce

Fish sauce (<u>nuc maum</u>) is an important sector of the food processing industry in Vietnam. It is regarded as one of the necessities for food preparation (like soy sauce in Japan and China or ketchup in the United States). Production of fish sauce is quite large, but the factories are small scale and widely scattered except for the Rach Gia area and Phu Quoc Island. The processing is based on traditional artemnal methods. (See above Table 3.12-C, for production.)

It will ultimately be possible to establish a modern, large-scale factory as was seen in the evolution of the soy sauce industry in Japan but preferably at Rach Gia, not Tay Do.

## 3.13 Peanut Processing

This subject has been discussed in Section 3.5 above.

## 3.14 <u>Nood Pulp and Other Materials for Pulp</u>

The utilization of rice straw as a source of pulp was discussed in Section 3.2.4 earlier in this report.

There are two other potential sources of pulp in the Delta, the mangrove forests in brackish water along the southeast coast line and the tram forests chiefly north of Rach Gia on high and where waters are fresh and the soil clayey. (Tram is sometimes called, "Climax Type Mangrove".) (The scientific name of the principal mangrove species is <u>Rhizophora conjugata</u> and the scientific name of the Tram is <u>Melaleuca leucadendron</u> -- Source 54-A).

## 3.14.1 Mangrove Trees

Mangrove trees grow in deep mid consisting of alluvial clay and brackish water. There are an estimated 329,000 hectares of mangrove in the Delta. (Another estimate is 500,000 ha.) Typically there are 10,000 mangrove trees per hectare and one hectare can yield from 50 to 100 m<sup>3</sup> of wood. The mature mangrove tree can be 5 to 6 meters in height and 10 cm. in diameter after about 10 years of growth. However, in the typical mangrove stand, the height of the trees is 2 meters and the diameter of the trunks range from 4 to 7.6 centimeters (Source 54-A in Bibliography). A continuous supply of mangrove could be obtained by a 10-year rotation. For example, by cutting 33,000 hectares per year, about 2.5 million  $\mu$ 3 of wood could be obtained. In harveeting mangrove, it is necessary to replant seedlings in the clear-cut areas.

Logging operations for the mangrove trees are often thought to be fairly simple and inexpensiva since the trees grow near the coast and roads and bridges are not required for logging. However, there are many difficulties encountered in actual practice. One of the difficulties is the fact that the denaity of mangrove wood is about 1.05 (Bibliography 54-A); the logs do not float and cannot be hauled through the water as rafts or logs. Therefore the collection and landing operatione entail substantial cost. A chip factory would have to be located at the logging eite to reduce transportation costs.

Mangrove logging operations elsewhere have not proved to be economically viable. There are two known attempts to harveat mangrove trees for pulp -- one in North Borneo where the Japanese company, Oji Paper Co. Ltd., abandonned operations and the other in southern Sumatra where Japan Inter Co. Ltd. also failed.

Some of the reasons for the difficulty of mangrove logging operations are:

(a) The ebb and flow of the tide permits logging only during the low tide period. (b) Along the coast line of the Delta area, logging eperations would be jeopardized by ocean waves which roll far inland during typhoons. (c) The shallow water depth for some distance out from the shoro precludes the approach of large vessels. (d) An economical chip plant would require a pier, the construction of which would be difficult due to the unstable soil and the need for extensive piling. (e) Since the mangrove trees grow in fairly narrow stande, the siting of the pier would pose probless; namely; as the adjacant stands of mangrove were harvested, the pier would become more and more remote from the sources raw material. This problem might be solved by a special purpose vessel on which the logs could be processed into chips. However the mechanisation of harvesting atill remains an ebstacla. (f) The mangrove tree is met an ideal pulp wood because of its density and the heavy chemical costs involved in pulping proceases.

The manyrove wood is now used for:

Charcoal Firewood Fishing Poles Raftera Tannin (for dyes)
### 3.14.2 Utilization of the Tram

Tram trees can grow to large size but are usually lumbered off when the trunks are 8 to 15 cm. in diameter. The atands of tram trees are estimated to cover an area of 200,000 hectares, but the totsl volume of wood of the tram tree is small. The tram is used for the following purposes:

> Fuel wood Charcoal Rafters to support the roofe of houses Piles for houses built on soft earth where flooding is likely Cork substitute, made from the bark

During our field survey, we did not inspect the tram forests, but discussions with a forestry expert (Le Viet Du, Director of Waters and Forests) and an inspection of the literature (Bibliography 54, 54-A<sub>j</sub> and 54-B), lead to the conclusion that the tram should not be considered a source of pulp until such time as reforestation and aforestation programs are well established. The pulp outlook is unfavorable because the stands of tram trees are uneven and would not lend themselves readily to mechanical harvesting and because the tram is being lumbered off quite rapidly to meet the growing demand for construction, chiefly for piling. For construction timber alone, some 20,000 m<sup>3</sup> of tram was being consumed in the 1960's (Bibliography, 54-A) and because of population increases since then, greater volumes are probably being consumed today. A sudden depletion of the tram forests for pulping purposes would give rise to a requirement for a substitute piling material, probably brought into the Delta from outside.

### 3.15 Agricultural Implements and Engine Repair Services

### 3.15.1 Engine Repair Services

There are some primitive, small scale repair shops and metal working shops in Cantho. They are operating without the benefit of modern tools, equipment and measuring instruments. Given their handicaps, it was surprising for to us see the quality of repairs being undertaken thanks to the ingenuity and diligence of the operators.

Regardless of their skille, however, it is impossible for these shops to repair engines, especially diesel engines, without precision equipment such as test benches for f.i.e. and nozzle grinders.

The number of diesel engines fitted on four wheeled tractors, boats and trucks and buses in the Delta is estimated to be 41,120 (See Table 3.15-A).

(narrative continues after tables)

### A PLAN WAR

## milestics-1975 (Civil alz) a Constant in Mile In

۶Ì

and the second se	2	<b>1</b>	<b>F</b>	F.	<b>2</b>	<b>F</b>		2			Contin 1
	9	F	3.92	ŧ	2,000	<b>3</b> 7		5.00		1	
trates tructure	•	ŧ	ş	K	1	<b>8</b> , -	3,43	2,063 (		2	261.1 UM
	2.947	5	ä	<b>.</b> .	<b>.</b>	Ŗ	13. <b>8</b>	£		1	2 <b>16</b> 15
Manil and a	<b>1.690</b>	53.09	8.698		<b>010</b> , 01	12,000	2.5	N. N.	<u>.</u>	8	540,15
Sub-To tal	161,11	¥	13.74	<b>632'''</b> '1	13,800	<b>16. B</b> io	104,103	6 88° 42	<u>B</u>	2	
Privately owned tructes b)	ŝ	5.	2.117	2,911	(a <b>) (</b> (	1. <b>386</b>	1. A	9.230	- <u>M</u>	8	x() 6.000
j					( <b>#</b> 2	1000		ş	Ŋ	3	3
Tatel.	12.312	2.2	16,000	00.12	15.35	16. JU	500°.511	36.4			11.12 11.12

### Ï

- 103 -

## Other figures estimated by FFL or provided by AUDACD agents for Lidicate. in Seigna-Seuros: 1968-71 from MailD Agr. Dag./AUPA Seigna. 3

- beed on new truck registrations as reported in the <u>Vietnes Statistical Vestions</u>, 1972 Mational Institute of Statistics, Said 3
- According to the official import statistics. <u>Vietnem Statistical Yearbook 1972</u>, in 1972, the total imports of tracks usighted 4.000 buse. The GTV of a 5-tem perjowed track is 9.2 teme; of the 9.2 teme; the body wight is usually about 4.2 teme. Next of the imported tracks are 5-tem perjowed tracks, and tracks are imported as chancies only. Therefore we are assuming the wight is 3.5 teme (4.2 teme leas 0.7 teme of local content = 3.5 teme). Therefore, 4,000 teme(3.5 teme = 1,396 tracks. The figure for 1973 is a FPL estimate and is not besed on official statistics. 3
- () This is a PPL estimate.
- The actual towards of bus channels imported une 477 (1970), 1,401 (1971) and 628 (1972). We are anomaling the body weight is 3-5 terms and have divided the towards figures by 3-5 terms. •
- The Delta population is about 6.7 million or 36% of the South Mietnam total of 18.7 million. We have multiplied the total bunne by 36% to estimate the number of burcs and to the Delta. G

### Table 3.15-B

### Jetimated Dissel Fuel Compution of Defines Conventing in Delte-1974

	Engines Operating in Dolto	Hours or Kile- meters of Op- gratics per year	Diesel Fuel Con- sumption per hour <u>er per Km.(litern)</u>	Total Annual Consumption KL
4-wheel tractors ever 30 HP	8,101	1,000 hours	10 liters/hours (b)	81 ,010 (6)
4-wheel tractors under 30 HP	1,135	800 hours	6 liters/hour. (c)	5,448
Marine diesel engines 200 MP and over	3,942	1,000 hours	84 liters/hour (d)	<b>94,608</b>
Diesel engines marine type (Ave 30 HP)	21.,542	500 hours (o)	6 litera/hour	64,626
Sub-total	34,720	•	•	245,692
Privately owned trucks	6,000	39,060 im (c)	15 liters/m.	12,000
Juees	400	30,000 lun (2)	15 litera/hm.	
Total	41,120	-	-	258,492
Converted to Barro	<b>)ls (1 kl =</b> (	6.29 bbls)		1,625,915 1016
Converted to Netri	Le tons (1 M	et. 70a = 7.7 bble	)	211,157 NT

### Notes:

- a) From Table 3.45-A
- b) John Deere 35 NP tractor operating at 50% maximum load consumes 2.5 gallons/hour or 10 liters/hour x 1,000 liters = 10 NT x 8,101 tractors. (32 HP International Harvester Tractor consumes 2 gallons/hour or 8 liters per hour.)
- e) A 23 HP International Harvester consumes 1.5 gallons per hour at 50% load or 6 liters per hour x 800 hours = 4.8 kilo liters.
- d) Assumes consumption of 24 liters per hour of 200 HP engine at 50% full load. (Some examples are: Massey Ferguson 183 HP engine consumes 6.2 gallons or about 24 liters/hour operating at 50% maximum load: 163 HP International Harvester turbocharged engine consumes 6.5 gallons or sbout 25 liters per hour.)
- e) Seasonal or sporadic use for rice mills, irrigation pumping, casava grinding, threshing.
- f. Assumes that trucks and buses travel 30,000 kilometers per year. (in Japan, typical annual distance travelled by trucks is 30,000 to 40,000 kilometers.)

- 104 -

Vietnam imported about 421,779 tens of dissel fuel in 1974 for civilian uses, an estimated 50% of which was consumed in the Bolta, mamply about 211,000 tens. (See Table 3.15-B). The cost of this diesel fuel is about U.S.\$22 million (See Table 3.15-C).

### Table 3.15-C

### DIESEL FUEL IMPORTS -- 1973-75

ĥ

	Berrols Imported	Converted to Netric Tons (7.7 barrels = 1 MT of diesel eil)	Value of Imports (VA& 000*s)
1973	3,753,000 661.	487,402 NE	\$30, 308
1974 (JanSept. actual)	2,430,000 661.	315, 984 HT	<b>\$)), 58</b> 2
1976 (JanDoc. ostimated)	3,240,0 <b>00</b> 661.	420,779 ME	844,777 (8106/WZ)
1975 (Setimated)	3,300,000 661.	438,571 NE	<b>845,</b> 606

The estimates of engineers vary, but from one third to one half of the fuel is being vosted because of improper maintenance. The Vietnam government, as a means of helping egriculturs, fisheries and transportation are subsidizing the users of diesel fuel by VN\$30 per liter; when this is taken into account, annual waste is conservatively estimated at US\$10.2 million, according to our calculations based on two interviews with Esso Eastern, the Vietnam subsidiary of Esson. (See Table 3.15-D)

### Table 3.15-0

### ESTENATED COST AND WASTE OF DEBOGL FUEL IN THE DELTA -- 1974

		U.S.S (millions)	Converted to VNS (millions SNS-VNS670)
۸.	Cost of 211,157 NT of Diesel Puel used in Belta annually (\$106/WT)	22.4	14,996
B.	Waste of Diesel Fuel in Delta annually 330% x A.	6.7	4,902
C.	Waste of Government Subsidy of VN\$30/lite (VN\$30 x 258,492,000 litere x 30% waste)	pr 3.5	2, 396
8.	Total Waste (B + C)	US\$10,2	viiji, 838

Added to this less, are similar lesses arising from the use of gasoline engines, the lower performance of all engines, and the shorter engine life.

It could be argued that Vietnamese governmental investment in repair facilities could be justified as a meane of roducing the national drain of foreign exchange.

The repair facilities need to be equipped with a variety of measuring instruments (for f.i.e. and nessles), jigs and tools and machine tools. The latter would pose several difficulties. Some machine tools are necessary, but the utilization would be inevitably low.

### 3.15.2 Amricultural Implements

We slow recommend that a small-scale foundry and forge be satablished adjacent to the engine repair facilities to help assure fuller utilization of the machine tools and to manufacture tools and pumps.

There is no foundry in Canthe, and a foundry is basic for all kinds of mechanized processing and plays an important role in the production and maintenance of pumps, and various other agricultural implements and fimhing vessels. Facilities for welding would be needed. We recommend unsophisticated facilities as the first step as described in Chapter 4. This is because a major facility such as the CARIC foundry and forge (scross the river from the Majestic Hetel in Saigen) would not be needed in Cantho for some years.

### 3.16 Forest Products

Timber in the Delts consists of two principal species described in Section 3.14 abovs, namely, the tram and mangreve.

The tram could grow to large eise if not cut prematurely. However, the stands are uneven and the trees are usually small. According to Source 54-A, tram typically have malformed trunks, of rapid taper, and of small value as morchantable timber.

The mangrove likewise is not suitable as saw mill timber; saw mills in Cantho cut light logs that are towed down from the north by tug beats. The logs are typically 60 cm. to 120 cm. in diameter. The logs are sawn into planks used to make boats and furniture for the local market. Since mangrove does not fleat, it cannot be towsd.

During our field survey, we visited what was described (by the Industrial Development Bank of Vietnam representatives) as the largest saw mill in Cantho. The mill had  $1,200 \text{ M}^2$  of floor space and was equipped with a 150 HP band saw. The manager said that they did not out tram or mangrove because the wood was not sufficiently uniform or was too small in diameter. Instead, the types of timber processed ware these listed in Table 3.16 below:

### Toble 3.16

TYPES OF TIMBER USED IN CANTHO -- OCTOBER 1974

Vietnemese 	Scientific Nome b/	Application &	Benarius &
Cam Lai	Balborgia barionsis	Boats/furniture	"Rose wood," the most expensive, US\$219/M3
Den Den 🗹	Xylepia pierreii	Boats	Best seller
Giai Ngwa 외	Svietonia	Boats	
GoN Nat	Sindera cochinchinensi	s Beats	
Sae	Hopen Dealbata	Boats/furniture	Dest seller
Thao Lao	Lagorstromomia divors	Boats	Best seller
Bay &	Norus indica	Boats	11 <b>11</b>

A/ Pacific Projects field survey, October 1974

b/ Source No. 53 in bibliography

s/ Losst empensive

The most scenemical location for modern saw mills would be close to the source of raw material. A modern saw mill at Tai Do would probably operate at a competitive disadvantage given the resumption of mormal commerce with the timber growing areas to the north of Saigon.

### 3.17 Gunny Bag Manufacture

In the event that a rice center is to be built at Tai Do Chaptar 4), an assured supply of bags will be required. The approximately 1.0 million tons of paddy shipped out of the Delta annually requires about 10 million bage. Additional bags are required for paddy milled for local concumption in the Delta.

The life of a kenaf or jute bag was reported to us as being 3 to 5 years. Therefore some 2.0 to 3.0 million new bags are needed annually for the Delta's surplus alone. For paddy milled for local consumption probably some 10 to 18 million additional bags are needed. Total annual requirement for paddy and rice alone ranges from 12 to 21 million yen bags per year, depending on the years in service.

The bag industry is facing the following uncertainties at present:

(a) In July, 1974 a new regulation was approved by the Government to reduce the capacity of bags from the former standard of 100 kg. down to 50 kg. It was argued that lifting and carrying of a 100 kg. bag of paddy was injurious to the health of porters. Some of the agencies are opposing the 50 kg. bag and the ultimate outcome is not clear.

(b) The plastic bags sold ex factory in October 1974 for 150 to 160 plasters which is cheaper per kg. of capacity than the 100 kg. kenaf or jute bags. The kenaf bag in October 1974 sold ex factory for 460 plasters in large volumes to the National Food Administration (Bibliography No. 34). New kenaf bage in smaller volumes sold for as much as 500 plasters in October 1974.

(c) A further uncertainty is the future trend in bags used for imported commodity shipmente. If fertilizer or rice were shipped or bagged locally in plastic bags, then these bags will further penetrate the market and replace many of the jute bags now in use.

(d) The supply of kenaf or jute is uncertain. At present, the raw material is partly obtained locally (Central Lowlands and Cantral Highlands) and partly imported from Thailand. The continued availability of the local supply is now in doubt because imports were chasper.

(e) There is substantial bag making capacity available in Vietnam as follows (the manufacturers cannot switch from jute to plastic or vice versa since the weaving machines are completely different):

### Toble 3.17

BAG MANUFACTURING CAPACITY IN VIETNAM -- OCT. 1974

Name of Company	Product Line	Naximum Capacity po 300-day Operating (equivalent in 100 bags.)	er Year ky.
SOVIJUTE	100 kg. Kenaf er jute bags	4.5 (4.5) millio	n ba <b>gu</b>
DOFITEX	Same	1.6 (1.6) "	••
Trieu Quang	50 kg. polypropylone bags	2.6 (1.3) "	Ħ
Vietnam Ky Nghe Mhua Dee	Same	1.8 (0.9) "	19
Dong Thank	Same	2.6 (1.3) "	19
Thong Dung	Same	2.6 (1.3) "	
Uniplastic	Same	3.0 (1.5) "	
Others	100 kg. Kenaf er jute	1.4 (0.7) "	Ħ

Total

20.1(13.1) million bags

. . . . . .

. .

The kenuf bag manufacturers told us that 1973 and to a lesser extent 1974 were unfavorable business years for them because USAID began to use plastic bags on a larger scale than before. Therefore much of the capacity in Table 3.17 above lay idle.

We believe that until there is a clarification of future uncertainties as to (a) plustic vs. kenaf, (b) the supply of domestic kenaf, and (c) the partially idle bag making capacity, that further study of bag munufacturing facility at Tai Do should be deferred.

### 3.18 Fertiliser and Limestone

### 3.18.1 Pertiliser

In the 1973-74 crop year, 2,039,400 hectares of paddy fields were cultivated (Bibliography, 19) in the 16 Delta provinces; this was 72% of the totsl paddy hectarage in Vietnam.

----

In 1973, Vietnam imported 407,000 tons of fertilizers, of which we estimate some 293,000 tone were used in the Delta, i.e. 72% of the total. This means that only 144 kg. per hectare was used in the Dulta. In contrast to this 144 kg. per hectare, the optimum application would have been at least 300 kg. per hectare and in the case of multiple cropping of TN (i.e. IRRI varieties), the application should have been 500 to 600 kg. (The figure of 144 kg. per hectare in close to the USAID/Cantho estimates of 150 kg. per hectare.) These figures are restated in tabular form below:

Therefore, the Delta paddy land is using about one fourth to one half of the fertilizer that is actually needed.

Despite the natural nutrients available in the Mekong irrigation water, paddy production needs fertilizer for economical farming. Using the effect of uses as an example, the farmer obtaine a yield 76% greater with uses than without uses in the dry season, and 39% greater in the wet season, as tabulated below:

	Paddy Yield <u>Without urea</u>	Paddy Yield <u>With urea</u>	Added Yield <u>from urea</u>
Dry season yield	3,400 kg.	6,000 ky.	2,600 kg. (+ 76%)
Wet season yield	3,300 kg.	4,600 kg.	1,300 kg. (+ 39%)

In terms of the financial return to the farmer, his expenditure of 24,000 plasters (\$36) to buy 120 kg. of urea for dry season cultivation brings him revenue of 236,200 plasters (\$338). That is, the investment in urea brings a 742% return to the farmer as follows:

- A. Cost of 1 kg. of urea ..... 200 plasters

- (2,600 kg x 87 piasters)..... 226,200 piasters
- E. Farmer's extra cash by using urea (D-C) 202,200 plasters
- F. Farmer's return on investment in urea
  - (E/C-1.00) = 742%

The conclusion is that the Delta's productivity could rise substantially, possibly 30 to 50%, if adequate quantities of fertilizer were applied. Instead of a Delta surplus of some 1.0 million tons, the surplus could be raise to 1.3 to 1.5 million tons by the application of fertilizer supplies. (This does not take account of various other measures to raise yields such as dry season irrigation, more deep plowing by tractors, and wider cultivation of TN varieties.)

Expressed in a different way, 1 kg. of urea brings 22 additional kg. of paddy, as shown below (this assumes the farmer applies an annual total of 180 kg. of urea):

Dry season crop	2,600 kg of extra paddy _ 21.67 kg
•	120 kg of urea applied
Rainy season crop	1,300 kg of extra paddy = 21.67 kg
•	60 kg of urea applied

Vietnam's imports of fertilizer cost \$45 million in the year ending June 1974 (USAID, \$36, GVN \$9 million).Tonnages for prior years are shown in Table 3.18.1 and have risen from 323,000 tons of material in 1963 to 407,000 tons in 1973. However, as indicated, this 407,000 tons is less than half the optimum quantities. Therefore to assure future supply, the manufacture of fertilizers is required in Vietnam. Other countries in Asia have proceeded with the construction of urea plants (Bangladesh, Pakistan, India, Philippines, Taiwan, Korea).

Ideally the Delta paddy farmers need a mixed fertilizer similar to what is being used today. USAID has been financing a mixed fertilizer, called AMOPHOSCO, consisting of 16 parts  $P_2O_5$ , another 16 parts of nitrogen, and 8 parts of potassium.

The intention is to shift to a different mixture: 18 parts of  $P_2O_5$  and 46 parts of nitrogen, and no potassium. In addition to the mixture, USAID will finance urea containing some 40% nitrogen. The mixed fertilizers will probably be imported in bulk and mixed at a mixing **plant**. Preliminary plans contemplate the site as being Vung Tau (SE of Saigon on the coast). We are recommending Tay Do (See Chapter 4).

- 111 -

Table 3. 18.1

PERTILIZER: INFORTS AND ESTIMATED USE, CALMIDAR YEARS\* 1,000 Netric Tome

		1	рот т и			list is to the	j
1			<b>a</b> t <b>s</b>				
L. 4	2	<b>A</b> .	1	Total			
1963	58.4	96.9	10.8	1.901	ŝ	•	·
	6.5	41.3	8.3	82.5	261	•	•
. <del>(</del>	26.4	6.95	1.11	<b>9.5</b>	2	ŀ	ł
<b>8</b>	65.7	5.5	13.2	114.3	5ęJ	193	•
67	2-65	0" 12	<b>č</b> .č	105.4	ŝ	217	•
- 58	r. <del>.</del> .	3.8	2"1	29-2	R	<b>8</b> 7	•
9	17.5	51.6	25.4	214.4	483	ŝ	•
R (	135-3	61.9	24-9	222.3	518	<b>6</b> 2	66 1
7	50-1	15.1	6.5	80.6	6/1	950	<u>z</u>
. 6	8.2 2	4.12	18.0	145.2	315	*	2
2 8	8° 121	8-6£	<b>K. K</b>	195.9	6	<b>R</b>	182
R							
8							
R							

. Through 1967 contracts americal; 1968 to dete-arrivals.

· Includes 0.1 of Mg.

Source: Imports based on reported unloadings. Use based on USAID estimates.

### 3.18.2 Urea Manufacture

To manufacture urea, an ammonia plant would be required, fueled by some form of hydrocarbon. Ideally, if gas in commercial quantities were found off the Vietnam coast, the hydrocarbon could be natural gas pumped by submarine pipe-line to the plant site. Alternatively, naptha could be provided from a petroleum refinery to be built in the future. As an interim step imported naphtha is suitable.

The ammonia plant would produce the two essential ingredients for urea, ammonia and  $\rm CO_{2^{+}}$ 

The economics of the following alternatives for hydrocarbons would have to be studied: (a) potential offshore gas from Vietnam, (b) naphtha produced in a refinery to be built in future in Vietnam, (c) imported naphtha, (d) possibly imported LPG or LNG (the last alternative is probably the least desirable since gasification facilities will be needed).

Naphtha, the principal feedstock for petrochemicals, is now in surplus in Japan, and only half the 17% naphtha fraction is consumed owing to the currently underutilized ethylene capacity. Even after the recovery of Japan's ethylene industry, long range supply arrangements could be made a geographically close supply source such a Singapore where crude is refined.

We believe that it would be logical to site the a urea plant at Cantho, on or near the Tay Do Park, depending on land availabilities when such a plant is built. (Of the 151 hectares in the Tai Do Park, 46.3 hectares are already committed -- see Section 1.2 above). A urea plant might well require 8 to 10 hectares for the ammonia plant and related facilities. (The Vietnam Industrial Development Bank recommends a 425,000 T/year plant -- Bibliography.Source No. 55).

The location in Cantho would seem economical because of the following factors:

(a) The distance from some of the drilling concessions to Cantho is greater than to other sites, but in the case of other concessions, is shorter. In general, the distances are similar (See Figure 3.18.1). If naphtha were imported, it could be piped to Cantho.

(b) Engineering studies made in Cantho by the Hazama Gumi Co., Ltd. (builders of the 33 MW power plant) show that it is possible to construct plant buildings and install machinery at costs that are higher, but not unreasonably higher, than for example, Japanese costs. Piling costs are \$51/meter in Cantho vs. \$32/meter in Japan (Cantho piling costs includes transport of steel piles from Japan to Cantho).

(c) From our talks with plant engineers, buildings in Cantho, if properly piled, could support 20 to 25 tons/M<sup>2</sup>, which is probably less than the weight of such equipment as the ammonia compressors,



### - 114 -

the furnance or urea prilling towers.

(d) Urea would be made near the paddy growing area, rather than 170 kilometers away, if the Vung Tau site were selected.

### 3.18.3 Phosphatic Fertilizers

1

Vietnam needs a phosphatic fertilizer plant because P<sub>2</sub>O<sub>g</sub> as a component in mixed fertilizers is essential to stiffen paddy stalks and thus avoid stalk breakage in typhoons. All phosphatic materials are imported.

A plant to manufacture phosphatic materials could be sited economically near a major port where bulk handling of phosphate rock is possible. By 1976, dredging programs are scheduled to be completed to allow vessels with a 36-foot draft to discharge at Cocan Gio near Vung Tau.

At present, the largest vessels that can discharge near Saigon are the 26,400 gross ton LASH ships (for example the Pacific Far East Lines, <u>Golden Bear</u> with a 32 foot draft). Vessels of such size can now discharge at Thien Lien, northwest of Vung Tau.

However the siting of a phosphate fertilizer plant cannot be decided until several factors are examined such as: (a) the comparative costs of barging the rock to Cantho vs. barging the finished fertilizer, (b) the possibility of having plants in Cantho for mixing and bagging phosphatic, nitrogenous, and potassic materials.

### 3.18.4 Limestone Crushing

Much of the Delta soil is characterized by high acidity, typically in the 4 to 5 pH range for dry soils and sometimes as low as 3, (See Section 2.2 and in Appendix C). In general, the closer the soils are located to the branches of the Mekong, the lower the acidity, thanks to the neutralizing effect of the irrigation water.

There are large areas of the Delta, shown in Figure 3.18.4, that are characterized by especially high acidity; on the official soils map (Bibliography, 79), these areas are labelled, "very acid alluvial soils". We estimate that these areas cover about 1.4 million hectares of which much is planted in paddy. We estimate 700,000 hectares are planted in paddy.

These areas, consisting of 700,000 hectares, represent the most serious acid problem areas. If 300 to 500 kg. of limestone were applied per hectare, the high acid areas would require 210,000 to 350,000 tons of limestone per year.

There is one USAID-assisted limestone crushing plant now under construction located southwest of Thotnot in An Giang Province. The capacity of the plant will be 5 tons per hour or 120 tons per 24 hour day and 36,000 tons per 300-day working year. This plant will be able to meet some 10% to 17% of the needs of the 700,000 hectares of high acid soils (There would also be requirements for limestone where acidity is more moderate.)

Given adequate limestone applications, the Delta's annual paddy yield of 5.1 million tons (See Table 2.3) would rise. Many variables will affect the yield, but assuming a 5% improvement, an additional 255,000 tons could be produced, valued at \$32.4 million (87,000 piasters/685 x 255,000 tons) at domestic prices prevailing in October, 1974. The value of milled rice on world markets would be 2 to 4 times the domestic paddy price, depending on the rice grade, that is an annual range of \$60 to \$120 million (See Table 3.2-A for Thailand prices).

One of the few mineral resources of the Delta is limestone located near Hatien near the coast of the Gulf of Thailand in Kien Giang Province close to the Cambodian border. We were told by USAID that the reserves near Hatien are very substantial, well over 20 million tons. The limestone reserves are large enough to have justified a major investment in a clinker plant near Hatien, a barge fleet to haul clinker to a cement plant (at Thu Duc, near Bien Hoa). The Thu Duc plant produced 263,300 tons in 1971, 243,100 tons in 1972 and 265,200 tons in 1973. Plans are underway to double this cement capacity.

This leads to the conclusion that enough limestone exists to justify a plant to crush limestone at Tay Do. However, we believe that the application of limestone on paddy fields is not widely practiced at present. As in the case of TN rice and the use of urea, several years will be required to build an assured demand. We think that the experience of the Thot Not plant, described earlier in this section, should be watched closely and the willingness of farmers to purchase limestone at current prices should be studied before a decision is reached to proceed with a large scale crushing plant.

Also the installation of some pilot facilities, described in Chapter 4, would be a useful preparatory step to assure wider acceptance of limestone by the farming community.



- 117 -

### 3.19 Low Lift Pumps

As mentioned, much of the Delta's paddy growing area is flooded in the wet season but becomes brick hard in the dry season. During February through April no paddy culture whatever takes place (See Figure 3.2-B). Because of the lack of irrigation, the abundant water of the Mekong and Bassac and the immediately adjacent "desert" presents an unappy contrast in the dry months.

The dry months could become productive given a wider use of low lift pumps. The Report of the PASA Water Management Team (Bibliography, Source No. 23) points out that dry-season irrigation could play a major role in raising annual paddy yields as follows:

Type of project and location	Additional Hectares that Could be used druing the Dry Season
<b>Opening of new areas for double</b> c <b>ropping in Go-Kong, Kien Hoa,</b> and Cai San	2 <b>50,000</b> hectares
Reclamation of new paddy growing areas in An Xuyen and Ton	200,000 hectares
Drainage of brackish water in Thie Tiep-Nhut, Vinh Loi, Quan-Lo, Phung-Hiep	330,000 hectares

Total

780,000 hectares

Assuming a yield of 2.5 to 3.3 tons per hectare, the above new, reclaimed or drained areas could produce 2.0 to 2.6 million additional tons, of paddy raising the Delta's output from the 1973 level of 5.1 million tons to 7.1 to 7.7 million tons, a percentage an increase of 39% to 51%.

The Report of the PASA Water Management Team (Source No. 23) concludes with the following statement,

> "The major need is for an efficient propeller-type pump to lift water from 3 to 10 feet that can be powered by a small gasoline or diesel engine.... We recommend that the development of this low-lift pump be given a very high priority as it is related to immediate increases and diversity in crop production dependent on water supply and drainage."

\* The source is listed in the Bibliography, starting on p. 241.

Since 1970, when the PASA report was written, there has been substantial progress in the local manufacture of pumps in Vietnam. We inspected the following operating pumps (among others) in October, 1974:

Pupping Capacity and Power Source	Costs
1,800 M <sup>3</sup> /hour, powered with a 36MP	US\$1,022 for engine plus
Yenmer Diesel engine (Jam ese)	\$1.606 for mum total \$2.628

54 H<sup>3</sup>/hour power by a 5 HP Kehler gaseline engine (U.S.) About \$250

The engines were imported, but the pumps where machined in Ankiang Province northwest of Canthe. Pumps with other capacities were also available.

The performance of the pumps was good and the technology, to us, was surprising to find in the Delta. The costs of the larger pumps however were high.

A plant to manufacture pumps, taking advantage of existing management and technical skills, should be given high priority in future planning for the Delta (See Chapter 4).

### 3.20 Sonp Manufacture

We are recommonding the establishment of a rice bran oil mill on the Tay Do Industrial Park. (See Chapter 4 later in this report). The recommended annual capacity will yield 408 tons (34 tons per month) of industrial grade oil extracted from rice bran.

This is not a large volume and probably not enough to support a soap plant of economic scale. For example, the NAKYDACO soap making plant in Ha Queo, northwest of Saigon, uses about 1,200 tons per year (200 tons p r month). In addition they use about 330 tons of tallow, if available, to make toilet soaps.

If additional rice bran oil extraction capacity in Tay Do is added later on, enough fatty acid (industrial grade oil) would be available for a soap plant. In order to make toilet soap, tallow would have to be imported or else obtained from the slaughter house near Saigon (see Chapter 3.9). Probably the best cdurse of action on the Tay Do Park would be to start out with laundry soap, utilizing the locally extracted bran oil. As more raw material were available, toilet soap capacity could be added.

Local kapok seed, peanut, coconut or soybean oil could also be used, but the collection of these oil-bearing materials would be far more costly than rice bran oil extracted near the site of a future soap plant on the Tay Do Industrial Park.

### 3.21. Starch

Table 3.21 shows the production and per hectare yields of manioc. Of the Vietnam total of 379,700 tons, in 1973, about two thirds was produced in the Southern Region, and the remaining one third in the Central Highlands and the Central Lowlands. The production in the Delta is small, accounting for only 13 percent of the total production, namely 50,600 tons (See Table 2.3 above.)

Manioc has the following characteristics: (1) Manioc can be grown in poor tropical soils where other crops cannot be cultivated. (2) The starch yield obtainable from manioc is far above the per hectare yields of starch equivalent from other types of starchyielding crops (e.g. corn, sorghum, rice). (3) The starch content of the root is 25%. (4) Normal production per hectare in Vietnam is now 8 tons, but 30 to 40 tons per hectare is attainable if soil conditions and fertilization are ideal.

There are many uses for manioc such as a substitute for rice in rice-deficit areas, an emergency food in the event of crop failures, an animal feed, and the starch from manioc is used as a raw material for food processing industries, baker's yeast, confectionary, soft drinks, brewing, monosodium glutamate (NSG), sizing of textiles, paper coating, adhesives, pharmaceuticals, alcohol., rubber, abrasives, printing ink pigment carrier, ceramics etc.

### Table 3.21

	Area ha,	Production <b>m. tons</b>	Yield <u>M.T/ha</u>
1963	51,570	389,460	7.552
1964	43,000	2 <b>88,</b> 600	6.712
1965	43,020	236,020	5.486
1966	38,960	2 <b>80,</b> 280	7.194
1967	36 <b>, 50</b> 0	261,185	7.175
1968	35,130	260,190	7.406
1969	32,150	233,485	7.262
1970	30 <b>, 380</b>	215,710	7.100
1971	3 <b>5,80</b> 0	2 <b>70,00</b> 0	7.542
1972	32,100	247,300	7.704
1973	47,780	379,700	7•947

VIETNAMESE PRODUCTION AND PER HECTARE YIELDS OF MANIOC -- 1963-1973

Source: Ministry of Agriculture, Source No. 200 See Sibliography, p. 241

- 121 -

Manioc is widely grown in Southeast Asian countries, Thailand being the world's largest exporter. Thailand exported 124,454 tons of starch, 1,109,363 tons of manioc pellets,2,150 tons as manioc chips, and 1,750 tons of manioc starch cake in 1972. These exports were a major contributor to Thailand's foreign exchange earnings. The pellets are exported mainly to the Netherlands, France, and West Germany where they are consumed as animal feed.

Various proposals have been made to create a manioc industry in Vietnam; these are discussed in a Dec. 13, 1973 memorandum by Maurice V. Sorenson (of USAID Washington; contained in Bibliography No. 10, Annex "O"). Sorenson's memo is entitled, "Prospects for Vietnamese Exports of Cassava Products." Sorenson points out that it will be extremely difficult to establish an export industry based on manufactured manioc products. This is because of the substantial advantage now enjoyed by Thailand and Thailand's reputation in a sophisticated market. The Thailand success originated from German and other European efforts to install quality control and standards of product uniformity so as to gain acceptance as a scientific feed for European consumption.

We agree with Sorenson's opinion about near-term exports, but there are many benefits to be gained from the encouragement of manioc production for domestic consumption for feed and sugar substitutes. The Delta is not as well suited to manioc cultivation as other areas of Vietnam since manioc grows best in dry soil. However wherever if can be grown, cultivation does not compete with paddy cultivation but is, rather, complementary because manioc grows in areas not suitable for paddy. The cultivation of manioc in soils not suited for paddy should be encouraged to supply chips and pellets as raw material for the mixed feed industry. Manioc by itself is not a high grade feed because the protein content is low, but is an important additive for mixed feed because its cost of production is low.(See 4.5.3).

At presentistarch and manioc pellets and chips are being produced in Vietnam on an artesan basis. The starch is for human consumption, glucose syrup, chips and pellets for feed.

Raw manioc contains glucoside which is toxic. The toxicity has to be completely removed during the processing and the quality of processed manioc has to be further improved over existing standards. These factors, as well as means for reducing the production cost, will require further study before this industry can be successful in Vietnam.

Where feed is not a priority objective, a modern, large scale starch producing plant would be a highly worthwhile investment for Vietnam. Bibliography Source No. 9, states manioc starch can be used as a starting point for sweeteners in the soft drink and confectionary industries and could save \$6.0 million of imported sugar per year. Starch can replace up to 80% of the molasses now used in making baker's yeast and MSG, and could save an additional nearly \$1.0 million in sugar import costs. (See 3.4 above).

- 122 -

Experience in the United States indicates that starch from corn, used as the basis for sweetners sold for about 44% less than sugar in the first half of 1974. The same order of savings are possible in Vietnam from manioc. Source No. 9 concludes that any country with a reliable local supply of starch such as Vietnam can use starch in substitution for costly sugar-based commercial sweetners.

Starch is also the basis for the moy mauce and monomodium glutamate ("Accent") manufacture.

### 3.22 Other Industry Candidates

### 3.22.1 Essential Oils

Some tropical countries are cultivating plants for the production of essential oils used for perfume and odorizing products. The outlook for essential oil cultivation in the Cantho area is not promising as shown in the tabulation below (see Section 2.2 above for soils discussion).

The main reason for the unsuitability of the Delta is the lack of well-drained and ventilated soils and the high acidity of the soils.

(Table 3.22.1 continues on next page)

### Table 3.22.1

### COMPARISON OF CONDITIONS REQUIRED FOR CULTIVATING ESSENTIAL OIL PLANTS WITH CONDITIONS IN THE MEKONG DELTA

(A) Type of Essentia Oil Plant	(B) al Suitable Soils and Weather	(C) Suitable pH Value Range etc.	(D) Suitability of <u>Delta Conditions</u>
Citronella	non-clayey, regular rainfall	<b>pH.</b> 6 <b>.5-7.</b> 3	Unfavorable because of 4 - 5 pH Soil
Lemon Grass	Can grow in lateritic (red) Soils (Xacat and Don Thuan) or gray soils (Ong Yem) Regular rainfall on high ground	n•a•	Clayey soils of Delta not suitable
Patchoul y	Well drained, loose soils, regular rain- fall	n <b>.e.</b>	Judging by the fact patchouly grows in well-drained, non-acid soils of Sumatra, Malaysia, Reunion on high ground, doubt Delta suitable, Also March-December cultivation precludes well-drained soild needed by patchauly.
Mentha Arvensis	Well drained, Sandy Soils. Regular rain- fall	n₊a	Delta <b>Soils</b> do not match conditions in column B
Cannelle	High ground as on western slopes of mountains in Quang Nam, Quang Ngai Provinces		Believe Delta too lowmand dry season would stunt plants dring 3 year growing cycle.
Eucalyptus	High humms, Well ventilated, Bandy soils	n.a.	Delta Soils are not sandy

Source No. 65 in Bibliography, Appendix F

- 124 -

### 3.22.2 Mushrooms

Mushrooms could become an industry, dried in the near future, but canned later on when Vietnam imports or makes blackplate and has economical canning processes. The mushroom could become an valuable additional cash crop for the farmers, as it has in Taiwan and Japan for example.

In Japan, mushrooms (called "Enokidake") are cultured in urns filled with rice bran and saw dust. Such culture has reached large-scale proportions and 20,000 tons are produced monthly, centered in Nagano Prefecture. (Bibliography Source No. 63).

In Vietnam, mushrooms are already being cultured successfully using chopped rice straw as a medium

### 3.23 Insecticides and Pesticides

In the calendar year 1973, 3,238 metric tons of pesticides and herbicides were imported or approved for import. The value of this 3,238 tons was U.S. \$5,985,448.

In the calendar year 1974, imports of similar magnitude were reported for all types, except that DDT imports declined.

Of the total imports in 1973, 11% were financed with foreign exchange allocated by the Government of Vietnam and 89% were financed by USAID. The top ten products imported in 1973 are listed in Table 3.23 on the next page. Not shown in the table are imports of "Endrin technical" which, along with DDVP (Vapona), is important for the new TN paddy varieties.

No records are kept, according to USAID, of the consumption of pesticides and herbicides by region, but the major use is for paddy cultivation. Since 85% of the paddy is grown in the Delta's 16 provinces (See Table 2.3), we estimate that about two thirds of the tonnage or 2,100 tons are used in the Delta.

We suspect that the application of these materials is substantially below optimum norms. The Industrial Development Bank of Vietnam branch in Cantho states that the ideal application is about 15 kg. per hectare. If this is correct, the Delta's 2.0 million hectares, would have a requirement of 30,000 tons or more than 10 times the tonnages now being applied.

At a later stage, it would be useful to examine the pesticide/ herbicide distribution system, the economics of importing in larger bulk, and packaging on the Tay Do Industrial Park.

The improved application of pesticides would reduce predatory insects,eliminate more weeds and could well increase the Delta paddy yields by 5 percent or more. At today's prices of over \$450 per ton (Table 3.2-A),this would represent substantial foreign exchange earnings for Vietnam (5.0 million tons x 5% = 250,000 tons x \$450/ton= \$112 million).

### Table 3 .23

### VIETNAMESE PESTICIDE AND HERBICIDE IMPORTS -- CY 1973 (Unit: Metric Tons of Active Ingredient) (Source: USAID/Cantho, Letter from Sr. Ag. Advisor, Am Emb/CG-4)

lank	Type	USAID	USAID	GVN Temporte	
		Imports	Licenses	and Licenses	Total
1.	DD1 75%	569	•	-	569
2.	Microsel A/ca Silicate	164	128	-	2 <b>9</b> 2
3.	Diazinon	2 <b>29</b>	52	-	261
4.	DDVP (Vapona)	167	62	•	2 <b>29</b>
5.	2,4-D Herbicide	161	50	-	211
6.	Dibron Technical	70	100	-	170
7.	Nalathion Technical	60	100	-	160
8.	Azodrin 5	108	24	•	132
9.	Sevin Technical	<b>7</b> 6	27	-	103
10.	Texaphine Technical	•	80 	-	80
	(Top 10 ( Subtotal)	1,604)	(623)	•	(2,227)
	Others	463	197	351	1,011
	Total	2,067	8:20	351	3 <b>, 238</b>

- 126 -

and a second second

### 3.24 Seed Center

In Chapter 4, we are suggesting the establishment of a future paddy processing center on the Tay Do Park. Central to this proposal, is a modern mill for maximum recovery of by-product bran and the production of premium grade rice for export. Hopefully this would be only the first of such facilities aimed at gradually upgrading the Delta's exportable rice surplus to world market standards.

The success of milling operations using the rubber roller system depends in large part on the continuous, uninterrupted processing of uniform paddy.

Frequent shut downs to adjust the rollers has been one of the main problems encountered in the period of mill start-up operations in developing countries. Such interruptions frequently stem from the fact that precise adjustments of the rollers are needed and any change in the size or shape of the grain can jeopardize the outturn quality and the profitability of the mill.

Since 1968, the new TN varieties have been introduced from the International Rice Research Institute at Los Banos in the Philippines. These varieties are rapidly being accepted (see Figure 3.2-A).

During our field survey, we interviewed a farmer who showed us seed for the TN 5, TN 8 and TN 20. He was growing all three varieties plus the local long grain (Nan Keg) variety.

The following principal varieties of paddy are found in Vietnam:

Paddy Variety	Grain
Local Long (Nang Keg)	Long
TN 5	Medium
TN 8	Medium
<b>TN</b> 22	Medium
Floating Red	n.a.
TN 20	Medium
Aromatic (Nang Thom)	Long, slender

(Bibliography Source No. 40)

To assure the success of future milling operations, agreement should be reached on the paddy varieties that certain farmers will grow; contracts should be executed to assure uniform raw material for the mill; and a rigid inspection system adopted at the mill · to help assure continuous out-turn of a quality product.

During our field survey, we found that the seed dealers and farmers were not always sure about the identity of seeds. Dealers were reported to have sold "sorghum" seeds, but the buyers learned the seeds were something else. Confusion over the new rice varieties will linger for years, especially in the more remote areas of the Delta.

To help assure the delivery of uniform paddy to the new mill, we are suggesting the establishment of a certified seed center, stressing the wholesale distribution of certified varieties of seeds (See Chapter 4, p. 171)

### CHAPTER 4

### POTENTIALLY FEASIBLE INDUSTRIES

### (PRE-FEASIBILITY STUDIES)

### 4.1 Rice Milling

Almost all the rice milling in Vietnam is based on the abrasive, out-dated type of milling resulting in a high percentage of brokens, 4 to 11%, (See Table 4.1-A)

Table 4.1-A shows typical mill outturn found by the FAO team in the autumn of 1974 (\*). Table 4.1-B shows the recovery goals of the Mekong Rice Mill in Saigon, as reported to our own field survey group. We think that the results shown in Table 4.1-B are goals only and are rarely achieved.

Rice milling at the village level is still predominantly based on the old "Kiskisan" or Engelberg type screw-type huller. Even the "hand pounding" method is still used as a milling method in rural areas.

In order to achieve a reduction in brokens and a better bran yield, and to produce rice of export grade, a modern facility is needed based on the rubber roller technique i.e. rubber or plastic rollers rotating a differing speeds to remove husk and bran by pressure rather than abrasion.

Rice mills should have a minimum daily capacity of at least 500 tons per day in order to yield 40 tons of bran per day (i.e. 8% of the paddy intake; see Section 4.3 below).

For village level operations, off the Tay Do Park, an improved disk or centrifugal type mill is desirable and represents an intermediate technology between the present wasteful methods and the moderately large scale milling, we are suggesting for the Tay Do Park. (See Appendix

### 4.1.1 Adequacy of Raw Materials and Potential Surplus in Delta

The demand-supply balance of paddy in Vietnam is not clear- cut, but we have made some estimates. The population of Vietnam was 19.9 million in 1973 and paddy production was 7,025,100 tons for the same year (Source 19). Assuming a milling rate of 62 percent (62 percent of the paddy by weight), the production of milled rice was 4,355,560. Dividing this figure by population, suggests an annual per capita consumption of milled rice of 219 kg. which is too high judging by other evidence. Other evidence suggests that annual consumption per capita is about 150 kg in the Delta; see Table 4.1-E for comparisons with other countries.

(\*) From Source No. 34 in Wibliography

- 129 -

### Table 4.1-A

LARGE MILL RICE MILLING TESTS CONDUCTED IN VIETNAM BY FAO RICE STUDY TEAM -- OCTOBER 1974

	Long local variety "good" paddy	TN 8 and TN 5 "ordinary" paddy	Local ordinary paddy. Medium to short
Grade of Rice expected to be produced	Long grain 1 - 15	Nedium 1 = 25	Medium 1 = 25
	×I	×I	×I
Finished milled rice	52.3	57.7	61.5
Proteeus No. 1-2	6*6	6. <b>4</b>	3.7
Brakens No. 3-4	1.3	1.3	0.7
(Broken subtotal)	(11.2)	(2-2)	(4.4)
Total rice	63.5	65.4	65.9
Fine bran (contains husk	7.3	5 <b>.8</b>	7.5
Coarse bran (contains more husk)	5.0	6.2	4.4
<b>Tot</b> al recovery rice + bran	75 <b>.8</b> %	77-4%	77.8%

## Preliminary FAO Report, Source 34

### Table 4.1-B

OUTPUT GOALS OF MERCING RICE HILL, SAIGON, SEPT. 1974

Product	Kg.	% of Total
Finished head rice	100	62 <b>.5%</b>
Brokens No. 1-2	7	4.4%
Brokens No. 3-4	3	1.9%
(Rice subtotal)	(110)	(68.8%)
Fine bran	13	8.1%
Course bran	7	4.4%
(Rice + bran subtotal)	(130)	(81.3%)
Hu øk	25	15.6%
Immature grains	5	3.1%
TOTAL	160	100.0%

- 131 -

We believe that if the process of threshing, drying, storage, milling and the utilization of bran are effectively managed, including the Village level milling, yields, as shown in Table 4.1-C, will be attainable. According to these yields, we have calculated the demand-supply situation for rice of the Delta and of Vietnam as a whole in the form of milled rice as shown in 4.1-D, namely a surplus of 1,375,560 tons annually.

In preparing Table 4.1-D, we assumed that the annual per capita consumption of milled rice in Vietnam as a whole im 150 kg. This figure does not include broken rice used for rice cake. Since Vietnam is producing large volumes of rice, and rice constitutes a major proportion of the total food intake, Table 4.1-D indicates a per capita consumption considerably higher than that of other rice consuming countries in Asia shown in Table 4.1-E.

Based on these assumptions, Table 4.1-D shows that there is a surplus of 1,370,560 tons of surplus rice in Vietnam, but in actual fact, the demand and supply appeared to be out of balance in 1973 and 304,000 tons were imported. (See Table 3.2-B).

The reasons for this "imbalance" include incorrect statistics plus the following loss factors

a. The loss of rice due to the security situation, which has probably been substantial. (figures are unobtainable.)

b. The losses during storage and transportation because of spoilage, insects, rats, and leakage from ripped bags. No exact percentage of loss is known, but estimates of 15%, were often reported to us in Vietnam. If such losses are actually 15%, then they could be 516,000 tons of milled rice annually in the Delta (3.4 million tons x 15%).

c. Excessive quantities of broken rice due to checked grains caused by inferior milling practices and inferior drying technology.

See Table 4.1-A for the results of tests conducted by the FAO rice team in Sept. 1974. More than 10% of the rice was broken in one case, about 7.7% in a second case and 4.44 in the third case.

During the field survey in Vietnam, we observed milling operations in village mills where the percentage of broken rice appeared to exceed 10 percent. The broken rice is sometimes mixed with milled rice to produce a low grade of milled rice for human consumption, but is usually sold for feed or for the manufacture of rice products at lower prices than normal milled rice. This means that the reported yield of milled rice is accordingly less. If the percentage of broken rice is reduced by modern milling methods, to half the existing percentages i.e. to about 3%, this alone would mean a Delta milled rice production increase of 140,000 tons of milled rice.  $(7\% \times 5.6 \text{ million tons of paddy minus 3% \times 5.6 \text{ million tons of}$ paddy x 62% = 138,923 tons.)

- 132 -

### Teble 4.1-C

ASSUMED YIELDS OF RICE AND PADDY BY-PHODUCTS IN DELTA (Assuming Efficient Processing)

	Gentity		Masis for Calculation
Paddy	100	kg.	•-
Brown rice	72	kg.	<b>Pnddy</b> x 0.72
Nilled rice	62	kg.	Paddy x 0.62
Broken rice	3	ing.	Brown rice x 0.04
Pine bran	6.5	hę.	Brown rice x 0.09
Course bran	3	kg.	Fine bran x 0.94
Bran oil	0.78	kg.	Fine bran x 0.12
Defatted bran	5.7	kg.	Fine bran - Bran oil

Senrce: S. Tauji, member of Cantho mervey team .

### Table hel-

# CONSTITUTION OF MILLED RICE FOR CAPITA.150 ha/year

	لو	Ĩ	)61,000 tc/	111,000 t	250,000 t	
	illion 2,000 t (publy)	Braten rice <sup>b</sup> /	166,500 t	51,500 t	115,000 t	
Fita (197	mlation: 7-1 - Muction: 5,55	Milled ríce	3,442,240 t	1,065,000 t	1,006,6 <b>8</b> 0 t	1,370,560 t
	ĒĚ		<b>Cont</b> ities	Conservation in Delta	Consumption out of Delta	Export
(1973)	19 <b>-9 million</b> 7,025,100 t (pa <b>dby</b> )	4, 355, <b>3</b> 60 t		1, 370, 560 t		<b>.</b>
Viet	Predict in:	Milled rice		Surplus		

- A The figure of 5,552,000 tens includes 410,800 tens from Long An Province By deducting Long An, the total is 5,141,200 as shown in Table 2.3
- b/ Assumes improved rate of 3% brokens only (5,552,000/ 166,500) as compared with about 7% under current milling technologies.
- $\frac{c}{c}$  6.5% fine bran yield under current processing cunditions; see Table 4.1-C

1

### Table 4.1-E

### PER CAPITA CONSUMPTION OF MILLED RICE IN FOUR SELECTED SOUTHEAST ASIAN COUNTRIES -- 1972

### (FOR COMPARISON WITH VIETNAM CONSUMPTION)

i -	Population	Paddy (MT)	Brown, Rice (MT)	<u>3/</u> Milled Rice(MT)	Milled Rice Per Capita <u>(kg)</u>
Indonesia	123,110,000	18,585,000	13,009,000	11,708,000	95.1
Philippines	39,000,000	5,437,000	3,759,000	3,383,000	86.7
Tha i Land	40,550,000	13,270,000	9,289,000	8,360,000	206.2
Malaysia	10,440,000	1 <b>,786,00</b> 0	1,250,000	1,125,000	107.0
TOTAL	213,100,000	19,078,000	27,307,000	24,576,000	115.3
					(av <b>erage</b> for f <b>our</b> countries)

1/ FAO Production Yearbook, 1972

2/ Pnddy x 0.70 (loss is largely husk)

3/ Brown rice x 0.96 (loss is largely brokens and bran)

The above calculations mean that there already exists a potential of surplus rice. Even if the losses attributed to the current hostilities were to continue, an improvement in the percentage of brokens would further contribute to a rice surplus.

Table 4.1-D above shows the demand/supply of rice in the Delta, and suggests that there is a theoretical surplus, after the consumption needs of Vietnam are met, of 1,370,000 tons.

If the surplus were in fact to materialize, rice would play an improtant role in the total economy. Furthermore, the bran is widely in demand as feed for chickens, cultured fish, and hogs. The bran would be even more valuable if the oil is extracted first and then the residual oil cake used as an ingredient in mixed feed.

The edible oil is an important material for human consumption the defatted bran is raw material for agro-industries and feed. This will be discussed in greater detail in Section 4.2.

The following factors could increase the surplus even more:

- a. Better irrigation in the dry season (Section 3,19)
- b. Dams to prevent inland flow of brackish water at high tide (Section 3.19)
- c. Better application of pesticides and herbicides (Section 3.23)
- d. More deep plowing by tractors (Section 3.15)
- e. Greater cultivation of the TN varieties (Section 3.2 and 3.24)
- f. More limestone and fertilizer (Section 3.18)

### 4.1.2 Integrated Receiving and Cleaning, Drying, Storage, and Milling

The proposed paddy processing facilities would consist of four elements as described below. The receiving and cleaning unit and the drying unit would operate in phase with the harvests (February and September for TN rice; see Figure 3.2.-B earlier in this report). These units would operate only 60 days per crop or 120 days during the year.

The milling unit would operate continuously 20 hours per day, with down time for maintenance and some hollidays, on an assumed 300-day operating year. The objective of the four units would be to provide an even flow of paddy in such way as to maximize utilization of the milling section. A description of the four units follows:

### 1. Receiving and Cleaning Section

This sections is capable of handling wet paddy at the rate of 50 metric tons por hour.

The above capacity assumes the following conditions:

1.	Quantity of paddy to be processed per crop:	30,000M, tons
2.	Working period per crop :	60 days
3.	Paddy receiving time per day ;	10 hours per day

From the above, the quantity of incoming paddy per day is,

$$\frac{10,000 \text{ tons}}{60 \text{ days}} = 500 \text{ tons /day}$$

As result, maximum receiving and cleaning capacity is

 $\frac{500 \text{ tons } x \text{ 1.2 } (*)}{10 \text{ hours}} = 60 \text{ tons/heur}$ 

(\*) Safety factor

The safety factor is needed to provide more paddy than the minimum capacity of the dryer see below.

### 2. Drying Section

This section is capable of drying wet paddy at the rate of 50 tonsper hour (straight-run capacity) and is capable of reducing the moisture content of wet paddy from a maximum of 18% down to 14% by weight of dry paddy. This requires five passes through the dryers with a certain tempering time between passes. The above capacity assumes the following conditions:

Quantity of incoming Paddy per day : 500 tons minimum
Working hours per day : 24 hours/day

From above, maximum drying capacity per day is:

500 x 1.2 (\*) = 600 tons/day (\*) Safety factor

- 137 -
As a result of the safety factor, the maximum drying capacity per hour is,

600 tons x  $\frac{2 \text{ passes}}{24 \text{ hours}} = 50 \text{ tons per hour}$ 

### 3. Storage Section

This section is capable of storing 15,000 tons of dried paddy having a moisture content of 14% in steel corrugated silos.

### 4. Rice Milling Section

This section is capable of taking in paddy of average quality and conditions and process the intake at the rate of about 15 tons per hour. (14.3 tons dry)

The above capacity assumes the follwing conditions:

	Questity of paddy to be	processed per	c <b>rop</b>	:	30,000 tons
1.2.	Working period per crop	pi decision (	-	:	150 daya/per crop
3.	Working time per day			:	20 nours/per day

From above, the intake capacity of the milling section is:

 $\frac{30,000 \text{ tons}}{150 \text{ days x 20 hours x 0.7 (*)}} = about 15 \text{ tons/hour (14.3 tons dry)}$ 

(+) Safety factor

### 5. Assumptions regarding Raw Materials (Paddy) and Final Products (White Head Rice)

- Incoming paddy is of Fair Average Quality (F.A.Q) of the last crop, having maximum moisture content of 24% w.b. and containing not more than 2% of immature chalky or red kernels, and not more than 5% of cracked grain by weight of paddy.
- Variety of paddy is TN or traditional varities grown in the Delta, Vietnam.
- 3. Specific gravity of the above paddy is assumed to be 0.58 ton/m .
- 4. Ratio of husk against paddy is below 23%.
- 5. The final product is to be export quality, white rice.
- 6. The specific gravity of white rice is assumed to be 0.8 ton/ $m^3$ .

### 6. Assumptions Regarding Operating Conditions

1.	Average mean temperature	:	25 <sup>°</sup> C
2.	Average mean relative humidity	:	70%
3.	Annual rain fall	:	1,500mm
4.	Wind velocities	:	Average 3.0-4.0m/second
5.	Soil bearing pressure	:	l ton per square foot (10.8 tons/M <sup>-</sup> ) at a depth of 2 meters
6.	Estimated required horse power	:	Approx : 850kw. (1,139 Hp)
7•	Fuel for dryer	:	Calorific value 10,000Kcal/kg Density 0,9 kg/liter

### Comments on Above Description and Assemptions

There is already a warehouse for storage near the landing jetty on the Tay Do Industrial Park (See Figure 1.2). This warehouse is 30 x 140 M or 4,200 M<sup>2</sup> and is capable of storing 10,000 tons of paddy Bibliography, Sources 10 and 27. If this warehouse were used, the total costs of the rice center would be reduced. (The Wildman Report," Source 45, recommended 69,000 tons of storage space of Cantho by 1974-75.)

The average mean temperature, humidity and rainfall at Tay Do are higher than the above assumed conditions: Mean temperature is about 26°, mean relative humidity is about 83%, and rainfall is 1,756 mm. Bibliography source 21 for An Xuen Province, and Source 11 for Phong Dinh Province.) The assumptions stated above are "average means," and up and down deviations are contemplated in the plant design specifications. We believe that conditions at Tay Do will be found adequate for the mill.

The soil bearing pressure of 1 ton per square foot or 10.8 tons per square meter can be met with about 93 piles. The steam turbine already installed and operating at the 30 MW power plant in Tay Do weighm 10 tons per  $M^2$  plus comment of 2 tons per  $M^2$ . This is only half the allowable maximum of 20 to 25 tons per M according to engineers who built the Tay Do power plant (Hazama Gumi Co., Ltd. of Tokyo).

The 30,000 KW power plant at Tay Do is currently underutilized, and there is ample power to meet the connected load of 1,139 KW assumed for the rice mill.

### Recapitulations of Annual Processing Capacity

The foregoing capacity data, re-expressed in terms of annual capacities, are as follows:

Receiving and Cleaning: 10 hours per day x 60 tons per hour x 60 days per crop x 2 crops per year = 72,000 tons of wet paddy per year.

Drying: 500 tons per day x safety factor of  $1.2 \times 60$  days of operation per crop x 2 crops = 72,000 tons of wet paddy per year.

Milling: 14.3 tons per hour x 20 hours per day x 300 days per year x 0.7 = 60,060 (60,000/300 days x 20 hours x 0.7).

The products of the mill would be approximately as follows:

Product (% of Paddy Intake)	Tons per 300-day Year	Tons per Day of 20 hours
Milled Rice (62%)	37,200	124
Br <b>ok</b> ens (3%)	1,800	6
Fine bran (6.5%)	3,900	13
Course bran (2.0%)	1,200	9

### Profitability on Dryer, Storage Facilities and Mill

We have estimated that the dryer, storage and milling sections would have a favorable profit outlook as follows (rounded to U.S. dollars, millions):

1.	Invested capital in fixed assets\$4.	6 million
п.	Investment in working capital	8 million
ш.	Total (I + II)\$10.	4 million
τν.	Annual recurring costs	9 million
v.	Annual revenue from sales	6 million
VI.	Annual profit (V - IV)	7 million
VII.	Return on sales (VI/V)	20%
VIII.	Return on fixed asset investment (VI/I)	81%
IX.	Return on total capital investment (VI/111)	36%

The calculations are shown in greater detail in Table 4.7.2below. The estimated profit of \$2.1 million assumes that export grade rice will sold to foreign buyers at \$480 per ton.

The direct foreign exchange capital for imported items or services is \$2.7 million (items A-1 through A=4 plus 85% of item A-18 and 60% of item A-20).

The local currency capital required is \$5.8 million (working) plus \$1.9 million (fixed assets) or a total local currency capital of \$7.7 million.

The ratio of local to foreign capital is about 3 to 1 (\$7.7/2.7 > 2.85).

### Table 4.1.2

### Profitability Estimates on Rice Mill

### Complex Proposed for Tay Do Industrial Park

(Unit: 000's of U.S.\$)

Assumptions:

### (1) The exchange rate from VN\$ to US\$ is VN\$685 = U.S.\$1.00

- (2) First class construction costs are VN\$70,000/M<sup>2</sup>(US\$102)
- (3) Rent will be paid for 10 hectares on the Tay Do Park at US\$0.50/M
- (4) Milled head rice of premium grade will be sold at \$480.
- (5) Power costs are VN\$36.8/KWH, 1974 industrial rate.
- (6) About 93 piles will be required, driven to a depth of 50 meters (as in the case of the 30 MW power plant at Tay Do), and piling will cost US\$49/Meter driven.

### A. Non-recurring Costs (Invested capital in fixed assets..... \$4,601

A-1 Imported machinery for milling section

		- 949
A-2	Imported transformer	17
A=3	Imported machinery for drying and storage Sections	1,441
A-4	Engineering fee for installation	10
N-5	Unloading of equipment and transportation to site from	•••
	Saigon	34
1-6	Transport of piles from Saigon (930M of steel tubes).	20

### Civil engineering work:

A-7	Storage silos (30 x 120 M $\approx$ 3,600 M <sup>2</sup> )	367
<b>A-8</b>	Paddy receiving station (8 x 16 = $128 \text{ M}^2$ )	13
<b>A-9</b>	Rice mill building, 3 stories $(10.5 \times 52 \times 3 = 1.638 \text{ M}^2)$	167
<b>A-10</b>	Husk collection station (7 x 7 M 49 $M^2$ )	- · · /
A-11	Warehouse, $20 \times 40 = 800 M^2$	י ק א
A-12	Transformer installation. 7 x 7 $= 49$ M <sup>2</sup>	15
A-13	Garage, 8 x 16 = $128M^2$	11
A-14	<b>Dormitory</b> and Canteen, $15 \times 7 = 105 \text{ M}^2$	11
A-15	Office, $15 \times 7 = 105 \text{ M}^2$	11
<b>A-16</b>	Truck weighing scales installation, 15 x 7 = 105 M <sup>2</sup> .	11
A-17	Guard house, $2 \times 2 = 4 \text{ M}^2$	1
A-18	Piling and cost of piles, 10 M long steel tubes, 5	•
	per piling welded end-to end, 93 piles 50 M deen.	
	costing \$49 per meter (93 x 50 x \$49)	9.98
A-19	5 trucks, 2 cars (average \$6,429 each, no customs duty)	
	see C. 15 below)	45
A-20	Two foreign engineers in residence 12 months	100
<b>A-</b> 21	Ocean freight	643
	(Subtotal)	181)
·	Contingencies 10% of subtotal	418

### B. Working Capital

<pre>of recurring costs below)</pre>		proce	Should have working capital in first year sufficient to tes 1 crop, i.e. 6 month or 0.5 x \$11,505 (see subtotal	
C. <u>Recurring Annual Costs</u> (1st year only; item C-16 lower in 2nd year)		of re	curring costs below).	<b>7</b> ,72
<ul> <li>Yager /</li></ul>	c.	Recurr	ing Annual Costs (1st year only; item C-16 lower in 2nd	
<ul> <li>C-1 Wages, day shift (54 men x VN\$800/day average x 300 days)</li></ul>			year /	907
<ul> <li>days)</li></ul>		C-1	Wages, day shift (54 men x VN\$800/day average x 300	
<ul> <li>C-2 Wages, evening shift (38 men at VN\$800/day ave. x 300 days)</li></ul>			days)	19
300 days)		<b>C-</b> 2	Wages, evening shift (38 men at VN\$800/day ave. x	
<ul> <li>C-3 Wages, night shift (38 men at VN\$800/day ave. x 300 days)</li></ul>			300 days)	13
<ul> <li>300 days)</li></ul>		C=3	Wages, night shift (38 men at VN\$800/day ave. x	
<ul> <li>C-4 Management and technical staff at plant (6 men x US\$8,000) 48</li> <li>C-5 Paddy procurement staff (5 at US\$8,000/yr.) 40</li> <li>C-6 Electric power costs (1,139 NV connected load x 300 days x 20 hours/day x VN\$36.80 per Kwh = VN\$251.5 mill.) 367</li> <li>C-7 Paddy raw material costs (VN\$109,500/T x 60,000 tons) 9,591</li> <li>C-8 Jute bags (half of 372,000 jute bags containing 100 kg each costing VN\$500 each polypropylene 50 kg. bags would be cheaper.)</li></ul>			300 days)	13
US\$8,000) 48 C=5 Paddy procurement staff (5 at US\$6,000/vn) 40 C=6 Electric power costs (1,139 KW connected load x 300 days x 20 hours/day x VN\$36.80 per Kwh = VN\$251.5 mill.) 367 C=7 Paddy raw material costs (VN\$109,500/T x 60,000 tons) 9,591 C=8 Jute bags (half of 372,000 jute bags containing 100 kg each costing VN\$500 each == polypropylene 50 kg, bags would be cheaper.)		C=/+	Management and technical staff at plant (6 men x	
<ul> <li>C-5 Paddy procurement staff (5 at US\$8,000/yr.)</li></ul>			US\$8,000)	48
<ul> <li>C-6 Electric power costs (1,139 KW connected load x 300 days x 20 hours/day x VN\$36.80 per Kwh = VN\$251.5 mill.) 367</li> <li>C-7 Paddy raw material costs (VN\$109,500/T x 60,000 tons) 9,591</li> <li>C-8 Jute bags (half of 372,000 jute bags containing 100 kg each costing VN\$500 each polypropylene 50 kg, bags would be cheaper.)</li></ul>		C=5	Paddy procurement staff (5 at US\$8,000/yr.)	40
<pre>x 20 hours/day x VN\$36.80 per Kwh = VN\$251.5 mill.) 367 G-7 Paddy raw material costs (VN\$109,500/T x 60,000 tons) 9,591 C-8 Jute bags (half of 372,000 jute bags containing 100 kg each costing VN\$500 each polypropylene 50 kg. bags would be cheaper.)</pre>		C-6	Electric power costs (1,139 KW connected load x 300 days	_
<ul> <li>C-7 Paddy raw material costs (VN\$109,500/T x 60,000 tons) 9,591</li> <li>C-8 Jute bags (half of 372,000 jute bags containing 100 kg each costing VN\$500 each polypropylene 50 kg. bags would be cheaper.)</li></ul>			x 20 hours/day x VN\$36.80 per Kwh = VN\$251.5 mill.)	367
<ul> <li>C-8 Jute bags (half of 372,000 jute bags containing 100 kg each costing VN\$500 each polypropylene 50 kg. bags would be cheaper.)</li></ul>		C-7	Paddy raw material costs (VN\$109,500/T x 60,000 tons)	9,591
<ul> <li>each costing VN\$500 each polypropylene 50 kg. bags would be cheaper.)</li></ul>		<b>c-8</b>	Jute bags (half of 372,000 jute bags containing 100 kg	
<ul> <li>would be cheaper.)</li></ul>			each costing VN\$500 each polypropylene 50 kg. bags	
<ul> <li>C-9 Spare parts, roller replacements, maintenance</li></ul>			would be cheaper.)	186
<ul> <li>C=10 Fuel oil for dryer (19% liters/hour x VN\$69/liter x 2% hours/day x 120 days/year =VN\$38.6 million)</li></ul>		C-9	Spare parts, roller replacements, maintenance	250
<ul> <li>24 hours/day x 120 days/year =VN\$38.6 million)</li></ul>		C=10	Fuel oil for dryer (194 liters/hour x VN\$69/liter x	
<ul> <li>C-11 Interest at 9% on invested capital in fixed assets (9% of Item A)</li></ul>		• • • •	24 hours/day x 120 days/year =VN\$38.6 million)	56
<ul> <li>(9% of Item A)</li></ul>		C-11	Interest at 9% on invested capital in fixed assets	
<ul> <li>C-12 Rent on 10 hectares in Tay Do Park (100,COO MUS\$0.50) 50</li> <li>C-13 Depreciation on plant @ 10% annually (Item A x 20%) 460</li> <li>(Subtotal of annual recurring costs)(\$11,505)</li> <li>C-14 Contingencies and overhead n.e.c. 10% of subtotal of recurring costs</li></ul>		• • •	(9% of Item A)	412
<ul> <li>C-13 Depreciation on plant @ 10% annually (Item A x 20%) 460 (Subtotal of annual recurring costs)(\$11,505)</li> <li>C-14 Contingencies and overhead n.e.c. 10% of subtotal of recurring costs</li></ul>		<b>C-1</b> 2	Rent on 10 hectares in Tay Do Park (100,000 M <sup>-</sup> US\$0.50)	50
<ul> <li>(Subtotal of annual recurring costs)</li></ul>		C-13	Depreciation on plant @ 10% annually (Item A x 20%)	460
<ul> <li>C-14 Contingencies and overhead n.e.c. 10% of subtotal of recurring costs</li></ul>			(Subtotal of annual recurring costs)(\$	11,505)
<ul> <li>recurring costs</li></ul>		<b>C-14</b>	Contingencies and overhead n.e.c. 10% of subtotal of	
<ul> <li>C-15 Taxes on income, land, and customs duties (exempted under Law 4/72, June 1972, Regulating Investment in Vietnam, Chapter III, Articles 3,5,7) none</li> <li>C-16 Interest at 20% on half of recurring cost sub-total 1,150</li> <li>C-17 Cost of sales at 5% of item D below</li></ul>			recurring costs	1,150
<ul> <li>(exempted under Law 4/72, June 1972, Regulating Investment in Vietnam, Chapter III, Articles 3,5,7) none</li> <li>C-16 Interest at 20% on half of recurring cost sub-total 1,150</li> <li>C-17 Cost of sales at 5% of item D below</li></ul>		C-15	Taxes on income, land, and customs duties	
Investment in Vietnam, Chapter III, Articles 3,5,7) none C-16 Interest at 20% on half of recurring cost sub-total 1,150 C-17 Cost of sales at 5% of item D below			(exempted under Law 4/72, June 1972, Regulating	
C-16 Interest at 20% on half of recurring cost sub-total 1,150 C-17 Cost of sales at 5% of item D below			Investment in Vietnam, Chapter III, Articles 3,5,7)	none
C-17 Cost of sales at 5% of item D below		<b>C-1</b> 6	Interest at 20% on half of recurring cost sub-total	1,150
D-1 Sale of 37,200 tons of milled rice (\$480/ton		C-17	Cost of sales at 5% of item D below	931
D. Annual Revenue from Sales		: -1 <b>9</b>	Delit service on A (Starting in 3d year), 30-year loan	170
D-1 Sale of 37,200 tons of milled rice (\$480/ton	D.	Annua	al Revenue from Sales	18.627
19 633		D=1	Sale of 37,200 tons of milled rice (\$480/ton	
transport to Saigon by truck at US\$6/ton		₩+	transport to Saigon by truck at US\$6/ton	17,633
D=2 Sale of fine bran (3,900 tons @VN\$90,000/ton)		<b>D-</b> 2	Sale of fine bran (3,900 tons @VN\$90,000/ton)	512
D=3 Sale of course bran (1,200 tons @VN\$50,000/ton)		n_3	Sale of course bran (1,200 tons @VN\$50,000/ton)	88
D=/t Sale of brokens (1,800 tons @VN\$150,000/ton)		D=4	Sale of brokens (1,800 tons @VN\$150,000/ton)	394

### S. <u>Revenue Compared with Annual Recurring Costs</u>

First Year

	Item D, Annual revenue from sales Item C, Annual recurring costs	\$18,627
	(includes debt service, uspectation, contigencies)	14,907
7.	Annual Profit (D - C)	3,720
G.	Return on Sales (F/C= 3,720/18,627)	20%
н.	Return on Invested Capital in Fixed Assets (F/A=3,720/4,601)	81%
1.	Return on Total Capital (F/A + B = $3,720/4,601 + 5,752$ )	36%

(Note: Profitability improves in second and later years because of reduced interest burden on working capital; see C-16 above)

### 4.2 Related Agricultural Programs

Before the rice mill is erected, careful steps should be taken by the Ministry of Agriculture, the related extension services, and the Agricultural Development Hank to initiate a program of farmer education regarding the cultivation of uniform paddy.

A related step is the establishment of a seed center as mentioned in Sections 3.24 and 4.8.

### 4.3 Milling and Drying outside the Tay Do Park

### 4.3.1 <u>Milling</u>

Mills in rural areas will doubtless serve a local purpose for many years. However, they are usually the Kiskisan or Engelberg type hullers which attain a head rice yield of about 50 to 55%, compared with a proper yield of about 62%.

Inefficient hullers are being replaced gradually by a type of mill representing an intermediate technology, the disk sheller (known in Japan as the "enshin ryoku riyo no shogeki shiki momi suriki"). These are already beginning to be marketed in the Delta through the Mekong Agricultural Machinery Co., No. 17 Le Van Duyet, Angiang Province.

Such mills typically have a capacity of 600 kg. per hour, they require 4 persons to operate and cost about \$5,000 c.i.f. Cantho (including diesel engine, mill, scales, parts, jigs and fixtures). The building should cost about \$3,000, raising the total price to \$8,000. Illustrations of this type of mill and the housing therefor are shown in Appendix F.

In the event that capital is not available for the large, modern mill (\$10.4 million; see Section 4.1.2), a series of these smaller-scale, but high-productivity mills, located within a radius of 10 kilometers of the Tay Do Industrial Park, could theoretically supply enough bran for a bran oil mill with a daily capacity of 40 tons per day. This bran oil mill is discussed in Section 4.4 below.

The calculations for this assumption are as follows: To obtain 40 tons of bran per day, about 104 modern disk shellers would work 200 days per year, milling 620 tons of paddy per day from two crops yielding 3 tons per hectare each or 124,000 tons of paddy per year. One hectare of paddy land could yield 6 tons of paddy from two crops; 124,000 tons of paddy divided by 6 tons would require 20,667 hectares, and **there are more** than enough, i.e. 31,400 hectares of paddy land within a radius of 10 miles of Tay Do.

About 104 rice mills, would be needed to process 124,000 tons of paddy annually i.e. 104 mills x 0.62 tons per hour x 10 hours x 200 days annually = 128,960 tons yearly of milled rice.

The required 104 mills would cost less than a million in U.S. dollars (104 mills x \$8,000 each = \$832,000). However, to be a successful source of bran, the 104 mills would have to be organized efficiently and bran transported to Tay Do within three days of milling to avoid spoilage.

### 4.3.2 Drying

The seriousness of the problem of drying paddy in the Delta can hardly be exaggerated. Paddy harvested in remote areas is spread on bare earth to "dry" in the sun (summatimes paddy is spread out on concrete slabs if the farmers are well off). Paddy collected near roads is spread along the edges of highways where it is frequently crushed by vehicles and re-dampened by squalls. Foreign matter, dirt, cracked grains, and bird losses abound. Some of our respondents in Vietnam guessed that only half the paddy harvested in rural areas ever reaches a mill. Because of double and triple cropping, drying by artificial means in the wet season warrants urgent consideration.

Aside from the large scale mechanical dryer (see Section 4.1.2), small, inexpensive dryers in rural areas could reduce much of the current loss.

### 4.4 Rice Bran Oil

### 4.4.1 Background

The refining of rice bran oil is not widely practiced in Europe or the United States. This is partly because of the comparatively greater availability of other oil-bearing raw materials (soy beans, oil palm, safflower, cotton seed, peanuts, corn, lard.) In Japan, a chronically oil-deficit nation, the rice bran oil industry has prospered, especially since 1950. Today, Japan produces about 70,000 tons of edible oil from rice bran and 30,000 tons of industrial oil (i.e. for soap) from rice bran.

Certain tropical countries are also successfully producing rice bran oil, such as Burma, India and Thailand. These countries share in common a shortage of oil-bearing crops other than rice (Bibliography, 37,41 and 42).

### 4.4.2 Market for Oil

Vietnam is chronically short of edible oils. Imports in recent years have ranged from a low of 5,000 tons in (7/1/73 to 6/30/74)to a high of 26,260 tons (7/1/71 to 6/30/72) valued at \$3.6 to \$5.6 million respectively (See Table 4.4). The deficit in edible oil is not really shown by the import data. Retailing at 1,000 plasters per liter, imported edible oil is out of the reach of most of the Vietnamese population who go without or satisfy themselves with locally crushed peanut oil.

The world market for edible oils is large and rising. In 1972, Vietnam's best trading partners imported edible oils, valued at more than a half billion U.S. dollars; see following tabulation from Appendix B.,

SITC Code	No. Inc	orts	in	US\$,	Millions
421.2	Soybean oil	1	5	9.1	
421.3	Cotton Seed oil		2	7•3	
421.6	Sunflower Seed oil		11	3.1	
421.7	Rape, Colza, Mustard	oil	- 39	9.2	
422.2	Palm oil		24	3.1	
422.3	Coconut oil (copra)		14	1.4	
	Total	••	\$62	3.2	



17 24

I

# VINCENER DECENS OF MATTERS EDITED OIL, CALERONE AND FIGURE TEAR DATA CHREMENIC, 1990-74

## (Builds: Notrie Tous, abbroviated NE, and U.S. Dellars)

	The A - C	alendar Years		7	B - U.S. Fied	al Teers (7/1-	(4/30)
Calender Years	HT Jaported	SATT (Price/MT)	Import Value Unitedoto's	Piscal years	fi laparted	SAN (Price/NT)	Tarent Wales U.S. 2000's
0261	164.41	8 400 (Eat.)	\$5.796 (art.) a)	0461 24	3.4.	1	i
1.461	7.437	\$ 365 (Eat.)	\$2.877 (Eat.) a)	1971	9,600	<b>1</b>	<b>9</b> .58
1972	22,108	8 213 (Est.)	\$4.709 (Eat.) a)	<b>FT</b> 1972	26,260	513 1	5.35
6461	¥.'9	1.134 (Lanta)	1. (I.I.I.I.I.I.I.I.I.I.I.I.I.I.I.I.I.I.I	FT 1973	5.200	8463	8
191	2.490 c)	( <b>F * %</b> )	(Letal)		2,552 c)	0214 <b>8</b>	8£8, 18
	ble A containe the after June Bov. 1974 dee	cutom figures:	ate for the yet publicates	1911 - 19	2,000	CE LA	<b>3</b> <b>3</b>

### The total import value is available in Vietnamene planters from the Directorate of Cantam; houser, the communications rate of the YMB to the USB changed during the particle 1970-72; therefore, us have estimated the Aultar import value as the basis of prices puid for PL 400 imports. 3 Ï

- The away decline of inperts as between 1973 and 1972 is accounted for by the follewing factors: The price of appear wil in 1972 was only \$213 per ton and the USAID thought imports would find a ready market and help generate local currency: in 1973, because the price had almost doubled and the market fell. Almo the Title II PL 45D imports originally intended in 1972 for refugee relief and other charitable purposes were sold in 1975 by the recipients thereby derivating the local demand for edible oil. 3
  - The discrepancy between the January-June 197% data in TABLES A and B may be accounted for by the fact that the colondur year figures are based on the customs clearance date in Vietnam whereas the U-S. fiscal year data for the same period are based on the payment date to the shipper in the Umited States. Ĵ
    - Table A Calendar year data are from <u>Vietaem Statistical Yearbook</u>, 1972 Mational Institute of Statistics; : Starting

Tuble B U.S. Fiscal Year Data, USAID, Saigua.

### 4.4.3 Harbet for Defetted Bree

Bran oil extraction is primarily intended to yield highvalue edible oil, but it is slow necessary to take into consideration the high value of defatted bran as an animal feed.

From bran is utilized for various purposes world-wide, but in Japan, it is usually not used as feed directly, but the defatted bran is used as a feed. The fresh bran becomes rancid in a few days and, the nutrients are no longer svailable.

On the other hand, defatted bran is stable for longer periods of time and is a valuable mixed feed ingredient. Furthermore the protein content of defatted bran is higher than fresh bran and thus is a more valuable feed. Table 4.4.3 shows that defatted bran contains 17.6% protein as compared with 14.8% in fresh bran. (17.6/14.8 =19% more protein on a weight basis.)

### Table 4.4.3

### COMPARISON OF FROM AND DEFATTED BRAN BY AVERAGE FUNCTIONAGES OF COMPONENT MATERIALS (Unit: Porcentages)

	Presh Bran	Defatted Bran	Range of Content Found in Defatted
Notaturo	13.5	11.5	10.4 - 12.8
Crudo Proto in	14.0	17.6	16.6 - 18.2
Cil context	18,2	1.0	1.2 - 2.2
Solusble Non-nitrogens Noteriais	<b>10 35.</b> L	48.5	47.3 - 49.6
Gross Fiber Centent	9.0	9.5	8.) - 10.9
Ach	9.4	11.4	10.3 - 18.1

Source: S. Touji, Homber, Canthe Burvey Team

> Those data were propared by conducting a series of tests in Japan during 1962.

In other countries, defatted bran has gained acceptance because of its stability and acceptance by animal breeders. We believe that the added quality of stability will enhance the acceptability of rice bran in Vietnam. (As mentioned in Section 3.12.2, rice bran is in demand for fish culture and for hog and chicken raising.

### 4.4.4 Problems in Oil Extraction

There are problems in bran oil extraction, but we believe these can be overcome in Vietnam as they have in Thailand or Burma. Bran oil is recovered by the solvent extraction process as in the case of soybean oil. This solvent extraction process requires a large. scale processing plant. The minimum economic size of an extraction plant is 40 tons per day (described in Sect. 4.4.7 and 4.4.8)

The solvent used is hexane, a volatile petroleum fraction, and plant safety standards must be carefully observed.

Bran oil is composed of very small particles so special care is required to separate the micella. Soybean oil, for examples consists of particles which are larger than those in bran oil and the separation of the micella does not require the sophistication characteristic of bran oil processing.

Fresh bran from rice mills contains enzymes which cause deterioration and instability and lowers the quality in as little as 3 days of storage. Therefore the fresh bran from rice mills has to be processed almost immediately. An integrated system is needed ideally consisting of a large-scale rice mill and a nearby bran oil extraction plant.

The oil extraction efficiency is directly related to the oil content of the bran. The existing rice mills in Vietnam are producing course bran containing substantial percentages of husk plus fine bran which is relatively free from husk. An admixture of husk seriously reduces the efficency of extraction. It is therefore desirable to obtain a bran supply from mills which produce relative pure bran, as is currently being done in Japan.

### 4.4.5 Outer Limits on Delta Production of Rice Bran Oil

In most countries, it is unreasonable to assume a 100% recovery of the oil content of rice bran. However, a desirable goal for the Delta would be to extract oil from about half the surplus paddy. Assuming an oil recovery of less than one percent (0.78%) of the paddy by weight, the Delta's 1,371,000 ton paddy surplus (See Sect. 4.1.1) could yield about 5,347 tons of oil (1,371,000 tons x .0078 x 0.5 = 5,347 tons).

We are recommending that production be started on a minimum economic scale at the Tay Do Industrial Park, and then be gradually expanded as more sources of bran materialize (See below, 4.4.7).

We are not implying any reduction of efforts to maximize the production of other oils, coconut, kapok, peanut or soy bean oil. However, rice bran in the Delta offers the most promise for an edible oil industry since the raw materials for other types of oils are widely scattered or limited.

In the event that capital is not available for a bran oil plant described in the following section or in the event that bran or other oil-bearing materials are in short supply, an intermediate technology for extracting oil in smaller volumes could be adopted for a plant at Tay Do. We are referring to the expeller method of extraction which is often used when raw materials are in short supply. This method has the disadvantage of producing a by-product cake containing an oil residual; this becomes rancid in a few days and is therefore not a suitable for feed unless consumed almost immediately after crushing. However, in the long run, oil-bearing by-product cake, just like bran, can be further processed by the solvent extraction method. Therefore an expeller plant, although not an ideal process, could find a market for its cake when a full scale bran oil plant were to go on stream in the future.

### 4.4.6 Raw Materials for Bran Oil Plant

In Section 4.1.2 above, we pointed out that the modern, rubber roller type mill could yield 3,900 tons of fine bran per year or 13 tons per day in a 300-day operating year. This would be only one third of bran requirement for the 40 ton per day bran oil plant.

We are proposing that the required bran supply be collected as follows:

Source of Bran	Tons/Day to be Supplied	Remarks
New mill recommended in Section 4.1.2 of this report	13	In building such a mill, allowance should be made for the 8 - 10 month delivery
·		plus 12 months or more for civil engineering work at Tay Do.
VIMECO	5	See Table 2-2 in Introduction of this report. This mill is already planned for Tay Do.

(Tabulation continued, next page)

Source of Bran	Tons/Day to be Supplied	Bomarks
Mills in Ba Xuen Province or in Cantho (Cai Rang District)	22	For mills in Ba Xuen and Phong Dinh Province, see Section 3.2.Ba Xuen is the largest Delta milling center with 244,000 tons of paddy capacity per year or nearly 16,000 tons of fine bran per year or 53 tons per day. Distance to Tay Do from Ba Xuen is about 70 km on L.T.L. 4.
Total Bran Supply	40	

### 4.4.7 Processing Method

Bran from the mills is sorted to remove immature grains, brokens or other non-cil-bearing foreign matter. After passing through the bexane solvent, the volatile bexane is evaporated off for subsequent liquefaction and reuse. The cake falls to the bottom with crude bran oil floating on surface. The oil cake is then removed for feed. An alkaline process removes about one fourth of the crude oil as industrial oil and the balance becomes edible oil after deodorization and color removal processes and dewaxing. See Figure 4.4.7 below.

Blown polyvinyl bottles would be good containers as they are in Thailand (Industrial Enterprises Co., near Bangkok.)

### Figure 4.4.7

RICE BRAN OIL PROCESSING -- FLON CHART



.

- 152 -

-----

. . .

### 4.4.8 Profitability of Bran Oil Plant

The bran oil plant appears profitable as shown in Table 4.4.8. However a critical factor in the profitability is the cost of raw material. Mills were now selling bran in small lots to farmers or animal husbandry companies at VN\$100/kg. in Oct. 1974. We are assuming that the oil plant can buy in large lots for VN\$90/kg. and sell defatted bran to the feed mill for VN\$105 because of the higher protein content (see Table 4.4.3). In Saigon, fine bran was sold by the Mekong rice mill for VN\$90/kg. (Oct. 1974).

In summary, the calculation in Table 4.4.8 is as follows (in U.S. dollars thousands):

Α.	Investment in fixed assets\$	806
Β.	Investment in working capital	<b>8</b> 62
C.	Annual revenue from sales	65:
D.	Annual recurring costs	380
E.	Annual profit (C-D)	272
F.	Return on sales (E/C)	10%
G.	Return on fixed assets (E/A)	- 34%
H.	Return on total capital (E/A+B)	16%

### Table 4.4.8

### Annual Profitability Estimates on Rice Bran Oil Plant

(Unit: 000's of US\$)

### (for assumptions, see Table 4.1.2 above)

٨.	Non-r	ecurring Cost(Investment in Fixed Assets)	806
	A-1	Plant building, $15N \times 40M \times 2$ Stories = 1,200 M <sup>2</sup> = VN\$ 84 million	123
	<b>A-</b> 2	Warehouse, 20M x 20M = $400M^2 = VN$ \$ 28 million	41
	A-3	Related buildings, 15M x 10M = $150M^2 = VN$ \$ 10.5 million	15
	A-4	Office, 8M - 10M = VN\$	8
	A-5	Machinery, VN\$300 million	438
	<b>A-6</b>	Two Foreign engineers to supervise start-up	100
	<b>A-7</b>	Short-term Training of 2 Vietnamese engineers in Thailand, Burma, or Japan	6
	A-8	Contingency @ 10% of A-1 through A-7	71

- 153 -

в.	Recuri	<u>ing Costs</u>	380
	B-1	2 Managers, @\$8,000	16
	<b>B-</b> 2	21 Skilled workman @VN\$1,400/day x 300 days (3 shifts for some, 2 shifts for others)	13
	B= 3	ll unskilled @VN\$1,000/day x 300 days	5
	Banda	Purchase of fresh bran, 12,000 tons (300 days x 40 tons) @VN\$100/kg., less 10% for volume purchases = VN\$1,080 million 1	, 577
	B-5	Hexane Solvent, alkali etc	35
	<b>B-6</b>	Fuel, electricity, water	79
	B-7	Land rental, 5,000M <sup>2</sup> x US <b>\$0.50</b>	2
	<b>B8</b>	(subtotal of recurring costs)	,727)
	B-9	Depreciation @ 10% of A (less item A=8)	73
	B-10	) Contingencies @ 10% of B=8	173
	B-11	Interest on A @ 9%	73
	B_1*	Interest at 20% on working capital to buy 6 months sup	ply
	0-14	of bran and raw materials and other recurring costs (\$1,727/2 =6 months working capital = \$863 )	173
	<b>B-1</b> )	] Cost of sales 5% x C	131
	8-1	4 Debt service on A, starting 3r year, 30-year loan	<b>3</b> 61
c.	<u>Reve</u>	<u>nue</u>	2,652
	C-1	Edible oil sales, 960 tons x VN\$ 650/kg. = VN\$624 million	910
	C-2	2 Industrial oil sales, 408 tons x VN\$300/kg. = VN\$122 million	179
	<b>C</b> =;	<pre>3 Sale of defatted bran, 10,200 tons x VN\$105/kg = VN\$1,071 million</pre>	1,563
D	. Worl	cing Capital (50% x B-8)	862
E	. Rev	onue Compared with Annual Recurring Costs	
	Re	venue	\$2,652
	An	nual Recurring Costs	\$2,380

.

. . . . . . . . . . . . .

.

7.	Annual Profit (C-B)\$	<b>27</b> 2
6.	Return en sales (272/2,652)	10%
N,	Return en fixed assets (272/806)	34%
I.	Noturn en total capital (272/836 + 862)	16%

### 4.5 Mixed Feed Industry

### 4.5.1 <u>Demand</u>

Imports of ingredients for feed have been as high as \$17 million in recent years (See Section 3.6 earlier in this report). The commercial chicken and hog raising industries and individual farmers raising animals have suffered and some have discontinued or curtailed their operations. The Huynh De Chicken Farm in Cantho reduced their holdings from 20,000 birds in May to 2,500 in October, 1974, and exports of frozen chicken to Singapore and Thailand were stopped. The Ngoc-Thanh hog farm was losing money because the price of feed rose faster than the price of pork in 1974.

There are only two small mixing plants in the Delta at present (My Tho in Dihn Tuong and Long Xuen in Kien Kiang Provinces). These plants depend heavily on imported raw materials.

Mixed feeds are needed in the Delta for hogs, chickens, and fresh water fish. Mixed feed for these animals and fish should be supplied by a medium scale plant, which we are recommending be considered for the Tay Do Park. Ducks can be raised with a minimum of human attention and special feeding, and the traditional practice of feeding them waste vegetable matter or letting them forage in harvested paddy fields minimizes the feed requirements for ducks.

A mixed feed plant is needed to increase productivity in raising pigs and chickens and to make better use of the feed materials available in Vietnam and particularly in the Delta.

If the production of pigs and chicken in the Delta is increased by providing a sufficient supply of feed, hogs and poultry can become the basis for an agro-industry and a major export industry for Vietnam. Appendix B (see, BTN 010500 and 020200) shows that Vietnam's major trading partners imported \$352 million worth of poultry in 1972.

The best policy to stimulate the growth of the mixed feed industry in the Delta would be to make maximum utilization of the resources available in the Delta. At present, the resources for a mixed feed industry are partially in deficit supply, but it will be possible to increase the degree of self sufficiency by starting a medium scale factory and expanding the industry gradually. The potential for obtaining the necessary raw materials is explained below. Vietnam's major trading partners imported feeds valued at nearly \$2.0 billion in 1972 (App. B. see, BTN 230700) and thus feed could find an export market if it were surplus to Vietnams needs.

### 4.5.2 Locally Available Raw Materials

Defatted bran -- If oil were extracted from bran removed from the Delta's surplus paddy, about 89,115 tons of defatted bran would be available for mixed feed (1,371,000 tons surplus x 6.5% = 89,115, See Section 4.1.1). Bran will undoubtedly become the most important ingredient for the mixed feed industry in the Delta.

Defatted bran is, of course, not being produced in Vietnam at the present time. With the modernized milling system (See Chapter 4 above) in the future, the Delta's bran resources can be more fully utilized.

Corn -- About 12,405 tons of corn was produced in 1973 in the Delta (See Table 2.3) much of which can be used for mixed feed. An increase in the production of corn can be expected.

Sorghum -- About 21,000 tons of sorghum was produced in 1972 - 1973 in the Delta (See Section 2.3). When planned increases materialize, sorghum will be one of the most important components for the mixed feed industry.

Manioc -- The Delta produces only a limited volume of manioc, 50,600 tons (Table 2.3) but manioc from other regions of Vietnam can be used and could become an important material for mixed feed. Policies are needed to encourage the cultivation of this product.

<u>Oil Cake</u> -- Oil cake, especially the residual cake after extracting oil from peanuts, kapoc and coconut, is a suitable material for mixed feed. (Peanut oil cake is also important as the raw material for the manufacture of soy sauce.)

Molasses -- This byproduct from sugar refining is another important material for mixed feed, but the existing sugar mills are using a primitive process which does not provide any molasses byproduct (See Section 3.4).

<u>Fish Meal</u> -- This is a critical material for a future mixed feed plant. At present, about 20,000 tons of fish meal are being produced in Vietnam most of which is being produced in the Delta. Most of the output is being exported to Singapore, Hong Kong or other Asian destinations. (See 3.12.5)

<u>Bone Meal</u> -- Bone meal can be produced as a slaughter house byproduct and is important for mixed feed. It is not produced in the Delta, and only small quantities are being produced at VISSAN, the slaughter house near Saigon; see 3.9 above. <u>Peather meal</u> -- This meal can be made from the feathers of chickens and ducks, however the utilization of feathers is difficult at present since feathers have been exported in recent years; the value of exports in 1973 was \$1.3 million; See Appendix B, BTN 050700.

### 4.5.3 Ingredient Mix

The above materials can be mixed in different proportions depending on the end-use of the feed. To make the best use of locally available ingredients, it will be necessary to supplement local supply by imports at least for a transitional period. We are recommending the priority establishment of a mixed feed plant having a production capacity of 50,000 tons annually in the Tay Do Industrial park. According to Bibliography Source 21, imports of feed in recent years were 19,618 tons (1970), 20,117 (1971), and 13,820 (1972) but this has fallen far short of meeting local requirements.

Table 4.5-A shows the general composition of mixed feed is Southeast Asian countries. Table 4.5-B shows a suggested composition for feed that could be produced in a mixed feed plant at the Tay Do Park having an annual capacity of 50,000 tons as mentioned.

### Table 4.5-B

8,700t	40t/day-bran from extraction plant in Tay Do plus other mills	
8,000t	Partly from Delta	
5,000t	From Delta; see Section 2.3 of this repo	ort.
22,000t	Imported from Thailand	
300t	Partially from Delta	
6,000t	Partially from Delta	
	Ten and a	
	Imported	
	8,700t 8,000t 5,000t 22,000t 300t 6,000t	8,700t       40t/day-bran from extraction plant in Tay Do plus other mills         8,000t       Partly from Delta         5,000t       From Delta; see Section 2.3 of this reported from Thailand         22,000t       Imported from Thailand         300t       Partially from Delta         6,000t       Partially from Delta         Imported       Imported

### SUGGESTED FEED MIX FOR TAY DO PLANT

Note: Offal amounting to less than 100 tons (see Section 4.9)

The feed composition shown in Table 4.5-B has 12 to 13 percent crude protein content. This kind of mixed feed alone will be sufficient to raise 130,000 to 140,000 hogs weighing about 100 kg. at maturity. Supplementing the mixed feed would be other products available in the Delta such as roughage consisting of waste vegetable and fruit material plus fish residue from sauce plants. Such materials in conjunction with the mixed feed would be sufficient to raise even more hogs.

When a mixed feed industry is established, there will be increased demand for the several ingredients listed in Table 4.5-B and this will exert upward price pressure on the various ingredients now being consumed in the Delta as a single-element feed. But this could be a factor to stimulate the production of corn, manioc, sorghum, soybeans (in Long Xuen and Chau Doc) and fishmeal (at Rach Gia and Phu Quoc).

With the development of new materials for mixed feed such as sorghum and increases in animal husbandry, a mixed feed industry would appear promising. The production of defatted bran will, of course, play a key role in the development of the industry.

### 4.5.4 Profitability Estimates for Mixed Feed Plant

We are making the assumption that mixed feed would sell for VN\$120 per kg. This price is lower than what was being paid for vitamin-enriched balanced feeds which was VN\$150 per kg in Oct. 1975. (VN\$150/kg was paid by the Ngoc-Thanh Hog Farm in Cantho). However we are assuming that the volume purchase of ingredients will lower costs of raw materials. The assumed VN\$120/kg sales price is wholesale and the remaining VN\$30 is the retailers mark-up.

In summary, profitability is estimated as follows based on Table 4.5.3; units are thousands of U.S. dollars:

Α.	Non-recurring Costs (Investment in Fixed Assets)	. \$568
в.	Working Capital	\$3.581
с.	Sales of Mixed Feed	\$8.934
D.	Recurring Costs	\$7.32
E.	Annual Profit (C-D)	\$1.60
F.	Return on Sales (E/F)	18%
G.	Return on Investment in fixed Assets (E/A)	35%
н.	Return on Total Capital (E/A+B)	39%

(Table 4.5.4 follows on the next page.)

### Table 4.5.4

### Annual Profitability Estimates for Mixed Feed Plant Proposed for Tay Do Industrial Park

(Unit: 000's of U.S. dollars)

(Note: Please see assumptions at beginning of Table 4.1.2 above)

۸.	NON-RECURRENT COSTS (INVESTMENT IN FIXED ASSETS)
A-1	7,000M <sup>2</sup> of Land (See item B-1 for rent)
<b>A-</b> 2	Plant building
	(height is 4M, 1 story)
<b>A-</b> 3	Warehouse, $25x20$ , 3 stories = 1,500 M <sup>2</sup>
A-4	Office, 10 x 20M = $200M^2$
<b>A-</b> 5	Imported machinery
<b>A-6</b>	2 cars
A-7	(Sub total)
<b>A-8</b>	Cantingencies @10% of A-7
B.	RECURRING COSTS
B <b>-1</b>	Rent on 7,000M <sup>2</sup> of Tay Do Park Land
<b>B-</b> 2	2 managers
B-3	Skilled labor (3 shifts), 8 men x VN\$800 x 300 days 3
B-4	Unskilled labor, 9 men x VN\$700 x 300 days
B-5	Cost of raw materials
	(51,000 tons x VN\$95/kg = VN\$4.8 million)
B-6	Fuel, power, water
8-7	Depreciation @10% of A-7
B-8	Interest on A @9%
B-9	(Subtotal) (7,162)
B-10	D. Contingercies @10% of B-8
B-1	I Interest @20% on half of B-8
B <b>-1</b>	1 Debt service on A, starting 3d year loan 21
с.	WORKING CAPITAL
	Six wonths working
	Capital B=0 X 0.5
D.	ANNUAL REVENUE FROM SALES
	51,000 tons x VN\$120/kg
E.	REVENUE COMPARED WITH ANNUAL RECURRING COSTS
	Item D. Revenue
	Item C. Annual Recurring Costs

P.	NNUAL PROFIT (D-O)	
G,	HTURN CN SALME (1,607/8,934)	. 18%
И.	ETURN ON INVESTED CAPETAL IN FIND ADDRES (1,627/368)	· • • • • • •
Ι.	RTURN ON TOTAL CAPETAL (1,507/968 + 3,981)	• 396

### Table 4.5-A

	Moisture	G <b>ro</b> ss <b>Pro</b> tein	Gross Fat	Soluable Non- nitrogerous Material	Remarks
Corn	10-16	8-10	4-5	67-71	
Taken Nice	13-14	8-9	2-3	71-72	
Cattanaed	7-6	21-24	21-23	26-27	
Cottonseed Cake	6-0	41-42	3-6	28-29	Expeller method
Serbeans	10-15	16-30	14-18	23-25	
Soybean Cake	13-14	39-40	8-9	29-30	Expeller method
	9-13	45-47	1-2	29-33	Extraction method
Themats	3-6	30-31	47-48	11-12	
Peanut Cake	8-12	31-46	6-9	23-35	Expeller method
	8-12	47-52	1-2	25 <b>-</b> 27	Extraction method
Coconut Cake	7-14	30-21	6-7	45-48	Expeller method
	8-13	21-22	1-3	48-51	Extraction method
Nice Tan	9-14	13-14	15-18	37-41	
Palm Oll Cake	7-8	18-19	7-8	51-52	Expeller method
	12-13	18-19	1-2	55-56	Extraction method
Sesame Cake	9-10	44-45	12-13	14-15	Expeller method
Kapak Cake	14-15	26-27	7-8	20-21	Expeller method
blasses.	22-26	3-4	0-0.5	63-67	
Tabioca Meal	13-14	1,0-1.9	0,5-1	76-77	
Dried Chrysalis	7-9	50-59	23-30	3-8	
Fish Meal	12-11	41-42	9-10	1-2	
Ser due	10-13	9-11	2.5-3.4	71-73	

### Principle Impredients Used in Mixed Foods in Asian Countrios

Source: S. Tsuji, Member, Cantho Survey Team

### 4.6 Engine Repair, Casting Shop, Agricultural Implement, Pump Manufacture

### 4.6.1 Rationale for Facilities

Section 3.15, in Chapter 3 of this report, pointed out the estimated annual waste in the Delta of diesel fuel of \$10.2 million. The waste occurs because of maintenance neglect and the resulting waste of diesel fuel plus the VN\$30 per liter paid as a subsidy by the Government to support agriculture and fisheries, principal user industries of diesel engines.

Section 3.15 also pointed out that a foundry (i.e. casting shop) was critical to any mechanical processing and a pre-requisite to the growth of industry. Similarly, the local manufacture of tools in the Delta could take place with minimum casting and forging facilities. We think it would be prudent to start out with non-sophisticated and small scale operations. One of the priority products would be pumps(3.19).

### 4.6.2 Three Facilities Proposed

We are suggesting three related facilities to be sited close to each other on a 7,000M plot (less than 1 hectare) on the Tay Do Industrial Park. The three facilities are: (1) Engine Repair Shop, (2) Casting Shop and (3) Machine Shop. A suggested layout of these facilities is shown in Appendix D. A list of the equipment needed is shown in Appendix D-1.

### 4.6.3 Doubtful Profitability in Starting Years

If similar facilities were operating in Japan, the rule of thumb in the engine repair and metal working industries would suggest monthly revenue of over Y10 million (about \$33,000) or annual revenue equivalent to \$396,000. However Japanese experience is not comparable with the Delta since the cost of castings and repairs of engines are high in Japan by world standards. The utilization of the proposed facilities would surely be low in Tay Do at first since the practice of regular engine maintenance is not well entrenched. The number of castings ordered from the foundry on Tay Do would be low initially. Parts inventories for the many makes of tractors (See Section 3.15) would also be uneconomical.

We think that in planning these facilities, an assumption should be made that sales during the first two years would not be high enough to cover costs. Probably by the third year, operations would begin to be profitable.

For the above reasons, we are making the calculations in Table 4.6 for the third year of operation. We are also suggesting that this project not be viewed and financed as a commercial venture because the period of loss operations at the beginning can well justfy government financing in view of foreign exchange losses incurred annually due to fuel waste as mentioned above.

### 4.6.4 Profitability Estimates for Third Year of Operations

In summary, the estimated profitability for the third year of operations would be as follows in thousands of U.S. dollars:

More detail is shown in Table 4.6.4 and Appendixes H-1 through H-4.

### Table 4.6.4

### PROFITABLITY ESTIMATES FOR ENGINE REPAIR, MACHINE SHOP AND CASTING SHOP IN THIRD YEAR OF OPERATIONS

(Note: See assumptions in Table 4.1.2 earlier in this report)

(Units: 000's of U.S. Dollars)

Non-recurring Costs (Investment in Fixed Assets) ...... \$221 ۸. Engine Repair Shop Equipment (See Appendix H) ..... 41 A-1 Machine Shop Equipment (See Appendix H) ..... **A-2** 51 Casting Shop Equipment (See Appendix H) ..... A-3 g Construction costs (40x15, 30x8, 25x6 @\$102/M<sup>2</sup>) ..... 100 A-4 (See Appendix H for layouts) A-5 **A-6** Contingency @10% of A-5 ..... 20 Β. Managers, 3 @\$8,000/year ..... **B-1** 25 **B-2** Clerical, 2 @\$VN1,200/day x 300 days ..... 1 Engineers, 4 @\$6,000/year ..... 24 B- 1 Skilled workmen @VN\$1,400/day, 11 x 300 days ..... **B-4** 7 Unskilled workmen @VN\$1,000/day, 11 ..... B-5 5 B-6 Land Rent on 7,000 $M^2$  @at Tay Do US\$50/m<sup>2</sup> ..... 3 B-7 Fuel, power, water (Power is 51KW x 1000 hours x VN\$36.8/KWH) ..... 7 **B-8** 58 Raw materials, maintenance ..... B-9 B-10 Interest on A @9% ..... 7 B-11 Interest @20% on B-9 for 6 mos. ..... 8 с. \$33,000 x 12 mos. or \$396 annually based on experience of similar facilities in Japan. However, sales in the Delta would be much, lower; see narrative above in Section 4.6) D. Working capital ......\$50 6 mos. x B-9 E. Annual Revenue Compared with Annual Recurring Costs Anmual Profit (C-B) ..... \$ 55 F ..

G.	Neturn on	Sales	(55/200)	27%
н.	<b>Beturn on</b>	Fixed	Assets (55/221)	2 <b>5%</b>
1.	Return on	Total	Capital (55/221 + 50)	20%

### 4.7 Fertilizer and Limestone

### 4.7.1 Fertilizer

As described in Section 3.18 earlier in this report, Vietnam meeds a urea plant, limestone crushing and phosphatic fertilizer plants. The phosphatic fertilizer should be manufactured near ports where bulk discharge facilities for phosphate rock are available. Because of the shallow draft of ports in the Delta (See Figure 2.4), ocean going bulk carriers can not discharge in the Delta.

The urea and limestone plants, however, could be sited in the Delta and in the Tay Do Industrial Park if sufficient land from the Park's 151 hectares remains available when the plant is built.

The Delta's 2,039,400 hectares of paddy land requires a minimum of 300 kg. per hectare of urea annually or a total of 611,820 tons (See 3.18). Based on imported naphtha, a urea and related ammonia and CO<sub>2</sub> plants would cost about US\$250 million judging by the following recent examples.

Country:	Indonesia	B <b>angl</b> adesh
Total cost (US\$ millions)	\$192.0	\$249.4
Foreign Exchange		
(US\$ millions):	\$115.0	\$ 88.0
Local currency		
(US\$ millions):	\$ 77.0	\$161.4
Tons per day of urea:	1.900	1,600
Tons per 300-day year	-	,
of urea:	570,000	480,000
Foreign Exchange		·
Financing Agency:	IBRD	IDA
Est. Completion date:	1977	1978
Date of announcement:	2/27/75	2/18/75
Remarks :		Financing is provisional and depends on six

Even using imported naphtha, the plant should be profitable judging by the following estimates taken from Table 4.7.1. (For costing, we are using the Bangladesh plant as a model):

Non-recurring costs (investment in fixed assets)...\$252 million ٨. Recurring costs.....\$71 B. ... C. Working Capital.....\$20 ... " E. Profit (D-B).....\$41 .. Return on sales (E/D)..... F. 37% Return on value of fixed assets (E/A)..... 16% G. H.

The following table shows estimated costs for a 480,000 TPY plant. We believe by operating additional shifts, annual capacity can be raised to meet the Delta's annual requirement of 611,820 tons of urea.

### Table 4.7.1

### PROFITABILITY ESTIMATES FOR UREA PLANT

.

٨.	Non-r	ecurring Costs (Investment in Fixed Assets)§	251,717		
	<b>A-1</b>	Plant construction (ammonia and $CO_0$ plants, prilling tower, upea plant, bagging plant, naphtha tank farm 12,000 M x \$102	1,224		
	<b>A-</b> 2	Piles (150 piles x 50M x \$49)	367		
	A-3	Equipment (See Appendix I)	250 <b>,000</b>		
	A-4	Training overseas (20 persons x \$5,000)	100		
	<b>A-5</b>	Contingencies @ 10% A-1 plus A-2	26		
в.	Recurring Costs				
	B-1	Managers, 5 x \$8,000/yr	40		
	<b>B-</b> 2	Skilled labor (30 x VN\$1,400/day x 300)	18		
	B-3	Unskilled labor (300 x VN\$ 1,000/day x 300)	131		
	B-4	Imported naphtha 161,000 tons x \$85 (to make 277,500 tons of 82.4% N ammonia)	13,681		
	<b>B-5</b>	Land rental @ U.S. 0.50/M <sup>2</sup> x 100,000 M <sup>2</sup>	50		
	<b>B-</b> 6	50 kg. Bags 9,600,000 x VN\$160/bag divided by VN\$685	2,242		
	<b>B-</b> 7	Cost of Sales 5% x D	5,606		
	<b>B-8</b>	Power VN\$36.80 x 300,000,000 KWH divided by VN\$685	16,117		
	<b>B-9</b>	(Subtotal)	(37 <b>,88</b> 5)		
	<b>B-1</b> 0	Contingecies at 10% of Subtotal B-8	3,788		
	<b>B-11</b>	Interest @ 9% of A	22,654		
	<b>B-1</b> 2	Interest on working capital 20% x C	3,788		
	B-13	Debt service on A, 50 year loan, 10 year moratorium; mame terms as Bangladesh	2,517		

C.	Warking Capital	8,942
	Six months, i.e. 0.5 x B-9	
D.	<u>Sales</u> \$11	2,116
	<b>680,000</b> tons x VN\$200,000 divided by 685 less 20% (Retail price is VN\$200/kg.)	
E.	Sales Revenue Compared with Annual Recurring Costs	
	Item D Revenue\$11	2,116
	Item C Recurring Costs\$ 7	<b>'0,</b> 632
F.	Annual Profit\$4	1,484
G.	<b>Ret</b> urn on Sales (F/D)	37%
H.	Return on Invested Capital in Fixed Assets (F/A)	16%
I.	Return on Total Capital (F/A + C)	1 5%

### 4.7.2 Limestone Crushing

Section 3.18.4 earlier in this report described the AID-financed limestone crushing plant near Thotnot (30 km NW of Cantho). That plant, when completed, will be able to meet an estimated 10 to 17% of the limestone needs of the Delta's high-acid soils.

It would be reasonable therefore to site an additional and larger plant in the Tay Do Industrial park assuming transportation economics are satisfactory for barging limestone from Hatien to Tay Do.

As mentioned in 3.18.4, the acceptance of farmers would have to be tested and the experience of the Thotnot plant carefully noted. However, imported limestone has been sold by the owners of the Thotnot plant and these sales gave rise to the plan for building the existing plant and the owner believes he can thereby replace imports and sell at lower cost. This suggests that a substantial number of Delta farmers are familiar with the benefits of limestone on acid soils.

We are assuming an annual requirement of 225,000 tons (700,000 hectares of high-acid soils and 50,000 of medium acid soils using 300 kg. of limestone per hectare). Assuming a plant were built to operate on 3 shifts, 300 days per year to produce 200,000 tons annually, the capacity would be 28 tons per hour. (200,000 tons divided by 24 days hours x 300 days). We have not been able to obtain reliable data on such a crushing plant but we believe a crushing plant could be built for **about** \$2.0 million. If the limestone were sold at the 1974 price of VN\$90 per kg., total sales would amount to \$263,000 (VN\$90 x 200,000 tons x 1,000 kg/685). Assuming profits of 20% on sales, the profit would be \$53,000 representing about a 5% return on invested capital invested in fixed assets.

In the event that the 200,000 ton mill is not found acceptable, a lower cost and smaller scale alternative could also be considered:

The small-scale mill is a multi-purpose hammer mill, capable of crushing limestone or sorghum, manioc, bone meal etc. The mill would have the following characteristics.

Number of hammers: 16 Hourly crushing capacity: 350 kg. Power requirement: 4 to 5 HP diesel, kerosene or gasoline engine Depreciation: 8 years, but depends on utilization The cost of the mill would be as follows:

Hammer will, including jigs, fixtures, spare parts C & F Cantho\$1,200		
Engine (VN\$800,000) imported 1	,170	
Locally made steel covered table and mounting (VN\$274,000)	400	

Total....\$2,770

We estimate that the cost of grinding limestone would be about the equivalent of \$17.00 per metric ton or about \$0.02 per kilogram. The cost of limestone is about VN\$90 (Oct, 1974) or the equivalent of \$0.13. The ex-factory cost would then be about \$0.15 per kilogram.

### 4.8 Seed Center

To help bring about a standardization and upgrading of paddy and sorghum, a certified seed center is needed (see Section 3.24). According to Wendell P. Knowles (USAID Deputy Director, Cantho) the shortage of sorghum seeds is the main bottle neck in expanding the production of sorghum in the floating rice regions of the Delta. (See Section 2.3). Uniform paddy is a prerequisite to the success of the rice mill described in Section 4.1.2

It is difficult to say how large a seed center would be needed until experience were gained with sales volumes. However it would probably be prudent to start small. Farmers now produce their own seed or obtain seed from other farmers; in a few cases, farmers purchase seed from commercial seed dealers. These practices will continue and in order to make sure that proper varieties are planted, the cooperation of the Agricultural Development Bank would seem desirable. Possibly farmers showing evidence of seed purchases from the center would be given preferential treatment by the Bank or their applications for loans given priority. In future, assuming a large modern mill were built (Section 4.1.2), sales of paddy to the mill would be conditional upon the farmer's ability to deliver uniform paddy of correct grade.

To plant one hectare requires about 20 kg. of seed. To plant enough hectares to assure the delivery of uniform paddy to the mill, would require 4,000 tons of seed (20 kg. x 20,000 hectares yielding 3 tons each = 60,000 tons of paddy, the annual requirement of the large scale mill; see 4.1.2)

The achievement of uniform paddy plantings would probably be phased over several years and farmers would procure from each other in the traditional way. Therefore seed storage capacity for  $h_1000$  tons would not be required. Possibly a seed center to accommodate 1,000 tons only would be a reasonable starting point and this would require a building one tenth the cubic capacity of the warehouse already built on the Tay Do Industrial Park. The calculation for the storage space for 1,000 tons of seed is as a follows:

1 U.S. bushe1 = 1.2445 cubic feet (Wendell Adnerson, FAO Rice Team) 49 bushels = 1 metric ton 1 metric ton can be stored \_ in 61 cubic feet 61 cubic feet/35.3 = 1.73 M<sup>2</sup> of storage per metric ton 1,000 MT/1.73 M<sup>2</sup> = 578 M<sup>2</sup> of required storage space for seeds.

A structure 10M x 22M with a 3 meter calling would probably be sufficient to accommodate the above 578 M of seeds plus room for an office etc.(10M x 22M x 3M = 660 M).

The seed center, would be equippped with a dehumidifying system and sheet metal bins. The capital requirement for such a structure would be on the following order:

٨.	122M <sup>2</sup> floor space <b>@\$</b> 102/M <sup>2</sup> \$1	2,444
в.	Sheet metal bins, arranged with aisles and air	-
	space (4 rows, 19 M long)	8,000
c.	Office equipment	2,000
D.	Dehumidifying equipment	5,000
E.	Contingence 10% of A though D	2,740

Total.....\$30,184

### 4.9 Exported Frozen Food Products Plant

We are suggesting a plant capable of processing at least three products frozen for export. The three products would be cultured river fish (see Section 3.12.2), ducks (see 3.11, last paragraph) or frogs legs.

Later on, as the mixed feed plant goes into operation (see 4.5), the plant could be expanded to handle chicken and ultimately pork but only for export since fresh meats are available in local markets all year round. Chicken and hogs without proper feed, yield a type of meat that would not meet export standards.

We have estimated the profitablility of a plant producing 1,200 tons of frozen fish and duck. (See Table 4.9-B), however, freezing of <u>fish only</u> may prove to be more profitable since the raw material supply could be better assured by creating a captive supply from rafts moored in the Bassac river mear the Tay Do Park.

The freezing of frogs legs is a potentially high-profit business and is being done successfully in Bangladesh ( $viz_0$  the former Fancy Company plant on the airport road near Chittagong). Frog species in Southeast Asia are <u>Tigrana</u> <u>Rana Catasbiana</u>, <u>Rana Escalanta</u>; these or similar species are sold in Vietmamese fish markets, but are not a popular local meat. Frogs could be readily cultured with only moderate feeding.

The freezing of ocean products at Tay Do is probably not feasible because even the freezing plants in Rach fia adjacent to or near the marine fish market have problems maintaining a steady supply of marine raw material,

Shrimp especially, and to some extent fish are easily bruised in transport, and, unless kept alive in water or frozen, will quickly be affected by fungus, ruining their saleability in discriminating over seas markents.

Dr. V. R. Pantalou, Fishery Advisor of the Mekong Secretariat told us that as a converter of protein into flesh, fish are more efficient than most other animals. Some comparisons are as follows (based in part on our field notes):

Hogs: 4 tons of feed yields 1 ton of pork.
Chickens: 3 tons of feed yields 1 ton of meat.
Presh water fish: 2.5 tons of feed yields 1 ton of fish.
Presh water fish: 1.5 tons of rice bran plus 1 ton
of /vegetables and offal yields 1 ton
of fish.

- 173 -
Dr. Pantalou said that he had prepared a study of a project for fresh water fish culture (without freesing) that would require \$588,000 of invested capital and would yield an annual profit of \$214,600 (36% return on fixed asset capital).

The project we are proposing for freen Products in Table 4.9-B is modelled in part on the Kien Giang Co. plant in Rach Gia which freezes marine products and ships by refrigerated van to Saigon for export in containers to Japan. Although the products are not fully comparable, actual sales prices paid to the Rach Gia plant in January-September 1974 are listed below as background for evaluating the validity of the sales and prefit estimates in Table 4.9-B.

### Table 4.9-A

ACTUAL SALES PRICES FOR MARINE PRODUCTS FROZEN AND EXPORTED BY THE KIEN GIANG CO. -- JAN. - SEPT., 1974

Presen Product	Tens Sold	Sales Revenue (USS)	Aevenue per Ten (USS)
Large shrimp	464	\$1,759,000	\$3,791
Small shrimp	71	100,000	1,408
Cuttlefish	12.7	32,000	2,580
Red Snapper	39	115,000	2,949

The fresh water fish may not command as good prices as these shown above; therefore we are estimating the sales price of fresen river fish at \$2,000 per ton or \$2.00 per kg.

In summary, the annual profitability estimates on the freson food products plant are as follows in thousands of US dollars; based on Table 4.9-B:

A.	Non-recurring costs (investment in fixed assets).	.\$415
B.	Recurring costa	• \$1,384
C.	Working Capital	• \$535
D.	Sales	.\$2,100
E.	Profit (D-B)	.\$716
F.	Return on Sales(E/D)	34%
G.	Return on capital in fixed assets(E/A)	73%
H.	Roturn on total capital (E/A + C)	76%

In addition to the freesing of Fiver products and duck for export, there could also be an industrial fish culture project to supply local markets with fresh fish, iced for transport to Saigon.

The 1,200 tens of processed and freses animals would yield a email volume of offal, but less than 100 tens on a dry basis.

- 174 -

### Table 4.9-0

		NUAL PROFITABILITY ESTIMATES ON PROBEN FOOD PROBUCTS (RIVER F168 DUCK, FROGS LBG6.)
A.	Nen-	Nocurring Costs (Investment in fixed Assets)
	A-1	Thirty rafts, 16 <sup>M</sup> x 6 <sup>M</sup> x 2 <sup>M</sup> 05 9,600/raft assuming \$50/H <sup>2</sup> for construction
	A-2	Two 75 HP compressors for freesing plant
	A-3	Other refrigeration equipment, amonia, polystyrens sheets for insulation
	A-4	600M <sup>2</sup> , 1-story construction 06102/N <sup>2</sup>
	A-5	Three Nefrigerated 5-ton (payload basis) vans to carry fremen products to container port costing \$8,000 each (assumes 2 vans making evernight trip to Saigon with van for Standby.)
	<b>A6</b>	Processing equipment (sutters, seaveyers, fork lift, washers thrashing machine, packaging machines)
B.	hee	urring Costs
	<b>B-1</b>	Purchase of fingerlings. (These may be hatched by the plant or purchased.) The calculation for this item is 30 rafts x 192M each divided by 2 because fingerlings are half the size or less of mature fish x 104 kg. per W ef fish x US\$0.05/kg of fingerlings. Actual price in Oct. 1974 was VN\$27/kg for small fish 10 cm or less in length.)
	<b>-</b> 2	Feed for cultured fish (grown in 30 rafts to produce 6008348 tons of fish, Feed pap Faft costs \$11,388. Rafts are 16 x 6 x 2 producing 198 N of fish each; 1N , weight 104 kgs; 192M x 104kg x 30 rafts = 600 tens)
	B)	Purchase of ducks, 600tons (1.5 kg each @VN\$800 each means a requirement of 400,000 ducks x VN\$800 =VN\$320 million = US\$467,153)
	8-4	Packaging materials
	B=5	One Hanager (188,000
	<b>B</b> 6	Two-Sanitation and food processing Technicians 606,000 Technicians 606,000 each

í

- 175 -

	8-7	90 unakilled outlors, packers, dressers, machinery\$39 operators and raft tenders GVM\$1,000/day for 300 days
	-	Land rent on 900N <sup>2</sup> @J\$\$0.05/N <sup>2</sup>
	<b>}</b> 9	Baily trip to Saigon by refrigorated van
	<b></b> 10	Pawer (44,000 INN #VN\$36.80)
	<b>-1</b> 1	Water (24,000M <sup>3</sup> @VW\$40/M <sup>3</sup> )
	Del 2	(Subtet at ) ( \$1,070 )
	<b>B-1</b> )	Centingraties #10# of B-12
	<b>n-1</b> 4	Interest on A at M
	<b>m-15</b>	Interest on C at 20%
	<b>-1</b> 6	Cost of Sales 5.66 x D
c.	<u>teri</u>	ting Copital
	6 m	enths, i.e. 1/2 = 0-12
<b>D</b> .		<b>\$2,100</b>
	<b>-1</b>	600 tans of freesn fish <b>662,000/ten</b>
	<b>b-</b> 8	600 tens of freen duck
R.	<u>Are</u>	<u>rit (p-H)</u>
7.	Rot	urn en Sales (E/P)
G,		urn an Invested Capital in Fided sample (B/A)
M		an Antol Continel (B/A plane C)

### <u>Chapter 5</u>

### PRIORITIES IN ESTABLISHING NEW AGRO-INDUSTRY

### 5.1 Vietnem's Long Range Plans as Related to Recommendations Herein

In 1974, the long range development of agro-industry was described in two basic documents:

"Four-Year National Economic Development Plan, 1972-1975" (See Bibliography, Source 68)

"Five-Year Rural Economic Development Plan (Agriculture, Fisheries, Forestry and Animal Husbandry), 1971-1975" (Source No. 67)

The agro-industries proposed in this report are consistent with both plans although a few of our recommendations are not specifically stated in the plans.

The "Four-Year National Economic Development Plan" (page 255) specifically proposed the following plants for Canthe: manufacturing and repair of agricultural machinery, production of fertilizer, food processing ("such as edible oils"), and animal feeds.

### 5.2 Surmary of Recommendations and Time Scheluling Priorities

Table 1.2(\*) shows the agro-industrial development we are proposing in this report. We suggest that priority be given to the seed center (item A in Table 1.2) and related agricultural extension or information services. The seed center and related activities are critical to the cultivation of uniform puddy for Plant C, the Rice Conter, and to the expansion of the soughum output, needed for Plant F, the Mixed Feed Plant. The seed center itself will cost only \$30,000.

The importance of maintaining the Delta's fast-depreciating tractor fleet, justifies immediate attention to the Engine Repair Casting Shop, Agricultural Implement and Dump Monufacturing Flant (Plant B). This facilitity can be built with a capital expenditure of \$221,000. Deep plowing by tractors of the Delta's clayey soils will be a key factor in the increase of pathy and sorghum output.

### (\*) See Executive Summary, Chapter 1, p. 17.

and the second second

Second priority, in terms of time scheduling, should be given to the Urea Plant (Plant C) and to the Paddy Cleaning, Drying, Storage and Milling Center (Plant D). These will require a lead time of 3 or 4 years because of the need for mobilizing capital and the time required for construction. Meanwhile the steady improvement of seed quality and uniformity and the expanded use of TN varieties can proceed.

Plants C and D are of basic importance if Vietnam is to attain her potential as one of the world's leading suppliers of milled rice. The foreign exchange savings and earnings from these two projects are estimated at \$159 million (See Table 1.2). The rice center, Plant D, could hopefully serve as a pilot project and lead to additional storage, drying and milling capacity for export grade rice. Expanded milling capacity for high grade rice could prepare the way for future exports of edible oils as well.

In the event that the Delta's paddy production is required in the near future to supply the expanded needs of unified Victnam, it would still be worthwhile to export high grade rice and to replace such rice with lower cost brown or lower grade rices. (Indonesia is planning to import lower cost sulfurous crude oils and export high cost sweet crudes).

In terms of timing, we are suggesting the following projects be started after the usea plant and the rice center:

(a) Rice Bran Oil, (b) Mixed Feed, (c) Limestone Crushing on Tay Do Park, (d) Exported Forzen Food Products, (e) Milling and Drying Outside Tay Do Park, (f) Limestone crushing outside Tay Do Park.

Table 1.2 contains further information on the capital requirement, annual mavings or carnings of foreign exchange, production capacity and employment generated. In summary, key conclusions from the recommended program are as follows:

( nerrative centimes, next page )

٨.	Number of plants or complexes recommended for siting on the Tay Do Industrial Park:	8 plants
B.	Number of smaller scale plants recommended for siting outside the Tay Do Industrial Park:	104 rice mills, 10 drying plants, 30 limestone crushing mills
C.	Total capital requirement to finance fixed assets:	\$361 million
ס.	Total direct foreign exchange earnings per year for Vietnam:	\$27.2 million
Ξ.	Total direct foreign exchange savings per year for Vietnam:	\$145.1 million
r.	Total improvement in Vietnam's trade balance ( $D + E$ )	\$172.3 million

### 5-3 Janartent Agro-Industries Decluded

The plants and facilities suggested in Table 1.2 do not include many useful agro-industrial or supporting industries. These have been omitted because we think other locations in Vietnam, outside the Gantho region, would be better suited. For example, insecticide blending, manufacture or packaging is important for the Delta, but we think this can be handled more efficiently in Saigon, close to major shipping facilities, the international airport, and to the head offices of importing companies.

Starch-based industries are also important, but the Delta's minimal production of manioc (see Table 2.3) suggests that other localities to the north would be better suited.

Phosphatic fertilisers are important for paddy, but the absence of deep water ports precludes the discharge in bulk of phophate rock, as mentioned.

Monosodium glutamate (NBG) and soy sauce can be produced in Vietnam, but areas outside the Delta are closer to the manioc supply.

Marine products such as fish meal, frozen shrimp, molluscs, cuttlefish, and snapper can best be processed at Rach Gia or other coastal locations.

ing manufacturing capacity already exist in the Saigen area and can be expanded to provide bags moded in the Tay De park.

### Chapter 6

### EXISTING INDUSTRY AROUND CANTHO

### 6.0 Existing Manufacturing or Processing Plants

Earlier in this report (Section 2.5) an overview of industry in the Cantho vicinity was provided. Of the industries listed earlier, some will have a direct bearing on the viability of the projects we are proposing for Tay Do. The 33 MW plant, already on stream, is needed to meet the connected load of the 8 plants or complexes listed in Table 1.2 at a reasonable cost per KWH.

The 5 ton per day limestone plant at Thot Not can play an important role in heightening the awareness of farmers to the need for limestone. The fishmeal plants in Rach Gia will be important as suppliers of raw material to the Mixed Feed Plant.

Other industries in the Cantho area are listed in Table 6.0. The majority of these plants are of artesanal scale, producing medium quality goods in small volumes. Most of the establishments in Table 6.0 will continue to serve local markets with lower cost products and should neither be contributors to or competitors of the plants we propose for Tay Do.

### 6.1 Integration with Proposed Tay Do Plants

There are some plants in the Cantho region that will have a definite contribution to make to Tay Do as follows:

The larger rice mills included in the 857 total in Table 6.0 below will have to be counted on as suppliers of fresh bran for the Rice Bran oil plant (See 4.4.6 above). Out of the 40 TPD fresh bran requirement, only 18 ton TPD can be supplied by mills on the Tay Do Park. The residual 22 TPD could best be supplied by the larger rice mills in Ba Xuen Province or the Cantho region (Cai Rang especially).

The 40 machine shops listed in Table 6.0 can become customers of the Engine Repair and Casting Shop for more complex repairs that they are not capable of handling (e.g. nossle grinding, f.i.e. adjustments, timing adjustments). In the long run, some of these machine shops or their personnel could be integrated with the Tay Do facility.

The 92 saw mills shown in Table 6.0 will be important during the construction phases of plants on the Tay Do Park. Similarly the brick kilns and tile manufacturers can be suppliers.

- 180 -

The 63 fish sauce plants might become suppliers of fish for the Mixed Feed Plant, but the effect of fermentation on protein in fish residue will have to be confirmed.

The other industries shown in Table 6.0 should have no connection with the Tay De complex, i.e. bakeries, ice making, soy sauce and sugar mills. The sugar mills, if modernized to produce molasses, could make an important contribution to the Mixed Feed Plant.

( Tuble 6.0 fellowagnest page )

<u>Table 6.0</u>

# MARTACTIZING OR PROCESSING ESTANLICHMENTS IN SIX PROVINCES ADIACENT

Ĕ.
ŧ
Ê
6
3
Ë
Ē
3
<b>P</b> .

			The second			Pile		Ice Mak-	Soy-	e	
PROV INCE			Rating Plants		Machine Zons	Fac to-	STICK Kilns	Ing Plants	Plants		
	5	ħ	, 19	R			ñ			R	*
No. YOU - YO	5	<b>C</b>	~	•0	in	~	co	<b>6</b> 0	4	2	\$
HALF - CARGA	<b>%</b>	R	5	5	53	11	•	<del>د</del>	5	~	<b>R</b> 7
800 - 78	F.	đ	•	•0			11	~			Į,
1107 - 1414	R	8	19	2			5	6		15	318
HALL - MALA	i.	ų	n	~	2			12		12	212
	I			I	I	I		I	ł	I	ļ
Truck.	222	<b>*</b> 0	<b>.</b> ;	Ж	¥	۲ ۲		59	6	8	214.1

aure : Martin artise, Cartes

.

### Chapter 7

### POLICIES AND INCENTIVES TO ENCOURAGE AGRO-INDUSTRY

### 7.0 National Goal

Given the world food shortages and Vietnam's need for foreign exchange, development of the Delta's agro-industrial potential could be an important national policy objective.

Given the basic importance of paddy and feed to Vietnam's economy, Government-wide coordination, cutting across traditional jurisdictions, is needed. The successful development of the paddy and feed resources will require a total effort directed toward improving all phases of production and sales outlined below:

### **ELEMENTS OF AN INTEGRATED PROGRAM TO DEVELOP VIETNAM'S PABDY** AND FEED **DESCURCES**

- Better and uniform weeds especially for sorghum and TN varieties.
- Notter and higher volumes of urea and phosphatic fertilizers,
- Nore limstone for acid soils,
- Notter farm equipment, especially threshers and tractors.
- Nore deep plowing by tractors and better tractor maintenance.
- Facilities for the cleaning, drying and storage of paddy.
- Willing to raise the yields of head rice from the typical 57% to 62% of the paddy intake.
- Intter recovery of bran to permit oil extraction.
- Local bran oil production to reduce edible oil imports,
- Notter distribution of bags to reduce leakage and spoilage.
- Reconstruction or new building of irrigation canals.

In terms of financial returns, execution of such programs can be significant. The expert of 1.3 million tons of the Delta's surplus rice could bring \$624 million (1,300,000 tons x \$480/ton) and raise Vietnam's rice expert earnings above the Thailand level of \$488 million of experts in 1974 (See Table 3.2-A  $p_0$  48) Even if domestic consumption needs were to preclude such high volume exports, the international exchange of Vietnam's export quality rice for lower cost grades could yield earnings in the hundreds of millions of dollars.

Vietnam's mills are typically losing about 5 percent of the head rice milled (See Tables 4.1-A and 4.1-C). This loss at wholesale prices (VN\$182,000 per ton in 1974 or US\$266 per ton) is \$58 million, or at international prices (\$480 per ton), would become \$105 million (7,025,100 tons of paddy in 1973 x 5% = losses of 351,258 tons of paddy x 62% = milled rice).

In theory, if all the oil resources in Vietnam's paddy were utilized, the total recovery would be 54,800 tons (1973 paddy production of 7,025,100 x .0078). Of this oil, two thirds would be edible, valued at US\$33.2million (Jan-June 1974 price of \$920 per ton x 36,500 tons) and the industrial oil would be valued at about \$8.0 million (18,300 tons x \$438/ton).

Vietnam's rice bran, even the "fine" bran, contains husk and is therefore not comparable with ideal bran out-turns (See Tables 4.1-A and 4.1-C). However, the continuation of conventional milling almost precludes oil extraction because of the husk contained even in the fine bran from most Vietnamese mills.

A recapitulation of the above figures is as follows:		
Potential value of exportable milled rice	\$624	million
Current losses of rice from poor milling	\$105	**
Value of potential edible oil production	\$33	**
Value of potential industrial oil production	18	11
		د من الله عليات عليمي من الله الله الله الله الله الله الله الل

Total

\$770 million

To put the above figures in perspective, the total imports of South Vietnam were \$676 million in 1972 and \$717 million in 1973. South Vietnam's exparts were \$23 million in 1972 and \$60 million in 1973 (Bibliography, 24).

The conclusion is that focussing chiefly on the production of paddy, bran, and food, Vietnam's economic potential from the Delta is voly favorable, given the administrative mechanisms to assure the best use of resources.

### 7.1 Fereign Investment Policies

Treatment of foreign investment has been highly favorable to the investor and numerous incentives were offered under the terms of Law No. 4/72, June 2, 1972, "Regulating Investment in Vietnam." (Bibliography, 70)

Investors were given a five-year exemption from customs duties on imports, income taxes, property and land taxes, and export taxes, among others. Capital and profits could be repatriated. There were no percontage limits on expatriate personnel.

Special services for investors were provided such as market research and site location services (performed by the Industrial Development Bank of Vietnam, 40 Nguyen Hue, Saigon and the Investment Service Center, 100 Tu-Do, Saigon).

Despite the chronic insecurity of recent years, the policies to attract foreign investment have been reasonably successful. From June 1972 to December 1974, 144 foreign investment projects were approved involving the investment of \$61.9 million in foreign exchange:

Country	Equity	Country	Equity
of	Capital	of	Capital
Investor	(US\$)	Investor	(US)
Liechtenstein	15,000,000	Thailand	621,125
Hong Kong	8,736,461	Israel	337,600
Indenesia	7,500,000	Khmer Republic	281,171
Japan	6,414,771	Panama	250,400
The U.S.A	4,835,051	Australia	216,854
France	4,623,799	Switzerland	155,275
Republic of China .	3,917,558	Sri-Lanka	127,634
Federal Republic of		Laixo <b>nibur</b> g	50,000
Ge <b>rmany</b>	3,664, <b>600</b>	Spain	50,000
Great Britain	2,093,443	Laos	1/1.000
Korea	1,783,030		
Singapore	1, 191, 292	<b>To</b> tal	,864,064

### FOREIGN INVESTMENT

1972 - 1974

Source : Industrial Development Bank of Victual.

### 7.2 Policies regarding Domestic Investment

There are three banks in Vietnam which in 1974 were providing credit for economic development:

Industrial	Development	Bank of	Vietnam	(1 <b>D</b> B)
Agricultura	1 Developmen	nt Ba <b>nk</b>		(ADB)
Investment	and Develop	ment Ban	k (11	DEBANK)

There is an additional institution, NEDEF, which has acted as an intermediary bank, collecting funds from the Government and USAID for dispersal to the above three banks against approved loans.

The above institutions have some very worthwhile achievements to their credit, as summarized below from Source 24, see bibliography, Appendix F.

The ADB was established in 1967 to provide low cost credit to farmers. The ADB has made substantial gains in recent years and by the end of 1973 was estimated to be the source of twenty percent of total rural credit. Expansion of rural credit is one aspect of the government's overall effort to make agriculture the leading edge of economic development, and ADB can be expected to continue playing a dominant role. ADB branches, which are located in every province, are complemented by a Private Rural Banking System, which was inaugurated in 1969. Almost seventy private rural banks (PRB) were in operation as of mid 1974 in almost every province of the country, with a heavy concentration in the Mekong delta. PRBs are privately owned with the ADB holding a minority capital interest, providing expertise and certain banking assistance. PRBs make loans which are required to stress agricultural development and production.

The IDD makes medium and long term credit available to investors for construction of factories as well as equipment purchases. In some situations the IDB will take a temporary equity position in an industrial project. The IDB is also involved in the development of industrial estates in Vietnam, an activity it pursues along with SONADEZI (National Company for Development of Industrial Estates). (The most notable achievement of SONADEZI has been the Bien Hoa Industrial Park where 52 factories have been completed as of mid-1974 producing a variety of products from asbestos reafing to steel reinforcing bars.) The <u>IDEDANK</u> is a private development lending institution which was organized by a consortium of private banks in 1961 with the Credit Commercial du Vietnam currently holding approximately 45 percent of the outstanding shares. IDEBANK, initially named SOFIDIV (Society for Financing Industry in Vietnam), enjeys Government support and encouragement in making medium to long term loans at concessional rates to new and expanding industrial projects. IDEBANK also participates in industrial projects through direct equity participation.

Despite the successes, even during a trying period of insecurity, there are some changes in policies and procedures that may help to accelerate industrial development and give incentives for new industry to site itself outside the urban complex of Saigon-Gia Dinh-Bien Hoa. These suggestions are as follows:

- The approval of industrial projects in 1974 required the concurrence of the Ninistries of Trade and Industry, Agriculture, Interior, Communications and Transportation, Public Works, and Land Reform.
  Sometimes the approval of NEDEF was also required and in the case of foreign investment, the approval of the Office of the Board of Investment an agency within the Ninistry of Trade and Industry. To expedite the approval process, it would seem better to have a single authority where only the Ninistries directly concerned could be represented.
- b. Under Vietnamese law in 1975, the banks could not accept buildings as collateral unless the owner had a fee simple title to the land. This was a disincentive to investors seeking to locate on industrial parks such as Tay Do.
- c. IDB, as a matter of policy, did not attempt to influence entrepreneurs to build plants on the Industrial Parks. This seemed to be a factor in retarding the utilization of the SONADEZI Industrial Parks.
- d. The development banks as a matter of policy have not approved loans for industrial working capital. Working capital, for most of the proposed projects in Chapter 4, is equal in importance to the capital for fixed assets.
- e. Industrial borrowers meeding working capital can only borrow from the commercial banks at annualized rates ranging from 20 to 28%. If industrial development is to become a national goal, the necessary legislation should be enacted to enable the IDB to make loans for working capital and to reduce the interest rates charged by commercial banks.

- f. To encourage future manufacturing in Cantho, certain soning restrictions should be considered. (There was a recent case of a detergent plant whose sponsors wanted to locate in Saigon, but for pollution reasons, the plant site was changed to an area near Danang.) In order to site agro-industry near the sources of raw materials and on the Tay Do Park, zoning should be encouraged.
- 9. In general, the ADB and the IDB regard their spheres of activities as mutually exclusive. However in the case of agro-industry in Cantho, cooperation and joint planning will be essential on the part of the two banks and the Ministries of Agriculture and Trade and Industry as well, This will help assure (a) the availability of correct varieties of paddy (b) help make sure farmers commit themselves to supplying ingredients for the mixed feed plant and (c) help make sure rice mills agree in advance to sell their fresh rice bran to the bran oil plant. In short, the financial leverage of the two banks could operate as an incentive to coordinate the farmers and manufacturers.

- 100 -



### 82.09.22



## $\mathbf{3}$

MICROCOPY RESOLUTION TEST CHART

24 ×

×

### Chapter 8

### EXECUTION OF THE DEVELOPMENT STRATEGY

### 8.0 Angested Sequence of Actions

Preparatory steps to the execution of the agre-industrial development program in Canthe would include the following:

- a. Because of the capital investment involved and the potential benefits to the Vistnamese economy, Government-wide agreement in principle should be achieved on the acceptability of a program such as that outlined in Table 1.2 in the Executive Summary in Chapter 1.
- b. Policies and procedures should be agresd upon whereby the farming community and certain sxisting rice mills will cooperate in providing the raw materials for plants and facilities listed in Tabls1.2, namely: (a) TN varisty paddy and sorghum seeds for the seed center, (b) at least 27 tons per day of fresh bran for the bran oil plant, (c) raw materials for the mixed feed plant, (d) limestone for the crusher, (e) 60,000 tons of uniform paddy for the rice center.
- c. Preparations should be made for meeting the local currency and working capital needs of the new plants (See Chapter 7).
- d. Selection and training of managers for plants to be established (see below in this chapter) should be planned.
- e. Preliminary approaches should be made to lending and investment institutions, forsign governments, and private companies and channels of communications laid for close liaison with such sources of capital and technology (See below in this Chapter).
- f. Preparation of feasibility studies should proceed as seen as possible (See Chapter 9).
- g. Final approaches to financing institutions and private investors should follow the feasibility studies.
- Contracting for plant construction and plant management during the start-up phases should proceed for the (a) Urea Plant, (b) Paddy Cleaning, Storage and Milling Plant, and (c) Rice Bran Oil Plant (We believe that local Vietnamese management will be available for the other plants (See below in this Chapter).

### 8.1 Availability of Local Management for Future Plants at Tay De

Overseas training of Vietnamese and the recruitment of fereign plant managers for the start-up phases will probably be necessary for the plants mentioned under paragraph "h" in Section 8.0 above. However, locally skillsd people can be found for certain of the plants as follows (digits in parentheses refer to Sections of this report):

### Name of Proposed Plant

Engine Repair, Casting Shop, Agricultural Implement and Pump Hamifacture (4.6.2) Per the mi Van Xe Co Quan Cho I is already Per diesel repairs, A Nong Co., a C & I Mi tractors : Hanager of repair pl consortiu establish Herever, T

Paddy, Cleaning Brying, Storage and Milling Plant (4.1.2)

Rice Bran Gil Plant (4.4.7)

Mixed Peed Plant (4.5.3)

Limpstone Crushing Plant

Experted Presen Pool Products Plant (4.9)

### Locally Available Management

For the foundry and forging operations, local skills are available through the CARIC, No. 17 Ben Bach Dang Saigon (French-owned comapny).

For the manufacture of pumps, the Nhuyen Van Xe Co., No. 20 Gia Long Xa My Luong, Quan Cho Noi - Tinh, An Kiang Province is already building 1,800 M<sup>3</sup>/hour pumps.

Per diesel engine repairs and tractor repairs, a company called Trung Tem Heng Co., was established in 1967 under a C & I Ministry decree to repair tractors in Cantho. Mr. N. H. Gia, Hanager of the AMTRACO assembly and repair plant, helped organise a consortium of tractor importers to establish the Trung Tem Nong Co. However, the operations in Canthe were never started because of a shortage of capital. AMTRACO head office is located at No. 44/8 Mgo-duc Ke, Saigen.

torage One of the most progressive mills in Vietnam is the Nghia Thanh Mill in Ba Buyen Province. The mill is equipped with rubber roller hunkers and polishers (Source 40 in Bibliography)

> The Troung-Khac-Can Soap Co. in Saigon fermerly procured rice bran eil from a supplier in Ta Vinh. Because of a shortage of raw material and inferior equipment, the plant shut down in 1968.

There are mixed feed and bagging operations at Long Muhen, Angiang Province and at My The in Dinh Tuong Province.

The 5 TPH, crushing mill near Thet Net in An Kiang Province should be in operation 1975. The combat clinker plant at Ha Tion in Chau Doc Province has had experience in handling limestone along with the Ha Tion Compatible Co. in Bion Hea Industrial Park.

cts There are several frozen marine product plants in Rabh Gia. One of the most progressive is the Kien Giang Co.

### 5.2 <u>Sources of Foreign Exchange Financing</u>

Probably nó single source will be willing to provide financing for a program such as that suggested in Chapter 5. For example the \$249 million urea plant in Bangladesh is being financed by a consortium of lending institutions headed by the International Development Association (IDA), 1818 "H" Street, Washington, D. C.

The following financing institutions could also be approached:

- Asian Development Bank, Romas Boulevard, Manila, Republic of the Philippines.
- 2. Banque de l'Indochine, Paris (alse Asian office 5th Fleer Arasa Bldg. 8735 Paseo de Romas, Makati Risal, Republic of the Philippines)
- J. Caisse Centrale de Coeperation Economique, 233 Blvd, de St. Germain, Paris 7, France
- 4. Credit Lyonnais, 19 Blvd.des Italiens, Paris 2, France
- 5. Prankfurter Bank, 6000 Prankfurt 1, 3821, West Germany
- 6. International Bank for Reconstruction and Development (IBRD), 1818"H"Street, Washington, D. C., U.S.A.
- 7. International Finance Corporation (IFC), 1818 H Street, Washington, D. C. 20433
- Industrialization Fund for Developing Countries, Predericagade, 25 DK 1310, Copenga Copenhagen K, Denamark
- 9. Overseas Economic Cooperation Pund, Ine Bldg., 2-1-1 Uchiasiwai Che, Chiyeda Ku, Tokye, Japan
- 10. Private Investment Company for Asia S. A. (PICA), Kakusai Building, 3-12 Maruneuchi, Chiyoda Ku, Tokyo

Nany more sources of financing can be provided by UNIDO because UNIDO has sponsored international conferences attended by representatives of banking institutions and multinational companies. These meetings have served to identify organisations which are willing te invest in potentially viable projects in developing countries.

### Chapter 9

### TIME OF REFERENCE FOR FUTURE PRASEDILITY STUDIES

### 9.0 Purmeas of this Chapter

This Chapter has been prepared in compliance with Section 2,01.12 of Contract UNIBD Contract No. 74/21 dated September 23, 1974.

Section 2.01.12 of Contract 74/21 reads as follows:

"Based on the findings and conclusions resulting from the profeasibility studies, the Contractor shall write detailed Terms of Reference for carrying out subsequent work including, inter alia, a series of exhaustive technical and economic feasibility studies for the establishment of the industries identified by the Contractor as being potentially viable."

### 9.1 Basic Madel for Possibility Studies

We recommond using the standard format for feasibility studies contained in the "Technical Notes" at the beginning of <u>Extracts of</u> <u>Industrial Peasibility Studies</u>, Volume I, Industrial Planning Fories No. 7, UNIDO, published by the United Nations, New York, 1973. A bopy of the "Technical Notes" is attached as Appendix G.

Rather than repeating the standard wording from Appendix G for each of 8 types of facilities, we are only repeating the paragraph headings and are describing below the exceptions or special <u>Gavats</u> that relate to the situation in Vietnam or to the specific industry concerned.

These special points peculiar to Vietnam or to the industry are discussed below for 8 types of plants. The relevant paragraph designators, e.g. II, 2, are listed below, based on Appendix G.

### 9.1.1 <u>Rice Center (Receiving, Cleaning, Drying, Milling)</u>

### Major Input Materials, II, 2

A detailed study of paddy availability is required. This should include interviews with the Directorste of Agriculture in Cantho, the Agricultural Development Bank, large scale millers in Ba Xuyen large scale paddy farmers, and persons who buy paddy wholesale ("remasseurs").

The first key question to ask will be: "Can enough farmers be counted on to grow the same variety of paddy to supply a 15 TPH plant!"

A second key question is "What would be the best procedures for assuring the delivery of uniform paddy?"

### Alternative Technologies, Il\_, 3

The German Schuler system, the Japanese Kyowa and Satake systems, and U.S. systems at least should be considered and evaluated.

Suppliers should be asked for customer lists in developing countries and field visits then made to inspect the mills to determine operating history in at lesst three developing counties. Some countries have reported operating difficulties with rubber roller mills e.g. the Philippines.

### Locational Factors, II, 4

Nillers, farmers, and agricultural experts should be asked if the Tay Do Industrial Park poses problems of paddy delivery that could be avoided if another site were selected. (We believe, however, that the river-front location on the Bassac will permit low-cost transport of paddy by beat just as is being done in the Cai Bang District of Cantho.)

### Market, III, 1

The research team should have discussions with Vietnamese rice exporters to determine the acceptability of local TN rice in various fereign markets. Rice exports were stepped in 1964 prior to the introduction of TN rice, and there is no experience in TN exports. If such discussions prove inconclusive, the following further steps could be taken. The market for imported rice could be investigated in at least three of the principal rice importing countries and TN samples sent for evaluation. This should first be done by a data search of the te determine the imports of the nearby countries buying in large volumes (e.g. Singapore, Hongkong, India). See Table 3.2 below for importing regions.

### Table 9.1.1

### AVERAGE ANNUAL QUANTITY AND VALUE OF RICE INFORTED BY CONTINENTS AND WORLD TOTAL, 1962-1967

Centinent	Quantity (metric tons)	Value (milliens)
baropo .	864,400	\$ 130.5
VSOR	300,800	\$ 43.5
North and Contral America	407,600	\$ 67.0
South America	58,600	\$ 9.7
Asia and Middle Bast	4,799,800	\$ 644.6
Africa	687.800	\$ 96.0
Gooania	55,200	<u> </u>
Verld Total	7,174,200	\$1,000.1

### Source: <u>Rice Chemistry and Technology</u>, American Association Coreal Chemists, 1972

Notings should be held with the grain departments of at least two leading Japanese trading companies familiar with world rice trade. Samples af specific Vietnamese TN varieties should be tested by the trading companies and an assessment made of the suitability of the rice for various markets. (Veure suggesting these discussions take place in Japan because rice trade experts are concentrated in one location, and the Japanese trading companies, through their worldwide network of branches, would have access to information on world markets.

If a domand for imported parboiled rice is found (e.g. in India, Bangladesh, Bri Lanka) the study should include parboiling facilities in Para V, 1.3 of theAppendix G "Machinery and Equipment". (Parboiled head rice yields are higher than non-parboiled rice and shelf rice is lenger.)

### Product Nix, III, 3

In selecting the product mix, consideration should be given to the grade of paddy (e.g. TN 5,8,30 or 32 -- we suggest TN rather than traditional varieties because of the higher yields.)

### Land Site Development, V. 1.1

The cost of piling should be taken into account; see Chapter 4, Table 4.1.3, line A-18. This can best be done through the Masama Gumi of Tokyo, the only people with experience in building large structures on the Tay De Industrial Park.

### Merking Capital Annex, V, 2

The following payment practices should be investigated: (a) payments by mills or ramasseurs to farmere, (b) payments by brokers to mille. The need for working capital should then be assessed and entered in the cash flow analysie (Appendix G. part XIII).

The poseibility of loans for working capital should be investigated and the research team should schedule meetings with the potential leading banks to impress them with the importance of working capital leans, new prohibited by IDB policy in Vietnam.

### Other Investmente,V, 3

The cost of rubber rollers should be assessed (if the roller technology is to be used) eince frequent roller changes are necessary.

### Pricing Policy, VII, 3

The discount etructure as between dealers, millers and farmers should be determined.

### Planned Sales Organisation\_VII, 4

An assessment should be made of the advisebility of selling experts through brokers or direct from the mill to the overseas importer.

The research team should read Sources No. 27, 34, 40 and 46, including storage, marketing, and milling reports of the FAO rice team in September, 1974 (See Bibliography items Nos. 27, 34, 40 and 46 in Appendix F.)

Our suggestion is that the research team be composed of the following emports:

1. Techno-economist with experience in rice milling analysis,

- 2. Agricultural product marketing expert with experience
- in poddy procurement, rice exporting, and bran saler.
- 3. To obtain data regarding farmer's willingness to produce sufficient volumes of uniform pader, surveys could be made by them of professors and graduate etudents, based on a questionnaire under the direction of Dr. N.V. Truong, Ph. D. Vice Rector and Dean, Faculty of Agriculture, University of Cantho, 5 Hoa-Binh Avenue, Cantho, Vietnam Tel 21-061.

- 195 -

### 9.1.2 <u>Rice Bran Oil Plant</u>

This plant'is described in Section 4.4 of this report, and the costs are outlined in Table 4.4.8 on page 153. The annual capacity of the plant is 960 tons of edible oil, 408 tons of industrial oil, and 10,200 tons of defatted bran for feed. The capital required for fixed assets is about \$806,000 and \$862,000 of working capital is needed in the first year.

### Major Input Matsriale, II.2

The success of this plant depends on the availability of 40 tons per day of fresh rice bran containing a minimum of husk. The assumption was made in this rsport (Section 4.4.6), that the bran would be supplied by a new modern rice mill on the Tay Do Park (13 tons), by VIMECO a rics mill being planned for the park (5 tons), and by some of the larger mills in Ba Xuen or Cantho (22 tons). The 2 ton/hour Nghia Tan Mill in Ba Xuen is slready using rubber rollers and their bran should be especially good in quality with a minimum husk content. The SEDIC and Cuu Long rice mills near Saigon should be approached to see if they can ship fresh bran promptly to the Tay Do Park. These mills have 27 TPM and 9TPH capacity respectively.

The research team should make surs that mills are capable and willing to supply fresh bran in the required 40 TPD volumes.

### Alternative Technologies, II.3

We have recommended the solvent extraction method (page 151 of this report) for producing bran eil, however the research team should also consider the expeller process as an intermediate technology.

### Market, III

The Vietnamese demostic market for edible and industrial eil should be examined to determine the saleability of bran eil at higher prices than peanut eil or coconut eil. Expert markets especially in Singapore, Hong Kong, Japan should be studied.

The acceptability of defatted bran for animal feed should be studied.

The general practice in Vistnam's smaller mills is to mill paddy for individual farmers on a fee basic and to return the bran and milled rice to the farmer. Selling bran to the bran oil mill would be a departure from this traditional practice, and the millere' willingness to sell large volumes to the bran oil plant would have to be explored. The sales price of bran in large volumes would be critical to succes,

### Capacity of the Proposed Plant, IV

In Chapter 4, Section 4.4, we have recommended a 40 TPD bran plant, that is a plant having a daily processing capacity of 40 tens of fresh bran. This is a minimum acchomic size, and a larger mill would be better if circumstances warrant.

### Pricing Policy, VII. 3

The retail and wholesale prices of competing edible oils, i.e. peanut, soy, coconut should be studied. If the bran oil cannot sell competitively in Vietnam, export sales should be considered.

We suggest the research team should consist of (a) an expert familiar with the solvent extraction process for recovering oil from rice bran, (b)Agricultural economist to determine the marketing channels and pricing structures of rice bran at different distribution levels to determine sales prices likely to assure public acceptance of the edible oil, and soap industry acceptance of the industrial oils,

### 9-1-3 Mized Feed Plant

The mixed feed plant is described in Section 4.5 of this report. Capitalization in fixed assets is \$568,000, and annual capacity is 50,000 tons.

### Freescis, II.1

The mix of ingredients shown in Table 4.5-B of this report is in our judgement a nutritionally adequate mixed feed for chickens. Other mixes for (a) piglets of different ages, (b) mals and female hegs for breeding, (c) cultured fish are needed.

There may be farmer preferences favoring one type of feed over another, some of these preferences may be dietetically unsound, but nevertheless can affect sales volumes and pricing. It may be prudent to start with mixes that are less than ideal and gradually shift to better mixes later on.

### Major Input Materials, II, 2

Having determined the optimum nutritional mix for the above animals and fish, locally svailability ingredients should be identified and costs of volume procurement and the best procurement procedures identified. For ingredients not available locally, the cost of imports should be determined. These would probably include manioc from Thailand, and corn unlass local sorghum can be-used in substitution for corn.

### Locational Factors, II. 4

We have assumed that this 50,000 TPY plant would be located on the Tay Do Park. This would mean hauling mixed feed by truck or by beat. The location of potential users of feed should be analysed to see if any serious cost disadvantages would arise from siting the plant on the Tay Do Park.

### Product Mix III, 3

See commonts under "Products, II, 1" above

### Capacity of Proposed Plant IV, 1

We believe it would be useful to conduct a farmer opinion survey based on a questionnaire to determine (a) farmer preferences on types of feed, (b) acceptable prices, (c) volumes likely to be consumed, (d) seasonality of demands.(e) animals or fish being raised. The University of Cantho (Dr. N. V. Truong, Vice Rector indicated a willingness to assist in organizing a survey team.)

In the event that acceptability problems are indicated by the survey, the feed plant may have to Planned on a reduced scale, i.e. less than 50,000 tons, but room allowed for expansion.

### Marking Capital, V, 2

In studing the availability of raw materials (see above, "Major Input Materials"), seasonality factors involving sorghum for example (see Section 2.3, Chapter 2 of this report) may affect the need for working capital. Sorghum is harvested in December and the proposed plant would have to consider such options as (a) borrowing money te obtain sorghum at low prices after the harvest or (b) paying higher prices in the off-season.

### Planned Sales Organization, VII, 4

The sales channels for feed mixed at the existing My The and Long Xuen plants (see Section 4.5, page 156 ) should be explored. The desirability of having wholesale and retail distributors should be assessed.

### 9.1.4 <u>Engine Repair, Casting Shop, Agricultural Implement and Pump</u> <u>Manufacturing Plant</u>

This facility is described in the basic report in Section 4.6 starting on page 163. The capital required for fixed assets is about \$221,000 and annual sales of services and products could amount to \$200,000 to \$300,000 after two years of initial operations.

- 198 -

### Preducts, II, 1'

The product mix of the machine shep (see Section 4.6, page 163) should be tentatively decided in the feasibility study. Some candidate products to consider would be: medium capacity irrigation pumps of about 1,800 N per hour and small capacity foot pumps.

A farmer survey using a pretested questionnaire and photographs should be conducted to find out (a) what the farmers need and (b) what prices they would be willing to pay.

The organization and management of the engine repair shop could be recommended in the study. In 1967, a company called Trum Tem Name Co. was established by Government decree for the purpose of performing engine repairs at Cantho (see p. 190 of this report). If this company could be reactivated, good management of the engine repair shop would be better assured.

The size of engine parts inventories and the willingness of the tractor importers to finance the inventories should be explored. The principal tractor importers of 4-wheeled tractors in the autumn of 1974 were as shown in Table 3.4. A similar analysis of the 2wheel tractor importers and their sales would be needed.

### Table 9.1.4

MARKET SHARES IN VIETNAM OF FOUR-WHEEL TRACTOR IMPORTERS -- 1974

<b>.</b>	ertere	Mage of Sugalier	Market Share in Vietnam for 4- whoel tractors	Number of 4- wheel tractors imported in '75
1.	CINGECO	Massey Furguson, Ltd.	39%	1,400
2.	John Deere	John Deere	26	1,000
3.	ANTRACO	International Harvest	or 17	600
	(American Tradi	ng Co.)		
4.	<b>Bonault</b>	Renault	11	400
5.	Ford	Ford Notor Co.	3	100
6.	VIKYNO	Kubota Ltd.	1	30
7.	VINAPRO	Yanmar Diesel Engine	1	96
•		Co., Ltd.		
		•	100%	3.600

With respect to the casting shop, the existing market for castings should be explored and potential users of castings identified. This part of the work should be initiated after meeting with CARIC, the machinery and boat building company in Saigon, who have a 7-ton foundry making gears etc. We believe that the market for castings will be slow to develop because the industrial base in the Delta is small. Therefore this part of the complex might be deferred unless the production of implements creates a captive demand for castings.

### Inget Materiale, II. 3

Ingine parts, pig iron, steel, brass, sheet motal, send, pattern making material and voiding materials would depend on the forecast of sales of castings, pumps, and implements developed under the heading, "III Market" i.s Part III-1 of Appendix G.

### Alternative Technologies, II. )

The type of cupola and furnance will depend in part on what fuels are available at least cost. In this connection, the possible use of peat from An Xuon and Kion Giang provinces should be explored along with the suitability of coals from North Vietnam. These coals are new being experted from Hanoi to Japan.

### Helection of Product Mix, III, 3

See above under "Products, II, 1"

### Investment, Y

The following comments rolate also to "Annual Operating Costs and Profits" (Part VIII of Annual and "Financing Proposal (Part IX of App. G). As montioned in Section 4.6 of this report, the utilization of the engine repair, and casting shep, and machine shop may be slow to develop. This is because of (a) the lack of tradition for tractor maintenence, (b) some farmer poluctance to pay for first class tractor repairs, and (c) the practice of having castings made in Saigon.

Novertheless the high cost of fuel waste and the ultimate mod for motal working facilities when industry expands on the Tay Do Park are realities which require early remodial action.

The feasibility study should make assumptions as to costs and lesses or profits, on a year-by-year basis, for the first three years. We would suggest that the setablishment of the facilities might be justified as a Government investment and considered part of the necessary cost of building a foundation for future industry on Tay Bo and as a basis for training personnel. At the same time, makes of pumps, threshors, cultivators and customized parts should probably be leaked upon as the principal source of revolue in the start-up period.

### Implementation. X

Management contracts or other collaboration should be explored as follows:

Per pump manufacture: The Hyayen Van Xe Co., No. 20 Gia Long, Xa Ny Loung, Quan Che Noi-Timh, An Giang Provinge. For casting shop:

For the machine shop: Same as above.

### Survey Team

The team to make this feasibility study could consist of (a) a specialist in engine repairs to assess the needs for equipment and inventory and (b) a foundry and machining specialist.

### 9.1.5 Pertilizer Plant (Urea)

This plant is described in Section 4.7.1, page 167, of this report. Capital required for fixed assests is estimated at \$252 million. The output is a minimum of 480,000 tons of urea.

### Land Site Development, V. 1.1

Piling requirements for the prilling tower, ammonia compressors, and storage tanks should be discussed with the Hazama Gumi Co, of Tokyo which installed the steam turbine in the 30 NV thermal power plant at the Tay Do Park.

### Daildings, V. 1.2

This plant will probably employ more than 330 persons and dormitories near the site should be provided to assure manning of multiple shift operations.

### Machinery and Equipment, V, 1.3

In addition to the urea plant itself, cost estimates and site recommendations should be made for the naphtha pipeline and the storage tank, probably at Can Gio, opposite Vung Tau, near Saigon. In 1974, this port was being dredged to accomodate vessels with a 36 foot draft; the original plan was to complete dredging by 1976.

Our plant engineering associates, estimate that to supply 161,000 tens of naphtha, a continuously operating, six inch pipeline will be necessary. In Japan, such pipe costs \$6.85 per inch of diameter per meter of length. The installed costs in Vietnam will be about 61% higher; our estimates for the pipeline are \$8.9 million of foreign exchange (6 inch pipeline, 135 km long, Can Gio to Tay Do.)

The size of naphtha storage facilities at Can Gio or other port cannot be estimated until the status of the dredging program at Can Gio is determined. This will limit the size of incoming carriers.

### Vorking Capital, V. 2

The rate at which payments flow from the farmer, to retailers, to wholesalers and to the usea supplier should be estimated to determine the time lag in cash receipts by the supplier. On this basis, conclusions can be reached regarding the amount of working capital required.

### Inventories, V, 2.1

The speed of physical distribution from the plant to the farmer should be estimated and the size of urea storage capacity at Tay Do estimated to assure effective deliveries during the period of peak use. Figure 3.2-B on page 52 of the basic report shows the TN planting schedule.

### Total Annual Expected Maximum Output, VII, 1

The operation of multiple shifts for the CO<sub>2</sub>, ammonia, urea, and the bagging plants should be annalyzed to determine the combination of single and multiple shifts needed to raise urea yields to the maximum.

### Pricing Policy, VII, 3

Urea sold retail in the fall of 1974 for VN\$200/kg. All supplies were imported and it may be possible to reduce selling prices for locally produced urea and, at the same time, assure a reasonable financial return to the plant. On page 169 of the this report, the return on sales was estimated at 37% and the return on fixed asset capital at 16%.

### Financing Proposal, IX

On February 18, 1975, the IBRD announced the financing of a urea plant to be located as Ashuganj, 60 kilometers northeast of Dacca. The terms of the loan included a 50 year amortization schedule, a 1% service charge but no interest, on capital repayments, and joint participation in the foreign currency financing by a total of six lending agencies.

The above loan terms relate to the obligations of Bangladesh to the lenders. The loan terms applicable to the operators of the urea plant are less generous. In the financial planning for Vietnam's plant, the loan repayments of the operators should be adjusted in such a way as to assure viable plant operations, but yet achieve a maximum return to the Government of Vietnam from the plant operators.

### Implementation, X

The machinery suppliers should be required to provide start-up services for a period of one or two years and to provide training for Vigtnamese managers and engineers in similar plants built elsewhere overseas.

### Research Team for the Feasibility Study

The feasibility study team could consist of the following:

Executive Manager of an Operating Urea Plant, Project Leader Process Engineer, CO<sub>2</sub> Process Engineer, Ammonia Process Engineer, Urea Manager, Bagging Plant Financial Analyst, to estimate cash flow requirements. Agricultural Economist, to (a) determine sales prices for urea in relation to competitive products and historical price trends, (b) to determine prices likely to be acceptable by wholesalers, retailers, and farmers. (c) to estimate seasonal variations in demand resulting from double cropping of TN varieties, and (d)to estimate non-paddy requirements of urea, e.g. sorghum, vegetables, soybeans, (e) To determine desirability of mixing urea with phosphatic materials for a pre-mixed P and N material best suited to paddy cultivation.

### 9.1.6 Limestone Crushing Plant

This plant is described in Section 4.7.2, page 174 of this report. Total investment in fixed assets is about US\$2.0 million and capacity is 28 tons per hour or 216,000 tons per 300-day working year.

### General Comments

In order to gain farmer acceptance of crushed limestone, data should be collected from Vietnamese sources to determine the effect of limestone applications on acid soils in various sections of the Delta. (We were informed that tests were conducted by USAID and by the Directorate of Agriculture in Cantho and data recorded. During the survey in October-November, 1974, data were requested in Saigon, but there was insufficient time to collect the data from Cantho.)

### Products, II, 1

As mentioned earlier in this report, imported crushed limestone has been sold in the Delta for reducing the acidity of paddy field soils. There is also a new plant near Thot Not, An Giang Province, financed in part by USAID that should be on stream during 1975.

The prior experience in imported limestone and the experience gained by the Thot Not plant should be carefully noted to determine the quality and fineness of the crushed limestone that would be most acceptable to the farmers.

### Major Input Materials, II, 2

Estimates differ with regard to the limestone reserves near Ha Tien in Chau Doc Province. Some respondents said, "unlimited"; others said 20 million tons; others said, much less. In any case, the requirements of the Ha Tien Cement Plant in the Bien Hoa Industrial Park are substantial and several hundred thousand tons are being consumed yearly.

The Ha Tien reserves plus other reserves in Thua Thien Province and near Cam Ranh Bay should be investigated to make sure that reserves are sufficient for both future cement manufacture and for the proposed Tay Do plant.

### Locational Factors

The traditional practice is for tug-drawn barges to transport limestone through the Delta's canal and river network from Ha Tien to Bien Hoa. Because of the Bassac River and the connecting canals, we believe that transport by barge to Tay Do will be feasible, but this should be confirmed.

### Market, III

As mentioned, the farmer acceptance of crushed limestone and the preferred fineness should be confirmed and estimates of domestic demand prepared.

### Capacity of Proposed Plant, IV

The capacity should be based on the market findings in Section III, "Market" of Appendix  $\tilde{G}_{\bullet}$ 

### Working Capital, V, 2

Willingness of farmers to purchase at prices high enough to assure plant viability should be confirmed.

### Suggested Research Team

We suggest the research team include: (a) an agronomist, familiar with the effect of limestone applications on acid soils, to estimate optimum volumes needed for varying pH values, and to then estimate total demand in the Delta region. (b) Processing Engineer familiar with the commercial crushing of limestone, preferably for acid soil applications. (c) If limestone reserves are found to be adequate for future cement and agricultural use, techno-economist should be included to estimate the effect of limestone applications on total productivity and to determine export prospects. (d) A survey team using a pre-tested questionnaire should determine farmer acceptance of limestone and willingness to pay prices high enough to assure plant viability. This could be organized through the Agriculture Department of the University of Cantho.

### 9.1.7 Export Fromen Food Products Plant

This plant is described in Section 4.9, page 173, of the basic report. The capital investment in fixed assets is about \$415,000 and the proposed output is 1,200 tons of frozen duck, cultured river fish, or frogs legs. As the supply of feed improves, this plant could process chicken and pork, also for export.

### General Comment

We believe that the feasibility study for this plant should be approached in a different manner from those for other plants.

There should be a study of foreign markets for duck, especially in Singapore and Hong Kong where imports of poultry and the consumption of duck is high among the local residents of Chinese origin. Similarly, French duck imports should be explored. Importers of duck or wholesalers should then be identified in the market countries.

Samples of the Vietnamese duck species should then be frozen or vacuum-packed and sent to importers or wholesalers in the leading market countries to determine if the species is acceptable to consumers.

The same procedures should be followed with the type of river fish that can be best cultured but only after a local study of culture economics. In connection with fish culture, we invite attention to a Ministry of Agriculture publication, <u>Bulletin of Agricultural Economics</u>, <u>Special</u> <u>Issue</u>, <u>1974</u> pages 7 through 19, in Vientnamese with an English abstract (see next page for English summary).

The cited article contains data on culturing costs and technology, cultured species, and some market data. Despite the background information in the cited article, it would probably be desirable to confirm in Vietnam the availability of proper fingerlings, the optimum dimensions of rafts, and the economics of fish culture. This study, if favorable, should be followed by the same overseas market study described above for duck.

The foreign market for frogs legs in general has been confirmed, but the feasibility of frog culture, the economics involved and the feed requirement should be ascertained. We would recommend communicating with Dr. V. R. Pantalou, Fishery Adviser, Mekong Secretariat, Bangkok to determine where frog culture has in fact taken place. An inspection should then be made of existing facilities. This should be followed by a determination of the feasibility in Vietnam of duplicating successful practices of other countries.

After these studies, a market acceptance study should follow and the shipment of samples overseas as described above for duck.

### Major Input Materials II, 2

Aside from the economics of fish culture and frog culture, the procurement of duck in sufficient volumes should be studied as well. The volume of ducks available appears to be substantial in the Delta area. However, with the current feed shortage and shortage of chicken, there is a strong local demand and town market places are typically well stocked with duck. This means that the proposed frozen product plant would be competing with the traditional consumers. There may also be seasonal shortages and surpluses of duck because we believe fattening takes place just after the paddy harvests, twice yearly.

There should be a study of local river fish fingerlings and fish culture as mentioned under "General Comment" above plus a study of the frog culture economics.

### Alternative Technologies Available, II, 3

The freezing plants in Rach Gia should be inspected including the refrigerated vans being used to transport snapper, clams, cuttlefish and shrimp to Saigon for containerized shipment abroad.

The plants in Rach Gia use convential ammonia compressors, but freen technology could be considered as well. It is our recommendation that unless space is very limited and maintenance guaranteed, ammonia would be preferable to freen as a refrigerant. The main reason for this is that ammonia compressors are low stress, low RPM, long-lived machines.

### Locational Factors, II, 4

Tay Do Industrial Park would appear to be ideally located for fish culture, but there may be problems of riparian rights that would complicate the positioning of rafts or access to the rafts by attendants.

There may be transport problems of ducks from farms to Tay Do and the feasibility of transporting frogs to Tay Do from culturing ponds would also have to be confirmed.

### Capacity of Proposed Plant, IV

The decision on capacity would depend on the outcome of the export market study for duck and river fish and on the feasibility of collecting all three types of raw material as mentioned above.
### Conclusion on Frozen Food Plant for Export

All other Sections of the standard feasibility study in Appendix G herewith would depend on the sutcome of the foreign market and local supply availability studies mentioned above.

### Composition of the Research Team

We think that a techono-economics firm with connections in Singpore, Hong Kong, and France should conduct the overseas market surveys for duck and species acceptability.

A marine products specialist with experience in fresh water fish should study the feasibility of commercial fish culture in the volumes necessary for processing, for example at a level of 600 tons per year or more. The same expert should study frog culture economics with assistance from Dr. Pantalou of the Mekong Delta Secretariat.

The techno-economics firm should study the overseas markets for, and acceptability of, river fish and frogs legs. The river fish study would be confined to nearby Asian countries where the Vietnam species is likely to be acceptable. The frogs leg market study and the acceptability of the Vietnam species should be conducted in the developed countries, e.g. France and the United States.

### 9.1.8 Seed Center

This facility is described in the basic report in Section 4.8 page 171. The total capital investment in fixed assets is about \$31,000. Some parts of the standard outline in the Annex herewith are not applicable to the seed center since it is a service facility and no manufacturing is involved. However VII to XIII are applicable with adaptations.

We think the scope of work for a contractor writing the feasibility study could cover the following points:

- a. Determine the distribution channels for seeds in the Delta including exchange between farmers the existing wholesale and retail outlets their sales volumes, inventory, size storage facilities, adequacy of storage facilities and the present effectiveness of seed sorting and identification.
- b. Determine the pricing structure for seeds, the purchase price of seeds from farmers, wholesale prices to retailers and the ultimate sale to individual farmers at retail.
- c. Estimate the capability of the present distrubution system at current prices to disseminate enough seeds to plant TN paddy sufficient to yield 30,000 ton voumes of paddy for two crops annually (i.e. yearly total of 60,000).

- d. If the distribution system is inadequate to assure dissemination of enough and proper seeds, recommend the optimum storage capacity for a seed center at the Tay Do industrial park, taking into account the seasonality of demand (See Figure 3.2-B of page 52 of this report for TN planting season.
- e. On the basis of findings above, recommend optimally sized seed center, manning, distribution, sales prices, equipment, capital required, cash flow, profits.
- f. From the suppliers of rice milling equipment, identify the TN varieties that lend themselves best to milling and polishing with rubber rollers. (Suppliers in Italy and Japan know the varieties but other suppliers may insist on samples.)

### Suggested Composition of Research Team

We propose a team of two persons: (1) a specialist in the marketing of paddy seed from whoesale through retail levels and (2) a technoeconomist who would prepare the cost estimates in part (e) above, taking appropriate guidance from sections VII through XIII of Appendix G herewith.

### 9.1.9 Proposals for Industry Located Off the Tay Do Park

Village level mills to provide rice for local consumption are described in Section 4.3.1, page 144. These mills would be located off the Tay Do Park. The Kiskisan and Engelberg type mills now used in many localities are attaining a head rice yield of some 50 of 55% whereas a modern disk sheller can attain about 62%.

To introduce the improved disk sheller for local milling, literature could be obtained from the following firms. If the mills are suitable, licensing agreements or local assembly agreements could be reached with overseas suppliers of disc shellers.

Import licenses for improved types of mills only such as disc shellers should be approved.

Small scale limestone crushing mills, 350 kg/hour capacity, were described in Section 4.7.2 of this report, page 170 and in Appendix E. These machines can be obtained from almost any industrialized country.

> (Text of report ends here; appendixes follow.

### APPENDIXES

,

. 212
. 216
. 231
• 236
237
· 241 · 248

pa ge

ŧ

ł

------

### APPENDEX A

### STATISTICS OF WORK

(Encorpt from UNIBD Contract No. 74/21, dated 23 September 1974)

### 2.00 MERCHEINLITHE AND MURIE OF THE CONTRACTOR

### 2.01 Statement of Nork

The Contractor is required to propers a master plan based on the results of a series of pre-feasibility studies which he is to carry out to identify potentially viable industries which could be established in Can The, and form the basis of a vertically integrated agro-industrial complex. The components of the master plan shall consist of, but not necessarily be limited in, the following:

- 2.01.1 A general market survey to investigate and analyse the existing and future market demands to assortain which products should be produced to meet this demands
- 2.01.2 A general survey to ascertain the existing and future symilability of the rear natorials required to produce the products in demands

(Statement of work continued on following page)

2.01.3 The selection of industries for pre-frasibility study based on the results of the work carried out under 2.01.1 and 2.01.2 above and also the criteria agreed to resulting from consultations between the Contractor and the Industrial Development Authorities in Vietnam. Some of the criteria which can be used are listed in items (a) through (o) below. The following industries shall be examined as potential candidates for the subject of pre-feasibility studies:

rice processing; fruits and vegetables; sugar processing; vegetable oil; animal feedstuffu; dairy producte; meat processing; slaughter houses; tannery; tivestock-ranch using industrial methods for breeding and fattening high quality animale, fish producte and fishmeal; peanut processing; Italian poplar for pulp wood, manufacture of agricultural implements and equipment; forest products (low cost wooden houses etc.); gunny beg manufacturing; fortiliser plants; low lift fractional pumps and engines; seep.

- (a) Relative importance to the success of the complex
- (b) Exployment generation
- (c) Value added
- (d) Labour intensity (i.e. jobs per unit of investment)
- (e) Compatibility with the Pour-Year Development Plan of Vietnam (1972 1975)
- (f) Local aivailability of raw materiale
- (g) Availability of qualified labor
- (h) Current availability of infrastructure adequate to support the proposed industry
- (i) Availability of supporting industry
- (j) Proximity to national markets
- (k) Proximity of potential export markete
- Probable availability of financing (this means that industries requiring very substantial capital investment would be given a lower rating than those requiring a smaller investment)
- (m) Import substitution potential
- (n) Pollution characteristics and effect on ecology
- (c) Potential profitability to investors.
- P.OL4 Preparation of pre-feasibility studies for the establishment of optimumsism food processing industries to produce the semi-processed and/or final products in demand, and based upon existing raw materials or those which could be multivated in the future or imported.

- 2.91,5 Each pre-foasibility study shall contain a description of the industry studied, the remains why it is potentially viable, the size of the markets for its products, its production capacity, number of persons to be employed, fixed and working capital required, and its potential profitability. In cases where a pre-feasibility study indicates that an industry is not viable the reasons shall be given.
- 2.01.6 The industries selected for further etudy should be assigned a priority rating according to the Government's Four-Year Development Plan 1972-1975 and their importance as a component of the vertically integrated agro-industrial complex, and their interrelationship should be stated;
- 2.01.7 The master plan report and the pro-feasibility studies shall contain all of the data which has been used as the basis for preparing them;
- 2.01.8 A report on the existing agro-industries located in the Can Tho area, and indicate whether they could be technically and economically integrated into the complex in their present form, or by reorganisation. If these industries cannot be integrated into the complex the reasone will be given;
- 2.01.9 A review of government policies, legislation, taxation and infrastructure to determine any constraints which affect operating agro-industries or prevent them from operating in the most efficient and profitable way; and recommend actions on any changes needed to attract private investors, both local and foreign, to establish agro-industries in the Can The region; and advice as to the system of incentives and controls necessary to ensure an early agroindustrialization of the Can The area, in view of the considerable influence of the Bargon - Gia Dinh-Hien Hea metropolitan area on private investment activities.
- 2.01.10 The elaboration of a "strategy" for the rational development of a vertically integrated agro-industrial complex;
- 2.01.11 The industrial estate in Gan The shall be examined to determine whether it would be a suitable site for the agro-industrial complex:
- 2.01.12 Based on the findings and conclusions resulting from the pre-feasibility studies the Contractor shall write detailed Terms of Reference for carrying out subsequent work including, inter alia, a series of exhaustive technical and economic feasibility studies for the establishment of the industries identified by the Contractor as being potentially viable.

- 213 -

- Sol.13 The Contractor's draft and final reports must include an "Executive Summary" containing a short commary of the conclusions of the study to facilitate the work of these responsible for deciding whether or not it is technically, accommenting and financially feasible to establish the agro-industrial complex.
- 2.01.14 The tixt of the final report would be organised into chapters conforming with the foregoing outline, items 2.01.1 through 2.01.13. Supporting appendings would contain the basis data on which recommendations would be based.
- 2.01.15 The components of the master plan listed above are not necessarily enhanced to and the Contractor is required automatically to carry out any additional work which he finds essential for the preparation of the plan.
- 2.01.16 The Contractor is required to cross-reference the work performed with the work specified shows in items 2.01.1 through 2.01.13. This cross-referencing shall be presented in a separate supplement to the draft and final reports.

### 2.02 Contractor Services

۶

For the performance of his obligations under this Cantrast, the Cantraster mhall make available a total of 148 man-days of aprvice, as follows:

### a) Project Area Berriges

T' nun-days of service shall be carried out in the Project Area by a team comprising the 'Peam Leader and other personnel as named in paragraph 2.03 horninafter. Time spont in briefing and de-briefing of Contractor's personnel in UNIED, Vienna (see personal 2.04 hereinafter) and travel time to and from the Project Area or UNIED, Vienna, are not included in the number of man-days stated above. A man-day of -mervice in the Project Area is defined as an eight here work-day.

### b) Home Office Bervices

71 man-days of service shall to carried out at the Cantraster's Home Office by personnel as manned in paragraph 2.03 hereinafter. Per purposes of this Cantract, a man-day of service in the Home Office is

### APPENDIX B

### EXPORT MARKET APPENDIX

### COMPARISON OF (A) VIETNAMESE EXPORTS OF PRODUCTS NOW PRODUCED OR POTENTIALLY PRODUCED IN THE DELTA AND (B) IMPORTS OF MAJOR FIFTEEN CONSUMING COUNTRIES

### Implanatory Comments

1. The purpose of the graphs and data which follow is to show the potential outer limit of Delta export markets in countries which are Vietnam's largest trading partners. There are, of course, other countries where markets can also be developed in due course, especially markets for the Delta's surplus rice. (In 1972, the 15 countries reported herein imported 1,206,999 metric tons of rice valued at \$191 million or \$156 per metric ton.)

2. The following Vietnamese export data, on the right-hand side of each figure, are for calendar year 1973 and are taken from the IBM Printout of the Directorate of Customs, Saigon.

Import data on the left-hand side of each figure show the total imports of the 15 countries which have been the largest consumere of Vietnamese products in recent years. The import data are from U.N. sources where possible, and in three cases (Australia, Indonesia and Taiwan) are from the official import statistics of the countries concerned for 1972.

The 15 countries are: (1) Austrelia (2) Belgium (3) Canada (4) France (5) Germany, West (6) Hong Kong (7) Indonesia (8) Italy (9) Japan (10) Metherlands (11) New Zealand (12) Singapore (13) Taiwan (14) United Kingdom, and (15) United States.

It was not possible to find data for both (a) the 15 country imports and (b) Vietnam exports for 1973. Vietnamese data for 1972 were, of course, available, but Vietnam's exports in 1973 were used because they are considered as being more representative of the near future export pattern. Also the breakdown of 1972 exports from Vietnam was too broad.

3. The 1973 Vietnamese exports and the imports of the 15 counties are expressed in thousands of U.S. dollars.

4. In three cases (Australia, Indonesia, and Taiwan) where U.N. import data were not completely available, the value of imports was converted to U.S. dellars at the average exchange rate prevailing in 1972 as used by the U.S. Department of Commerce for trade reporting.

5. Abbreviations used in the following filles are: BTN-Srussels Teriff Nomenclature; SITC-Standard International Trade Classification; "neg."= negligible, that is, the figure was less than one thousand U.S. dollars.

- 215 -

### MEKONG BELTA PRODUCTS THAT HAVE HIGH POTENTIAL FOR PRODUCTION AND WHICH ARE IMPORTED IN SUBSTANTIAL VOLUMES BY VIETNAM'S MAJOR FIFTEEN TRADING PARTNERS

(Unit: Thousands of US\$)

(A) Rank in terms of market size in column D	(B) Product Potentia- 11y Produced in Mekong Delta(BTN No.)	(C) Amount Exported from Vietnam in 1973	(D) Amount Imported by 15 Major Tra- ding Partners in 1972
lst	Animal Feed(230700)	0	\$1,962,678
2nd	Shrimp, læbster, frozen (030300)	\$12,801	966,424
3rd	<b>Veget</b> able oils (includes rice bran oil) (150720)	0	754,919
4th	Dressed poultry (020200)	7	295,296
5th	<b>So</b> ap (340120 and 340190)	4	2 <b>80,</b> 425
6t <b>h</b>	Salted, dried fish (030200)	152	200,034
7th	Rice, glazed or polished (100630)	0	191,189(*)
8th	Vegetable for animal feed (includes bran) (120890)	7	110,331
9th	Other frozen animals including frogs legs (020400)	40	70,117
10th	Live poultry (010500)	4	57,024
llth	Cereal flour, except who flour (including rice flour) (110100)	bat O	4,090

(\*) Much larger markets exist if more than 15 countries are included, e.g. OPEC, India, Pakistan etc. 6. The Delta is ideally suited for the production of some of the products listed in the attachment but less well suited for others. Some products could be grown in the Delta but only at the sacrifice of more valuable crops. A few of the products would not be as suitable for cultivation in the Delta as in other regions of Vietnam. For the sake of completeness, we have included all types of products, ranging from those with a high to a low potential for cultivation or production in the Delta.

As a rough indicator of the Delta region's capability to produce or cultivate the products, up nave graded each product by a letter symbol, A, B or C. A means, high potential; B means, moderate potential, C means low potential. These ratings are based on the findings of the survey made prior to this report. (\*)

### 7. <u>Conclusions from this Appendix</u>

Only a fraction of the import requirements of Vietnam's most important trading partners are being met from the Mekong Delta at present. However, in the case of a few products, the Delta's contribution is more substantial. For example, in the case of feathers, chiefly duck feathers, Vietnam is supplying 2.9% of the requirements of the 15 trading partners. Of these feathers about 83% originated in the Delta (see <u>Bulletin of Agricultural</u> <u>Economics</u>, <u>Special Issue, 1974</u>, page 53, showing duck production; we used this as the basis for estimating the 83% figure). The export of feathers by Vietnam in 1973 was valued at US\$1.3 million whereas the imports of the 15 countries were valued at \$45.5 million (1.3/\$45.4=2.9%, cited above in this paragraph).

In the case of frozen shrimp and lobster, Vietnam's exports of \$12.8 million in 1973 amounted to about 1.3% of the imports of the 15 major trading partners. The Delta contributed about 95% of the exports (judging by the Bulletin of Agricultural Economics cited above).

Probably the most important conclusion from this appendix is that there are several products which the Delta could supply in larger volumes than at present, given better organization of production, and, in the case of rice, changes in policy to encourage exports. Some of these are listed below together with the recent year import volumes of the 15 major trading partners.

### (Table follows, next page)

(\*) The cited source is listed in the Bibliography as item 19.

## DEPORTS BY FIFTERIA COUNTLIES OF PADDUCTS CUMMENTLY OR POTENTIALLY PRODUCED IN THE NETONIC DELTA OF VIETNAM --- 1972 (See Explanatory Notes on Page )

(Unit: Quantity=Matric ton, Value=US\$1,000)

		SITC		calia	(2) helgi		(3) Canad		(Y)	
		-								
			Quantity	Value	Quantity	Alle				
	l time and the final dark)	• <b>190</b>	1	•	7,071	468.4	10,180,000 (H)	4.221	1,274	3, 392
		4-110	-	8	6.839	7.417	5,150	3,927	5,280	4,543
		011.090	10	23	14/.4	6,461	10	8	15,555	18,245
		1.11		16.751	121.04	34,178	16,121	16,705	129,776	114,027
		2.160		3.701	er. 6	5,420	1,332	1,610	20,563	14,546
		031.3	8	116.6	28,070	15,724	10.684	110,16	61'16	79,026
		<b>661.6</b>		14	2,543	1,546	592	519	154.4	2,548
		196-162	. 1	<b>\$</b>	*	520	1	274	2,807	5,461
		054.8		180	430,107	40,360	<b>n.</b>	4,289	154,001	15,490
		054.2	8	3,075	56,603	10,109	11,243	3,990	123,057	104.16
		1.420	1	•	219,544	13,457	106,553	8,458	172,392	20, 325
		051.1	1	1	196,472	St0. ME	206,782	39,631	814,127	141.400
	Manual fresh	051.3	•	•	34,086	14,115	207,28M	35,819	506,526	109,326
	Muts edible, fresh or dry (incl.	051.7		10,545	10,571	12,090	168,62	31,321	<b>666</b> *E <b>†</b>	53,991
	cocomets) Presh fruits. n.e.s. (incl. pimespie	051.9	8	2962	78,722	25,010	966'0/1	42,046	164,051	55,236
		074.1	1	22.573	1.412	2,349	23,837	24,625	4,130	8,256
		075.2	2	1,100	1,130	1,666	2,549	3,526	5,653	8,003
		1	ŝ	260	1.409.000	150,200	429,671	27,152	250.043	20,942
100630	tice, glazed or polished	042.2		66	35,119	5.928	766,62	6.413	100,545	26,571
110100	Cereal flour, encapt flour of wheat	0.7.01	8	13	8,647	1,071	19	<b>40</b>	2,403	87
011011	Flows of wheat or of meelin	0.946	•	2	14,359	2,500	<b>8</b> . <b>9</b> .		315	1
110600	Vegetables etc. preserved of preserved (jac]. mailoc)	55	i	1,525	19. 10	35,196	181.61	36,582	104,524	30,525

(Note: M:Number, V:Volume in cubic motor, L:Litre, A:Area in 1,000 mg. motor)

MEMORY IN FIFTHER COUNTLES OF PROBUCTS COMMUNITY OR POTENTIALLY PROPUSED In the neuros delta of vietness — 1972

	~	
	Ż	
	8	•
	Here.	
	ter.	
ł	Ű	

Ż	
8	4
Ne ce	
L'oge	2
ł	,

(Coec Jan

- 219 -

0010/0 005000

054.8

31,193

261,645 77.760

182,507

36,520 35,645

B.B.

74.740 205,546 3,000

414,543

8 8

4.630 166,93 15,949 20,104 20,239 25.229 3,770

117,035

345,566

4,573 147,669 30,374 10,964 26.473

13,479

8-8-

B. A.

,063,000

56,435

351,750

8 8 B

2,047

106,052

8,030 19, 302 7.569

15.851 118.94

137,751

112,228

051.7

8

1.519 32,005

24, 760

17,173

026,000

0090/0 00100

110,563

054.2 054.1 051.1 051.3

118,319 20,839

157,385

896,571 669,002

3,924

13,369 22,113

81

29.379 53,867 21,942

10,622 504,691

291.961

00/000

24,277

1.7

9.130

20,582 65,957

27,638 72,566 18,921 33,521

15,928 12,492 4,164

13,146

38,695 24620

\$

2

8

51, 795

743,196

2,184 15,024 40,145

551

ğ 8,746

516

14,000

2,925

16,146 1.7 3,197 14,786

14,348 38,436 23,604

46,106

42,185 28,822 1,922

X -

2.64

7.983

20,445 8,967

34,152 12,819

....

10.3

27.666

17,000 

5,06

031.3

0000000

. 2 8 8

431 5 Ĩ

: . 2 .

18,865 1,203

167.61

111,003 61.19 292,646

3,213

12,570

12,030

104,592

124,643

.

5.6 1,96 Ĩ

2,185 4.097 28,910

2,343 4,416

22,402

29.278 3,596 131,151

4,633

4,591

1,8

3,497

5,627

2,203,000(m)

12,066

15,973

....

ji

16,002

28,209 30, 314

2,955

5.8

4.18

005010

.

8 . .

16,361 254

232

24,219

011.890

1.160 031.2

224.470

278,706 21,005 208, 375 29.855

4.110

020200 020400 030100 030200

.

Value

Quantity

Value

Questing

**Val e** 

Value

Quenci Ley

Value

Questiy

Value

(S) Cerr Quest cy

ĔġÌ

EjĮ

(e) Ine Long

(0) Italy Quantity

(7) Indonesia

**.....**(6)

(10) Netherlande

(Mait: Quantity-Matric ton, Valua-1994),000

216,212

2,483,000

378,769 373

6,052,000

316,587

4,842,000

2,230 32,166

2,129

2,610

32,330

1.436 9,116

> 134,326 155,876

269.544

3,280,000

¥

11,636

10,465

009060 100500

7,623 2,452

15,472

10.007

074.1 075.2

00200

213,836

662,761

051.9

4,000

12,147

34,878

5,917

36,313

3,197

1.831

17,554 3

8 - **B** - **B** 

6.00.3 23,900

\$5,272

16,639

72,673

042.2

100630 110100

.....

2,139

13,874 78,149

4

222,000 66.760

2

35 47.706

. .

2 I

2.444

26,973

5,942

38,234

110110

ł

4,744

047.01 046.01 X, 312

57.061

273.947

627,963

3

11066

\$

3

Ş

111.62

33,242

26,783

# DEPORTS BY FIFTERN COUNTRIES OF PHOBUCTS CONMENTLY ON POTENTIALLY PHOBOCED IN THE MEDDIC BULLA OF VIETNAM --- 1972 (See Explanatory Notes on Page ) (Continued)

E	SIAC	(11) Nue Zai		(12)Steere		131 7.1							
3	1							··n (+T)		(12)	l.s.	ž	otal
Ĭ	Ĭ					Quantity	Value	Quantity	Value	Quantity	Value	Quanity	Value
10500	<b>001.4</b>	151,593 (8)	127	5,430(II)	2.147	0	2		1				
20200	011.4	- <b>1</b>						(w)nonderet v	7/047		1,10	9/,103	57,024
					C2142	•		9, 333	7,032	1	9-9	375,095	295,296
		•	I	1	I	0	n	I	I	166	1.000	48.606	20,117
	031.1	8	100	<b>D.A.</b>	7.401	1	8	<b>B.A</b> .	82.138	704 201	029 535	716 317 1	
90200	2.160	8.8	8.8.	5,952	2.210	0	01	3.216				417°CT4°T	112.812.1
0000	031.3	:	1	18.000	241 01	-				C7C*nc	*01°/7	184°412	¥60°00Z
		8	1			•	-	9./36	26,428	n.a.	442,379	305,831	966,424
		1	1		67	•	•	16,462	9,629	17,672	8,900	116,443	65,327
8	14.167	1	•	1	I	•	1	1,989	369	2,284	4.869	34.185	45.438
	8.40	8	<b>X</b>	10.994	111	175	519	27,213	7,192	0.9.	16.048	1.958.568	215.806
020	054.2	1,429	510	22,048	4.269	8	3,182	154,702	43.094	18.933	4 070	1-172-414	212 212
890	054.1	9.9	i	28,184	2,648	0	I	280,607	43.613	7	000 6	2 307 657	CT/ COCT
0100	051.1	18,863	161.4	41.854	10,524	0	1	446.882	20	30C 97			
	051.3	26,926	5,207	20,481	918	ñ.2.	E E		100 05	000 000 1			10C • /0C
_	051.7	2,520	1,780	45,645	2.343	ı	163	877 TA		000°044°T	000 161	21 <b>8</b> , /26, C	ECB. 44/
	051.9	1,049	ŝ	78,702	8.055	ı		111 111			220°CTT	95 C. 984	494,236
0200	074.1	8.117	8.623	1.427	9	I			<b>K0</b> <sup>(7)</sup>	:	860° IE	1,629,756	511,739
800	075.2		Ĩ			I	1	212.240	205,706	68,717	62,827	395,935	417,091
200	1	8	i		<b>5</b>	4.4		7,066	4.928	26,813	32,914	153,585	141,808
				R / f r T	2	X,	<b>R</b>	3,145,000	209,594	31,151	090.4	22,175,788	1,605,875
	7.7 <b>5</b>		P.1.7	101,955	44,676	0	1	71,969	13,915	14,157	1,811	1,206,999	191,189
3	5.5		-	I	1	8.8.	I	1,462	26	14,157	1,811	32,030	060.4
0111	10.040	: :	8	ı	•	I	1	64,412	1,136	ı	1	380,922	14.381
8	055	8	558	43,504	13,607	B.8.	<b>.</b>	265,955	82.260	0-9. 1	169 004	1-451_060	785 440

- 220 -

.

### DUCTS CUMMUNITY OR POTENTIALLY PROBUCED Law of Vietnam - 1972 Mainteau

î

(Dait: Quantity-Matric ton, Valua-US\$1,000)

		SITC	(1) <b>An</b> tra	114	(2) heig	5	(3) Canad	9	(f) Pra	
į			Quantity	Value	Quantity	Value	Quantity	Value	Quantity	Value
120110	Ĩ	221.1	<b>8.6</b>	451	3,150	1,145	53,978	14.836	148,685	39.230
120120	Cocommet, will h	221.3	Ċ	~	<b></b>		8.8.	8.8.	20,707	2,510
120130	Soybeens	1.122	7	s	336,928	44,485	306,482	367.65	458,447	56,996
120700	Medical herte	292.4	<b></b>	÷.	2,472	1,429	8.8.	ž	13,046	8.779
120090	Vegetable for animal feed	1.190	ł	9	1 83,052	9,150	81,346	3, 359	6,288	452
140200	<b>Kap</b> oc	292.941	•	1	ı	1	3	33	~	2
150720	Vegetable oil (Soybeen oil)	421.2	3,597	2	16,120	1.321	17 <b>,01</b> 2	4,755	39,562	13,006
	(Cotton seed oil)	421.3	X	101	102	21	10,191	2,897	40	m
	(Sumflower seed oil)	421.6	ž	8	41,506	14,434	1,926	623	62,478	23,529
	(Repe, colze, mustard oild	A21.7	111	125	5,770	1.444	n.a.	n.a.	6,169	1,8%
	(Liness oil)	422.1	1,863	426	2,832	579	B	£	11,907	2,558
	(Palm oil)	422.2	7,168,448(L)	1,400	22,732	5,291	30,861	5,776	41,532	10,466
	(Cocomut (Copra)oil)	422.3	9,782,638 (L)	2,376	22, 317	5,510	32,295	6,374	37,978	9,745
	(Palm kermel oil)	422.4	<b>493, 352 (L)</b>	106	4,883	1,425	5,749	0,270	28,870	6.900
	(Castor oil)	422.5	1,806	1,156	4,215	1,873	2,171	1,046	39,066	15,413
151200	Animal fats and oils	164	8.8.	2,460	61,968	17,060	24,798	9,282	125,661	36,750
200120	Vegstable preserved or prepared	055	r	7,525	60% °61	35,196	767.61	36,582	104,524	50,525
200200	(including canned wegetable) Pruit preserved or prepared	023	F	4,452	93, 342	367.66	n.a.	88,412	284 ,893	696,98
200500	(including canned fruits) Fruit or vegetable juice	053.5	Ŧ	765	28,519	11,423	r	44,086	146,203	29,380
200600	(including canned juice) Pruits and nuts. n.e.s. preserved	053.9	r	1.568	50.168	20.272	88.324	33,259	103.541	44.405
230700	Animal feed	Ĩ	I	7,230	2,038,000	191,270	359,106	43,535	2,038,000	246,716
330100	<b>Essential</b> oils	551.1	£	3,286	197	1,337	1,011	8,109	5,674	47,330
340120	deog	554.1	ł	1,164	9,428	6,518	4,207	3,494	7,119	4,580
340190	Other soap	554.2	F	5,064	72,279	28,817	n.a.	23,082	95,785	42.948
442700	Improved, reconstituted wood	631.4	I	687	51,518	8,383	9,525(A)	6,597	1 20,688	17,551
570400	Jute	264	12,269	1,656	85,994	25,450	2,447	734	51,671	16,060

### DEPORTS BY FIFTERN CONTRILS OF PADDUCTS COMMUNITY OR POTENTIALLY PRODUCTS In the Neurone Delita of Vietnam -- 1972 (Continued)

(Unit: Quantity-Matric ten, Value-US\$1,000)

E	SITC	(S) Cert	Î	(9)		() Imbo		(8) Ita	Ly	adar (6)	4	(10) Neth	er lands
ġĬ	żİ	Questicy	Value	Que a l'est	Value	Questicy	Value	Quest ley	Value	Quantity	Value	Quantity	Value
120110	221.1	55,836	18,769	10,873	3,775	i i i		956,111	30,131	62,325	24,304	50,230	17,109
120120	221.3	24,719	2,734	i i	#-A-	8	z	£	3	21,682	2,083	233,418	28,339
120130	221.4	2,237,000	204.415	18,543	2,740	8	8	818,642	106,612	3, 396,000	474.175	1,631,000	213,807
120799	292.4	22,793	18,973	17,446	42,056	875	181	7,340	5,013	12,938	10,218	5,117	3,315
120090	081.1	321,485	21,638	6,874	451	<b>B. J</b> .	i	828, 18	5,116	521,994	2,6	529,663	26.027
140200	292.941	1,219	3	•	7	8	8	593	16	5,779	301,280	*	13
150720	421.2	26.719	7,885	4,103	1,322	8	8	12,297	3,852	19	274	14,155	3,900
	421.3	26,274	8,786	i i i	<b>B</b> - <b>B</b> .		8	12	12	13,648	4,685	467	243
	421.6	146,262	49,065	:	:		8	2,354	855		B.B.	160,84	14,617
	421.7	12,907	3,431	29,099	9.26	2	:	56,840	16,045	3,265	822	10, 345	2,455
	422.1	37, 324	7,213	921	118	262	102	12,508	2,555	4,751	26	2,36	7,138
	422.2	151.375	34,333	8.8.	i i	<b>B. B</b> .		51,973	12,552	54,865	11,202	165.373	319" %
	422.3	44,534	9.2%	<b>%</b> ]	5	Ŧ	8	21,265	5.867	10,741	2,045	76,186	17,846
	422.4	22,021	5,637	<b>R. 2</b> .	n.e.		8	15,765	4.419	A. A.	<b>B.B</b> .	1,969	567
	422.5	10,299	4,514	8	2	T	8	5,401	2,358	2,855	1,289	8,054	3,222
151200	431	<b>667,621</b>	28,582	264	153	1,284	\$	<b>1814, 2</b> 5	15,215	10,757	5,590	61,850	16, 784
200100	655	627,963	273.947	57 <b>,06</b> 1	26, 342	8.8.	B. B.	47,706	26,783	66,760	33,242	78,149	111.62
200200	053	802,606	287,378	22,545	10,515	t	8	50, 796	16, 366	<b></b>	44,716	150,132	61,233
200500	053.5	283, 365	105,922	4,649	2,707	622	148	16,525	5,121	(1) 676"6	6,257	51,219	26,569
200600	053.9	417,437	140,700	16 <b>,8%</b>	7,276	1,246	217	27,701	8,304	70,600	27,375	62,771	22,161
230700	1	4, 396, 000	461,289	132,297	14, 363	<b>.</b>	ļ	1,160,000	190,541	1.407,000	121,757	4,243,000	024.766
330100	551.1	3,164	19.047	270	1,257	<b>E9</b>	728	1,287	5.774	3, 383	25,456	1,391	8,310
340120	554.1	4,543	4.972	5,966	3,493	3,046	959	4,135	3,242	3,226	4,133	5,786	5,567
061096	554.2	51,995	25,122	15,257	5,663	456	146	57,992	30,231	11,487	9,405	69,745	32,281
442700	631.4	287,997	33,827	n. å.	8.2.	<b>7</b> . <b>6</b> .	<b>.</b> .	23,097	3,752	2,944 (V)	352	262,435	32,896
570400	264	53.140	14.906	5.450	1.199	:	2	13.699	4.383	61.091	15.996	13.756	3.668

### A CALIFORNIALLY Î Ē 2 STATE & STATE

Î Į

	SINC	(11) was and		(12) \$1	-			(14) 1.	<b>K</b>	(15) 0.1		F	) tal
ġ	j	Quantity	Value	Quantity	Value	Questicy	Value	Questicy	Value	Questity	Value	Quantity	Value
120110	21.1	5,096	1,956	162.4	1,481	•	•	64.A3	21,0%	1,286	583	572 <b>,895</b>	174,846
120120	221.3	<b></b>	÷	1,759	613	•	1	28,148	3,370	8.8.	B. A.	160,766	39,714
120130	221.4	8	8	63,065	614.6	223	\$	536,536	72,372	8		9,807,975	1,306,595
120700	292.4	8	8		3.6	s	•	3,500	3,196	1	24, 396	85,612	128.400
120070	<b>001.1</b>	8	8	25,715	1.070	•	1	564,85	1,301	62,383	6,212	1,847,121	110.331
140200	292.941	8	8	18	-	•	1	•	10,295			7,823	311,732
150720	421.2	2.792(V)	1,152	ž	Ţ	I	I	62,680	17,401	I	8	199,323	59,062
	421.3	(1)(22	<b>7</b>	ė		•	•	30,037	19,286	ł	8	81,997	27,320
	421.6	(2) 442	1	1	8	•	1	28,789	651.6	I	2	178,926	113,134
	421.7	174(7)		1	:	•	•	8,720	2,167	4,713	1.33	138,713	39,208
	422.1	<b>.</b>		8	8	1,869(L)	1	33,418	7,009	n.e.		139,553	28,728
	422.2	2	8	8	42,776	7	I	207,998	45,908	195,548	38,739	929,427	243,061
	422.3	8	2	16,173	3,051	~	•	50,304	12,636	544,146	66,412	663,902	141,441
	422.4	8	3	2,244	2	ı	•	58,663	14,795	45,822	12,000	186,499	47.717
	422.5	*	8	218	110	1	1	19,201	7 .441	61,14	16.804	134,561	55,228
151200	164	8	1×	<b>8</b> - <b>8</b> -	1,189	<b>8</b> - <b>8</b> -		75,950	19,728	8,579	8.384	530,334	161,948
200100	055	1	2	43,504	13,607	ł	8	265,955	82,260	п.а.	169,096	1,451,068	785.440
200200	650	1	3,316	78, 790	23,996	1		588,144	212,780	z	127, 361	2,071,250	1,010,312
200500	053.5	S3B(V)	258	5, 320	2,527	1	1	155,962	45,808	(A) 734 (A)	38,290	1,047,636	319,201
200600	053.9	7.924	2,846	166,23	18,195	ı	•	376,496	143,604	922	647	1,289.955	470.849
230700	1	<b></b>	392	557.687	39,800	24	12	1,589,000	210,756	n.a.	97,624	17,920,114	1,962,678
230108	551.1	71	575	280 (V)	1,136	-	-	N.a.	31,198	0:030	48,969	24,212	203, 315
340120	554.1	ë		4,109	3,730	I	I	482	<b>36</b> 4	1,791	3,260	53,840	45,976
340190	554.2	4,640	2,926	6,801	3,974	I	•	20,881	15,486	22,795	9,304	430,113	234,449
442700	4.168	8.9.	<b>B. B</b> .	25	118	I		563,287	77, 383	15,967	2.496	1,378,296	184.044
570400	ž	3,057	658	1,774	619	•	•	95,741	35,724	13,421	3,645	413,510	124,638





- 225 -



### - 226 -

والمراجع والمحمومة المعرفين والمعامين والمحموم والمحموم والمحموم والمحموم والمحموم والمحمور والمحمور



- 227 -

. . . . . .





### Appendix C

### CHARACTERISTICS OF SOIL AND WATER IN THE VICINITY OF CANTHO (By Tadayuki Aoike, member of the Cantho Agroindustry Survey Team.)

### 1.0 Soils

### 1.1 Physical Properties

The soil of the paddy growing region is olive-gray in color and clayey (called "cat clay" in some literature). The soil consists of fine particles and is impermeable to air.

### 1.2 Chemical Properties

### 1.2.1 Acidity (PH)

The soil acidity ranges from values of 4.3 to 5.0 after analysis by the KCl method. It is therefore necessary to raise alkaynity of the soil by applying limestone to the soil.

### 1.2.2 <u>Phosphoric Acid</u> (P<sub>2</sub>O<sub>g</sub>)

The soil tests disclosed no  $P_2O_g$  content whatever. Therefore phosphastic fertilizers are important for agriculture in the Delta.

### 1.2.3 Iron (Fe, and Fe,) Content

The tests showed that content iron ranged from 1,500 PPM to 2,500 PPM so that the iron content is suitable for paddy production.

### 1.2.4 Calcium Oxide Content (CaO)

The CaO content was found to be 0.07% to 0.2% and is therefore sufficient for paddy growing.

### 1.2.5 Magnesium Oxide (MgO)

The content of MgO was found to be 300 PPM to 350 PPM. This is a very soluable element and is radpidly leached from the soil.

When the pH value falls below 6, the water soluable NgO sharply increases and tends to be washed away. However the content of the Delta soil contains sufficient NgO for paddy production. The minimum content of NgO is 100 PPM is required for paddy growing.

### 1.2.6 Nitrous Antimony (NH\_N)

The content of  $NH_{L}N$  in the soils tested was 10 to 50 PPN which is barely sufficient for paddy. Nitrogen in the form of  $NH_{L}N$  is less than 1% of the total nitrogen content in the soil.

Organic nitrogen is generated mainly by bacterial action and is then absorbed by growing plants and vegetables. Therefore it is important to break-up and loosen the soil to encourage the growth of bacteria and permit root ventilation.

### 1.2.7 Soluable Alumina (Al.)

We found 250 to 300 PPM of Al, which is too much for ideal growing conditions, A content of less than 100 PPM is more nearly ideal. It is therefore necessary to deactivate a portion of the alumina by raising the changing the pH values to about 6 by the application of limestone and to plow-under husk or straw.

### 1.2.8 Potash (K,0)

The potash content is nearly zero whereas the desirable content should range from 8 to 15 PPM. More green manure should be applied or the ash of burned plants and foliage, husks and straw.

### 1.2.9 Manyanese Oxide (MnO)

More than 50 PPM of mangamese oxide was recorded in the tests. This is nearly an ideal content. The mangamese is stable when the soil is acid or when the mangamese is deoxidated. When the soil is dried out, or when the pH value is above 7 (i.e. the soil is alkaline, the mangamese oxide becomes unstable and makes it difficult for plants to absorb.

### 1.2.10 Phosphorous and its Absorption Coefficient

We found that the absorption coefficient of phosphorous ranged from 1,000 to 2,000. This measurement indicates the capacity of the soil to absorp phosphorous, and a measurement of 1,200 is the dividing line between good and poor absorption. The higher the measurement, the worse for growing plants. Much of the Delta Soils have too high a coefficient for ideal agriculture, and thus the soil supplies too little phosphorous as a plant nutrition.

### 1.2.11 Humus Content

The 5 to 9% humus content found is excellent for agriculture.

### 2.0 Characterics of the Mekong River Water

### 2.1 Introductory Comment

Among the elements contained in the water, those that affect agriculture are: mitrogen, phosphorous, potash, limestone, slilicic acid, iron, etc. The content of nitrogen, phosphorous, and iron are negligible if any. Therefore the water properties can generally be determined by analyzing the pH, and the content of other elements such as potash, silicic acid, calcium and chlorine. These elements in the water decline during the rainy season, rise in the dry season.

### 2.2 Location of Water Samples Tested

The water was sampled near the junction of the Cantho River and the Mekong River on October 25, 1974.

2.3 Acidity (Ph)

A pH value of 7.3 was recorded which is just above neutural and there therefore slightly alkaline. This pH level is good for vegetables plants and animals.

 $2.4 Potash(K_0)$ 

A finding of 6 PPM of  $K_0^0$  was made. This content is rather high considering that the tests were taken shortly after the rainy season in October.

2.5 Chlorine

A finding of 15 PPM was recorded; this is a negligible quantity and there is no fear of salt water injury to vegetation in the Cantho vicinity (although salinity is a problem in the lower Delta where tidal actions contaminate the soil),

The tests showed 65 PPM of CaCO, which indicates a high calcium content which is beneficial to vegetation and helps offset the soil acidity mentioned earlier.

2.7 Silicic Acid (SiO<sub>2</sub>)

The finding was 16 PPM of silicic acid which is barely sufficient for vegetation.

<sup>2.6</sup> Calcium (CaCO<sub>2</sub>)

### Conclusions

Although the period of sampling was brief and the number of samples limited to 6, the tests were made with the FHK method (Fuji Hira Simplified Test Procedure) and are believed to have given us valid conclusions regarding the properties of soil and water.

The soil in the Delta has a number of shortcomings, the most serious being that it is too dense, sticky, clayey, and does not permit adequate root ventilation for plants and trees.

Vegetation requires a soil containing substantial bacteria; the bacteria in turn require air. Coconut trees, banana and papaya appear stunted and the large trees usually found in tropical areas are not seen in the Delta. This is a further indicator of the lack of bacteria and air in the soil.

Given the existing soil structure, the Delta is generally not suitable for growing vegetables, fruits, legumes and corn. On the other hand, the Delta is one of the best suited areas for paddy cultivation in the world. The productivity of paddy in the Delta is higher than the average of Southeast Asian countries and the local famers have developed tillage methods ideally suited to the conditions of the Delta.

The high productivity of the Delta, in spite of some shortcomings of in soil, should be credited largely to the properties of water from the Mekong River. As mentioned, the Mekong River water has properties making it nearly ideal for irrigation and capable of compensating for shortcomings in the soil.

We believe that the water, in combination with the diligence and skill of tocal farmers, have made it possible to attain high productivity in the Delta paddy land.

Improved, continuing, and stabilized policies will be necessary, however, to maintain and increase the rice potential of the Delta. One such policy relates to calcium. In acid soils, bacteria do not form readily and the decay of organic scenial is retarded. In acid soils, the structure of soils is adversely affected which means that the ventilation and water permeability of the soil is inadequate and the downward thrust of growing roots is inhibited. When the soil is acid, the activated aluminum and iron combine with phosphorous to form insoluable phosphorous which means vegetation cannot absorb the necessary phosphorous.

The absence of air in the soil also means that leguminous plants connot grow well in the Delta soils and, of course, leguminous plant cannot perform their usual function of conveying nitrogen from the air to the soil. The calcium deficit makes the soil acid and destroys the balance of elements usually absorbed by vegetation and causes various plant diseases and thus places a limit on the productivity of certain vegetables. Legumes, vegetables and corn require more calcium than paddy (some species require 20 times more calcium than paddy).

The most suitable amount of powdered limestone for given crops can only be determined after a very throughgoing and detailed study. But on the basis of assumptions valid elsewhere, we believe that 300 to 500 kg. of powdered limestone is needed per hectare per year. This means a very substantial volume of limestone is needed for the Delta's paddy land. (16 provinces of the Delta have 2,039,000 hectares of paddy x 500 kg/year = 1,018,500 tons annually if optimum quantities of limestone were applied.) (\*)

A second policy required to improve productivity is to increase the supply of green manure for the soil. This means a return to the soil of husk, straw and other organic materials. This will make better use of chemical fertilizers and thus save paddy production costs.

The improvement of animal husbandry and poultry raising is an important objective to increase the production of manure of high quality. Fortunately hog raising is widespread, but the cost of feed remains a basic deterent to hog raising. Fatty rice bran is being fed to hogs, but the residual cake from defatted bran would be a substantially better feed for hogs. This oil cake can be mixed with fish meal and other ingredients as a mixed feed. This will be necessary for the effective increase in hog and poltry raising and fish culture. Therefore a plan to make mixed feeds is an important goal.

Industry can play a role in achieving a interchange of plant and soil nutrients. It is important that industry be planned to support agriculture.

(Appendix C ends here)

(\*) On p. 115 of the report, we are recommending that a start be made by focussing on the high acid zones of the Delta. This would mean a minimum requirement in the range of 210,000 to 350,000 tons per year.

- 234 -

APPENDIX D

Ð

LAYOFT OF MACHINE AND CASTING SHOPS ON TAY DO INDESIREAL PARK (PRELENENARY PLAN)









Juch





Drawing courtesy I. Aoike. Member of Cantho Survey Team

### Appendix D-1

MACHINERY SUGGESTED FOR ENGINE REPAIR, MACHINE AND CASTING SHOPS (See Section 4.6 of Report)

I. Engine Repair Shop (Total cost of this equipment would be about \$41,000 C.&F. Cantho)

### I-1 Testing Equipment

Diesel fuel injection pump tester Nozzle tester Diesel injection timing tester Diesel compression gauge Valve and clutch spring tester Con-rod aligner Sound-scope Compression and vacuum gauge Cylinder gauge Exhaust tester (for carbon monoxide testing) Universal electrical tester Timing light Megar Circuit tester Engine scope analyzer Wheel aliginment tester set Torque wrench Thermometer Vernier caliper Outside, inside micrometer sets Engine dynamometer Prinnel hardness tester Tachometer Thickness, radius and pitch gauges Other tools

### I-2 Machining Equipment for Engines

Cylinder boring machine Cylinder honing machine Fast charger Valve seat grinder Valve refacer Brake reliner Tire remover Push-pull master Air compressor Plain trolley chain block Parts cleaner High pressure car washer Portable electric disc grinders Bench grinders Sanders Other tools

1-3 Lubricaters, Fuel, Oil Servicing Equipment Pumps for oil drums Chassis lubricators Gear lubricaters Bucket pumps Other tools 1-4 Plyers, hand tools, jacks, etc. Machine Shop (Total cost of this equipment would be about \$51,000 2. C.AF. Cantho) (Note: The abbreviation, w/a, means, with accessories) Lathe (7 feet) w/aLathe (5 feet) w/aUniversal milling machine w/a Shaping machine w/a Vertical drilling machine w/a Abrasive cut-off machine w/a Bench grinder w/a Bench drill w/a Air compressor w/a A.C are welder w/a Oxy-hydrogen welder, cutter w/a Belt hammer w/a Blower, surface plate, anvil, swage block, others black smith's forging tools Vice, wrench, hammer, taps, dies, others hand tools Dial gauge, micrometer, vernier calipers, steel V block, calipers, others measuring equipment Other tools Casting Shop (Total cost of this equipment would be about \$9,000 3. C.AF. Cantho) (Note: The abbreviation, w/a, means, with accessories.) Small cupola w/a Blower w/a Compressor w/a Sand mill mixer \*/a Chipping hammer w/a Jet chisel w/a Grinder w/cyclone Light thermometer Electric hand, planes, groover, mortiser, circular and jig saws, others carpenter's hand tools Platform scale Chain block

-----

Hand and measuring tools Materials (gray iron, ingst, sand, wood for patterns etc.) Other tools

(Appendix D-1 ends here)

### Amendán R



Note: The mounting tables, covered with sheet steel, own be made locally. This is a multi-purpose mill to grind linestone, ensave (i.e. manice), grinal uses, sorghum.

Brawing courte of Nr. 2. Asite, Noniter, Cantho Agro-industry Dervey

- 239 -

.....

BIBLIOGRAPHY OF REPORTS AND DOCUMENTS USED FOR CANTHO AGRO-INDUSTRY SURVEY

### (A) Mon-specialized Publications (Nos. 1 through 24)

- 1. Belhomme, Philippe, "Suggestions pour le Developement des Agro-industries dans la Bégion du Delta du Mekong", prepared for UNIDO, (draft), Saigon, Oct. 1974
- 2. Industrial Bank of Vietnam, "List of Agro-based Industries Projects being Promoted", Saigon, Oct. 18 1974
- 3. Industrial Development Bank of Vietnam, "Investors Guide", Industrial Development News November, 1974, Saigon
- 4. Industrial Development Bank of Japan, "Economic Indicators "74", about June 1974, Saigon, Vietnam
- 5. McAllister, Michael, "Chemical Market Survey" for AID, USAID/Vietnam about June, 1974
- Matienal Bank of Vietnum, "Economic Bulletin" Vol. XX, No. 3 4, April, 1974, Saigon, Vietnam
- 7. Matienal Bank of Vietnam, "Annual Report", 1972, Saigon, Vietnam
- 8. Sonadesi(National Corporation for the Development of Industrial Zones) "SONADEZI," circa 1973, Saigon, Vietnam
- 9. Sydney M. Cantor Associates, Inc., "Investigation of Foodstuff Import Substitution Opportunities in the Republic of Vietnam," 1973, AID, Washington, D.C.
- Turner, Frank L. "Trip Rocord, Saigon, Cantho, Bangkok, Vienna, Washington, Menlo Park, Culif. (Cantho Region Agro-Industrial Survey) (Includes appendixes provided by SONADEZI, and U.S. Consul General, Cantho), Oct. 9, 1974, Tokyo"
- 11. USAID, Saigon, AGR/CG4, "MR-4 Rainfall," Oct. 25, 1973
- 12. USAID, JEO Library, "Special Studies and Reports on Vietnam," As of June 30, 1974, about July, 1974, Washington, D.C. (General bibliography of prior reports)
- 13. U.S. Dept. of Agriculture, Economic Research Service, International Development Center, "Agriculture in the Vietnam Economy, a System for Economic Analysis," June 1973, Washington D.C.

- 14. U.S. Department of Agriculture, Economic Research Service, International Development Center, "Export Opportunities for Vietnam Agricultural Products in Japan," Oct. 1973, Washington D. C.
- 15. Vietnam, Government of, Ministry of Finance, "Statistiques du Commerce Exterieur du Vietnam", 1972, Saigon, Vietnam
- 16. Vietnam, Government of, Export Development Center, "Vietnam Exports and Products, 1974," about June, 1974, Saigon, Vietnam
- 17. Vietnam, Government of, National Institute of Statistics, "Monthly Bulletin of Statistics," No. 6, 1974, July 1974, Saigon, Vietnam.
- 18. Ibid.Nos. 7, 8, 9, October 1974, Saigon, Vietnam
- 19. Vietnam, Government of Ministry of Agriculture, "Bulletin of Agricultural Economics, Special Issue, 1974, "1974, Saigon, Vietnam
- 20. Vietnam, Government of, Ministry of Agriculture, "Agricultural Statistics", 1960-73, 1974, Saigon
- 21. Vietnam, Government of, Mational Institute of Statistics, "Vietnam Stutistical Yearbook, 1972," 1973, Saigon
- 22. U.S. Consul General, Cantho, "Kien Giang Province", 1974, Cantho, Vietnam
- 23. USAID, Saigon, "Report of PASA Water Management Team," Nov. 1970, Saigon
- 24. USAID, Saigon, "Economic Background Data", July, 1974, Saigon, Vietnam

### (B) Electric Power (25 through 26-A)

- 25. Vietnam, Government of, National Energy Committee, "Energy in the Republic of Vietnam", Saigon, 1974
- 26. Vietnam Power Co., "33 MW Cantho Thermal Power Station and 66 KV Delta Provinces Transmission System Project," no date, circ. 1973, Saigon, Vietnam

26-4.

Hasama Gumi Co., Ltd., "Foundation Engineering (Kiso Ko)"Vol. 3, No. 1, January 1975," "Foundation for Thermal Power Plant in Ganthe, Vietnam" (in Japanese only), Jan., 1975, Tokyo

- 241 -

### (C) Paddy, Buck, Street and Man Proceeding (27 to b6)

- 27. Anderson, Wondel, FAO Rice Task Porce proliminary report on drying, UNDP Saigon, Sept. 1974, Saigon
- 28. Anderson, Wondel, FAO Rice Task Force preliminary report on rice storage, Sept., 1974, Saigen
- 29. Beagle, Eldon C., "Basic and Applied Research Hoeds for Optimising Utilization of Rice Musk," International Conference on Rice Technology, Instituto de Agroquímica y Techelogia de Alimentos, Sept. 30, 1974, Valencia, Apain
- 30. Beagle, Eldon C., "Summary Basis and Applied Research Heeds for Optimizing Utilization of Rice Husk," International Conference on Rice Technology Institute de Agronomique y Tecnologia de Alimentos, Valencia, Spain, September 30, 1974, Valencia, Spain
- 31. Bougle, Elden C. and Boagle, C. Alan, "Paddy Husk Villantien," WHISD, ID/WG.89/16, April 22, 1971, Interregional Seminar on the Industrial Processing of Nice, Vienna, Austria
- 32. Chittenden, A.E., "Paddy Nuck Viligation," UNIDD ID/ME.89/14, Interregional Seminar on the Industrial Proceeding of Nice, April 8, 1971, UNIDD, Vienna Austria
- 33. Cornelius, D. J.K., "Transportation and Storage of Paddy and Mice," UWIDD ID/WG.89/4, Interregional Seminar on the Industrial Processing of Mice, Jan. 15, 1971, UWIDD, Vienna, Austria
- 34. Coudert, FAO Rice Task Force proliminary report on Marketing, UMEP, Saigon, 1974, September
- 35. Dept. of the Army, Engineer Agonay for Resource Inventories, "An Giang Province, Vietnam, Resources Inventory"
- 36. Garvie, D. W. "The Importance of the Rice Processing Industry as an Agro-allied Industry in the Developing Countries." UNIDD ID/NG/89/2, Jan. 22, 1971 Vienna, Austria
- 37. Hulliday, D., "Utilization of Rice Bran," UNIDO ID/W0.89,'9, Interregional Seminar on the Industrial Processing of Rice, UNESD Feb. 2, 1971
- 38. Nasama Gumi, Ltd., "The Piling Report on Structure Meinforcement of Intake in the Cantho Thermal Station", June 18, 1976
- 39. Manning, Eric D., "UNIDO Papers on the Industrial Processing of Mos." March 1972, UNIDO, Vienna Austria
- 40. Masumete, Tayojire; FAO Rice Task Force proliminary report on milling, Sept., 1974, Suigen
- 41. Pe, Myint, "Blee Bren GL1 Technology," UNIDD ID/VG.89/6, Interregional Bominer on the Industrial Proceeding of Mice, 18, Peb. 15, 1971, UNIDD, Vienna, Austria
- 42. Takeshita, Yacuhike, "Technical Advances in Rice Brun Oil Processing," UNINO ID/M.89/5, Interregional Seminar on the Industrial Processing of Rice, Jun. 28, 1971, Vienna, Austria
- 43. WHIRD, FAO, ECAFE, "Report on the Interregional Sominar on the Industrial Processing of Rice," UNIDD ID/VE.89/24, Nov. 19, 1971, Vienna, Austria
- W. U.S. Department of Agriculture, Beenemic Research Service, International Development Center, "Vietnum Agricultural Situation and Near-Term Prospects," January 1973, Machington, D.C.
- 45. Wildman Agricultural Rosearch Inc., "Economic and Engineering Study Grain Storage and Marketing System," March, 1970, Toledo, Ohio (AID contractor)
- 45. Young, Robert B., "Investment and Management Considerations for the Modernization of Traditional Rice Mills and for the Establishment of Nodern Integrated Rice Processing Systems," UNIDD Interregional Seminar on the Industrial Processing of Rice, ID/WG.89/17, June 23, 1971, Vienna, Austria
- 45-A. Pierce, Walter, USAID Saigen, "A Rice Straw Pulp Plant in the Canthe Area -- An Invoctment Opportunity in South Vietnam, Draft Copy, Nay 1976, Saigen

#### (D) Augur (No. 47)

47. Viotnam Sun, The, No. 74-80, "Sugar Industry" (article only), Nov. 17, 1974, Saigen

#### (E) Yamptable GLL (bl. and blink)

(See also rice bran oil publications No. 37 and 41)

48. King, Frank P., Reenemist, "Yeasibility Study -- Edible Oil Seed Production, Marketing and Processing in Vietnam," Oil Seed Team, Production Section, AID, Mashington, July-August, 1971, Mushington, D. C.

NO-1

WEATE Suigen, "CIP Licensing by Conmodity (Grant & Loan)", FY 1973 and 1974, 1974, Saigen, Vietnam

#### (G) Slaughter House (No. 50)

50. VISSAN, Viet-nam Ky-Nghe Suc-An, (Description of Slaughter House, 424 B, Ng-v-Hoc, Gia Dinh financed by West German aid), 1974, Saigon (in Vietnamese and English)

#### Marine Products (No. 51)

51. Agian Development Bank, "ADB Lends US\$6 Million to Vietnam for Second Figheries Development Project," Manila, Dec. 17, 1974 (Press release)

#### Pulp and Paper (Nos. 52 through 53)

52. Taiwan Pulp & Paper Corp., "Teasibility Study of Bleached Pulp Manufacture from Cereal Straw in the Republic of Korea for Korea Paper Manufacturers' Association, Nov.-Dec., 1968, Taipei, Taiwan

52-4.

- COGIDO, "Pulp and Paper Processing" (Title is in Vietnamese, but contents English), 1973, Bien Hoa Industrial Park
- 53. U.S. Department of Agriculture, Economic Research Service, International Development Center, "Timber Development Opportunities in the Republic of Vietnam", June 1974, Washington D. C.

#### (H) Forest Products and Fiber (Nos. 54 through 54-B)

54. Industrial Development Bank of Vietnam, "Wood Industry in Vietnam -Ky Nghe Go tai Vietnam", circ. July, 1974, Suigon

#### 54-4.

USON Saigon, "The Forests of Free Vietnam", 1957, Saigon

#### 54-B.

Dempsey, James M., "Long Vegetable Fiber Development in South Vietnam and Other Asian Countries", February 1963, Saigon, (USAD contract)

#### (I) Fertilizer and Raw Material (Nos. 55 through 56)

- 55. Industrial Bank of Vietnam, "Investment Opportunities in Vietnam", no date, circ. 1974, Saigon
- 56. Industrial Development Bank of Vietnam, Saigon, "Industrial Development News", Nov. 1974, p. 6 ("Oil Exploration in Vietnam"), Nov. 1974, Saigon

## (J) Muchinery Repair and Manufacture (Mos. 57 to 62)

- 57. CARIC, "CARIC" (brochure describing products of CARIC), no date, about 1974, Saigon
- 58. CARIC, "List of Materials and Machinery Interesting the Vietnamese Industry", about 1973, Saigon
- 59. Davis, Parets and Duff, "Summary Situation re Vietnam Participation in IRRI Industrial Outreach Program for Small Form Machinery," USAID, no date but circa 1974, Saigon.
- 60. Davis, D.; Parets, Gaston; Campbell, Joseph and Duff, Bart; TA/AGR, "Industrial Extension of Small Scale Agricultural Equipment Developed at I.R.R.I.", USAID, no date, but circa 1974
- 61. Kamijo, Morio (The Overseus Agricultural Development Co. Ltd.), "On Vietnam -- Special Feature -- Views and Propositions on Agricultural Development." Tokyo, May 1969
- 62. USAID, in Cooperation with U.S. Customs Advisory Team, "Republic of Vietnum Tariff Schedules, English Translation", 1973, Saigon, (included to document duties on tractors)

# (K) Other Industries -- Mushrooms, Essential Oils, Insecticides (Nos. 63 through 65)

- 63. Sekurai, Jun-ichi, "Development Prospects of a Nodern Rice Products Processing Industry in the Republic of Vietnam", prepared for Committee for Co-ordination of Investigations of the Lower Mekong Basin, Economic and Social Studies Division, Bangkok, July 31, 1974 (Mushroom)
- 64. USAID, Suigon, "Vietnam, 1973 Agricultural Pesticide Imports, Annual Report," USAID, January 1974, Saigon
- 65. Nguyen Phuog Du, Denis, "Huiles Essentielles Tropicales," Vietnam Industrial Development Bank, n.d. (about 1970), Saigon

## (L) Investment Policies and Incentives (Nos. 66 through 70)

66. Business International, "Business International Vietnum Roundtable, Feb. 24-27, 1974," Saigon, Feb. 24-27, 1974 (excerpts only)

- 67. Vietnam, Government of, Ministry of Land Reform & Agriculture & Fishery Development, "Five Year Rural Recommic Development Plan, 1971-1975 Agriculture, Fisheries, Forestry and Animal Husbandry, 1970," Saigon, Vietnam
- 68. Vietnam, Government of Min. of National Planning and Development, Directorate General of Planning, "Four-Year National Economic Development Plan," 1972-75, Saigon, Vietnam, Dec. 8, 1972, Saigon, Vietnam
- 69. Vietnam, Government of Ministry of Trade and Industry, Office of the Board of Investment, "On Conditions of Eligibility for Investment Privileges Stipulated by the Investment Law No. 4/72", March 15, 1974, Saigon
- 70. Vietnam, Government of, Investment Service Center, "Regulating Investment in Vietnam, Law No. 4/72 dated June 1972", June 2, 1972, Saigon

#### (M) Maps (Mos. 7) through 80)

- 71. Japan, Government of, Hydrographic Survey, Section of chart showing soundings near mouth of Bassac River (Hua-Giang River), 1974, Tokyo
- 72. National Geographics Lipectorate, Vinh Binh Province, Scale 1:150,000, 1971, Saigon
- 73. Ibid, An Giang Province, Scale 1:150,000, 1971, Saigon
- 74. Ibid., Ba Xuyen Province, Scale 1:150,000, 1971, Saigon
- 75. Ibid., Phong Dinh Province, Scale 1: 1:150,000, Saigon, 1971
- 76. Ibid. Kien Giang Province, Scale 1:250,000, 1971, Saigon
- 77. Ibid., Sa Dec. Province, Scale 1:100,000, 1971, Saigon
- 78. Ibid., Vinh Long Province, Scale 1:150,000, Saigon, 1971
- 79. National Geographic Service of Vietnam, "General Soils Map", Scale 1:1,000,000 (drafted by F. R. Moorman and edited by Ministry of Agriculture), 1961, Dalat
- 80. No author, "Chau Thanh Can Tho" (Map of Canthe City), Scale 1:10,000, no date, about 1970

- 246 -

#### APPENDX G

#### PRASIBILITY STUDY OUTLINE

Chapter 9 of this report starting or page p. 192 refers to the sections of the following "Technical Notes" published as a preface to <u>Extracts of Industrial</u> <u>Studies</u>, Industrial Planning and Programming Series, No. 7, United Nations Industrial Development Organization, Vienna, Volume I, United Nations, New York, 1973.

The text of the "Technical Notes" is reproduced here to facilitate readers in finding sections of the Notes referred to in the text of Chapter 9.

(Terms of Reference for Future Feasibility Studies page, 192.)

The standard form used in this volume consists of 13 sections of which one (X11) is reserved for any supplementary information that does not readily fit into the rest of the form. The major points to be considered in filling in the form are noted section by section and item by item.

- I. ORIGIN OF THE STUDY
- 1. This study was prepared by

Indicate only the type of the institution that prepared the original feasibility study from which this extract is derived, such as an independent consultant, a private consultant firm, a machine supplier, the staff of a governmental development institution, or of an international technical-assistance agency. This study was prepared for

Indicate only the type of the institution that requested the original study, such as an individual private investor, a ministry of industry, or an investment bank.

- 2. The study was intended to Specify the main investment opportunity that motivated the feasibility study: an investment priorities plan, preliminary sectoral studies, etc. Important factors underlying the opportunity may be mentioned.
- 3. Size of the economy considered

- Other information

List the main established industries and point out the particular resources with which they are favourably endowed. The geographical region in which the economy is located may or may not be mentioned. If the economy involves more than one country, this should be mentioned.

#### **II. GENERAL DESCRIPTION**

1. Products

Describe briefly the technical specifications of the products: e.g., chemical purity, mechanical properties, quality standards, composition. Quantities and product-mix are to be indicated elsewhere.

2. Major input materials

Give the qualitative specifications of major raw materials from local origins and any important features of their supply; and indicate important production materials, if any, that are needed but not locally available. 3. Alternative technologies available and technology adopted for the study

Indicate the type of process, such as singleproduct process, multi-product process, singletrain or multi-train production;

Describe briefly the technology, mentioning any special name popularly used by experts in the field. Give details in section X11 ("Supplement") as deemed appropriate;

State the degree of mechanization and automation. Special advanced features may be described in section XII ("Supplement").

4. Locational factors

- Particularly important factors Specify the key parameters for transport requirements: quantities of main input materials and output products in terms of gross weights and/or volumes;

Indicate means of transport and transport tariffs applicable;

State the desired degree of proximity to electricity, water and basic infrastructure, including the location of other industries with which the project is to be closely linked;

Indicate crucial regional policy measures recommended for the project feasibility, if any.

 Actually proposed locality
Describe simply the proposed locality in terms of differential priorities given to important factors.

#### III. MARKET

1. Tabulation of estimated demand on domestic and export markets

Tabulate separately for the domestic and the relevant export markets;

Indicate any special characteristics, possition margins of error (between optimistic and pessimistic), etc.

2. Notes on methodology

Indicate major determinant variables, demand elasticities and other key parameters for projections, and special consumer patterns; Give an indication of how crude or sophisticated the estimates are:

Describe any special market surveys conducted in the field.

#### 3. Selection of product-mix

Justify the selected product-mix from the standpoints of both the market and the production technology. Indicate potential additional products to be included in a future extension of the project.

#### IV. CAPACITY OF PROPOSED PLANT

1. Nominal maximum capacity according to major process

Give the time basis of calculation (operating time) and indicate specific groups of machines and equipment that determine the nominal maximum capacity of the major process or processes;

Indicate whether the proposed capacity represents the technologically acceptable minimum plant scale.

2. Maximum feasible capacity of the plant Calculate the feasible capacity of the plant, taking into account normal stoppage, desired shift patterns, indivisibilities of major machines to be combined, etc.;

Indicate potential bottlenecks and selective extension possibilities.

3. Expected maximum output of the plant Indicate expected maximum output, compatible with the expected markets, in percentage of the maximum feasible capacity. Efficiency of labour, demand irregularities, proportion of rejects, seasonal variations of raw material supplies etc. should be taken into account.

#### V. INVESTMENT

- I.I. Land, site development Indicate size of plot in parenthesis.
- 1.2. Duildings

Indicate floor space in parenthesis for each sub-item.

- Others

Describe water facilities, electricity works, reservoirs, waste-disposal systems, housing for employees, etc., separately if possible.

1.3. Machinery and equipment

Give total, including installation.

(Indicate in the foot-note here or in section XII ("Supplement") any extraordinarily high or low estimates of particular items that might be associated with special road and housing development schemes, power supply, leasing of equipment, special properties of inputs and outputs, time-phasing of a potentially larger integrated project, etc.)

#### 2. Working capital

3.

State in ex ante planning terms, referring to the normal level of operation expected after the completion of the plant and its start-up.

2.1. Inventories Indicate in parenthesis

Indicate in parenthesis the equivalent number of months.

- 2.2. Accounts receivable Indicate average period of deferred payment allowed to customers, in parenthesis.
  - Other investments Calculate expenditures prior to the start o production that are to be capitalized. Major machinery and equipment (table) List (a) production machinery and equipment and (b) auxiliary equipment (transport, laboratory, maintenance, power generation, office equipment, etc.) separately; Production machinety to the start

Production machinery should be listed preferably by departments or shops rather than item by item, with classification corresponding to the shop alignment as in section VI ("Manning table");

If space permits, indicate the capacity rating of the machinery and equipment by shop, or for predominantly important items. Use section XII ("Supplement") if necessary.

#### VI. MANNING TABLE

List of shops should preferably indicate the main processing stages involved. A process flow chart supporting the given shop alignment may be shown in section XII ("Supplement"); Seasonal workers should be so specified.

#### **VII. ANNUAL PRODUCTION**

- 1. Total annual expected maximum output Tabulate by product and by destination. "Unit price ex factory" would include production and/or sales taxes, if any. Special subsidized export prices should be indicated.
- 2. Expected sales and inventory build-up Give background information on the expected growth of turnover and capacity utilization during the first few years of production, as tabulated in section XIII ("Cash flow table").
- 3. Pricing policy

Compare the prices derived from direct costing with current import prices. The latter should be clarified in terms of the c.i.f. prices, free of import duties at the port of import, and the normal rates of surcharge including duties, sales taxes, trade and transport margins; State any specifics regarding the proposed export prices;

Indicate the results of any sensitivity tests concerning the desired level of profitability and the pricing of products;

Indicate any necessary governmental protective measures (the justification for such measures should be given in section XI ("Data for evaluation").

4. Planned sales organization

Describe own distribution network via sales representatives and own retail stores or agents, wholesalers and other trading organizations; also the significance of own transport facilities in marketing.

#### VIII. ANNUAL OPERATING COSTS AND PROFITS

The data in this section should correspond to the "annual expected maximum output" as shown in section VII.

State separately and in detail: (a) raw materials and semi-processed materials, (b) packaging materials, (c) repair and maintenance supplies, (d) energy, and (e) water and other materials.

- Note: Office supplies, advertisements, insurance fees, communication, staff travel and other business services to be purchased should be included in item 7 ("Administrative expenses and sales costs"). Item 7 should not include wages and salaries for the enterprise's employees.
- 3. Interests

Give average annual interest charges on borrowed capital as planned. Interest on foreign loans should not be included here.

- 5. Indirect taxes List value added tax, production tax, turnover tax, employment tax, franchise tax, etc. Profit tax is to be shown in item 9.
- 6. Depreciation Indicate rate and method of depreciation. Use space in section X11 ("Supplement") if necessary.
- 7. Administrative expenses See note above.
- 8. Other costs Include work by outside firms on contract basis, if any; otherwise, "contingency allowances" would normally fit under this item.

#### IX. FINANCING PROPOSAL

2. Long-term loans List separately loans of different terms.

#### 3. Other loans

List separately short- and medium-term loans; indicate interest and repayment conditions for each.

- 4. Suppliers' credits Indicate over-all repayment conditions.
- 5. Remarks on the financing policy Indicate whether the financial proposal is only a crude, somewhat arbitrary assumption or whether it reflects the proposal from a potential investment institution. Also state local and foreign sources of capital. Describe briefly any peculiarities of the financing proposal.

#### X. IMPLEMENTATION

- 1. Technical collaboration service Include projections or recommendations concerning further studies required, preparation of project blue-prints, contracting procedures and other major steps to be taken if the project proves acceptable for the purpose for which the original feasibility study was prepared.
- 2. Project management

Indicate, e.g., whether a turn-key contract is envisaged and any specific arrangements proposed for project management during the construction and/or initial operation period.

- 3. Recruitment and training of personnel Describe any programmes for training abroad and/or locally. Also describe the proposed time schedule for recruitment of technical personnel, skilled workers, etc.
- 4. Other items

Describe any organizational problems, infrastructural requirements to be satisfied, crucial legislative actions required to ensure the viability of the industry considered, etc.

5. Time schedule

Give time schedule proposed for major implementation activities, covering contracting and other pre-construction activities, construction schedules start-up and extension phase. The schedule would underlie the calculations presented in section XIII ("Cash flow table").

#### XI. DATA FOR EVALUATION

Check the type of analysis included in the original feasibility study and summarize the main findings. Any incorrect or inadequate treatments involved in the original study may be pointed out, and an alternative analysis may be undertaken and presented by those who prepare this extract. The original feasibility study document used for this extract may or may not be complete in terms of project evaluation. Recommendations made in the original document may or may not be reasonable. A space in this section may well be spared for an expert evaluation pinpointing any notable weak points of the original project study.

#### XII. SUPPLEMENT

1

Use this space to provide additional information relating to any sections of the extract. Any further details or commentaries that require extra space or non-pro forma presentation should be given here, and reference made in each case to the relevant section and its sub-item.

#### XIII. CASH FLOW TABLE

Use this standard cash flow table to ensure comparability among different cases. When more than one cash flow table is available and/or a given cash flow table is in a special form that cannot be reorganized in this standard form, it should be presented in section XII ("Supplement").

Note that, in this tabulation, no. 4 ("Production expenditure") does not include interests on loans and depreciation (which are included in section VIII, sub-sections 3 and 6 respectively). Interests are entered in sub-section B.5.1 ("Interest on loans"). Instead of depreciation allowances, the anticipated replacement expenditures are to be entered in sub-section B.1.3 ("Machinery and equipment (replacement)"). This table is arranged in such a way that internally accumulated profits and depreciation funds are not so isolated, but are absorbed into sub-section C ("Surplus/De-'), after being adjusted for yearly expenditures ficit' on the capital account (replacement expenditures and repayments of loans and credits).







# 82.09.22